UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, DC 20549

FORM 10-Q

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2009

Commission file number: 1-11416

CONSUMER PORTFOLIO SERVICES, INC.

(Exact name of registrant as specified in its charter)

California 33-0459135
(State or other jurisdiction of incorporation or organization) Identification No.)
19500 Jamboree Road, Irvine, California 92612
(Address of principal executive offices) (Zip Code)

Registrant's telephone number, including Area Code: (949) 753-6800

Former name, former address and former fiscal year, if changed since last report: N/A

Indicate by check mark whether the registrant (1) filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports) and (2) has been subject to such filing requirements for the past 90 days.

Yes [X] No []

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes [] No [X]

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer a non-accelerated filer or a smaller reporting company. See definition of "accelerated filer", "large accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large Accelerated Filer [] Accelerated Filer [] Non-Accelerated Filer [] Smaller Reporting Company [X]

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes [] No [X]

As of November 9, 2009 the registrant had 18,223,794 common shares outstanding.

CONSUMER PORTFOLIO SERVICES, INC. AND SUBSIDIARIES INDEX TO FORM 10-Q For the Quarterly Period Ended September 30, 2009

		Page
	PART I. FINANCIAL INFORMATION	
Item 1.	Financial Statements	
	Unaudited Condensed Consolidated Balance Sheets as of September 30, 2009 and December 31, 2008	3
	Unaudited Condensed Consolidated Statements of Operations for the three-month and nine-month periods ended September	
	30, 2009 and 2008	4
	Unaudited Condensed Consolidated Statements of Cash Flows for the nine-month periods ended September 30, 2009 and	
	2008	5
	Notes to Unaudited Condensed Consolidated Financial Statements	6
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	19
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	33
Item 4T.	Controls and Procedures	33
	PART II. OTHER INFORMATION	
Item 1.	Legal Proceedings	34
Item 1A.	Risk Factors	34
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	37
Item 4.	Submission of Matters to a Vote of Security Holders	38
Item 5.	Other Information	38
Item 6.	Exhibits	38
	Signatures	40
	2	

CONSUMER PORTFOLIO SERVICES, INC. AND SUBSIDIARIES UNAUDITED CONDENSED CONSOLIDATED BALANCE SHEETS (In thousands, exept share and per share data)

	September 30,	December 31, 2008	
	2009	2008	
ASSETS			
Cash and cash equivalents	\$ 20,396	\$ 22,084	
Restricted cash and equivalents	132,236	153,479	
Finance receivables	1,009,079	1,417,343	
Less: Allowance for finance credit losses	(33,066)	(78,036)	
Finance receivables, net	976,013	1,339,307	
Residual interest in securitizations	4,144	3,582	
Furniture and equipment, net	1,410	1,404	
Deferred financing costs	6,755	8,954	
Deferred tax assets, net	52,727	52,727	
Accrued interest receivable	10,193	14,903	
Other assets	20,891	42,367	
	\$ 1,224,765	\$ 1,638,807	
LIABILITIES AND SHAREHOLDERS' EQUITY			
Liabilities Liabilities			
Accounts payable and accrued expenses	\$ 24,396	\$ 21,702	
Warehouse lines of credit	5,171	9,919	
Residual interest financing	59,850	67,300	
Securitization trust debt	1,012,787	1,404,211	
Senior secured debt, related party	20,842	20,105	
Subordinated renewable notes	21,232	25,721	
	1,144,278	1,548,958	
COMMITMENTS AND CONTINGENCIES	_,_ : :,_ : 0	_,_ ;,,,,,,,,	
Shareholders' Equity			
Preferred stock, \$1 par value;			
authorized 5,000,000 shares; none issued	-	-	
Series A preferred stock, \$1 par value;			
authorized 5,000,000 shares; none issued	-	-	
Common stock, no par value; authorized			
75,000,000 shares; 18,415,427 and 19,110,777			
shares issued and outstanding at September 30, 2009 and			
December 31, 2008, respectively	55,209	54,702	
Additional paid in capital, warrants	8,371	7,471	
Retained earnings	23,934	34,703	
Accumulated other comprehensive loss	(7,027)	(7,027)	
	80,487	89,849	
	\$ 1,224,765	\$ 1,638,807	

See accompanying Notes to Unaudited Condensed Consolidated Financial Statements.

CONSUMER PORTFOLIO SERVICES, INC. AND SUBSIDIARIES

UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(In thousands, except per share data)

	_	Three Months Ended September 30,		Nine Months Ende September 30,				
		2009		2008		2009		2008
Revenues:								
Interest income	\$	49,320	\$	87,706	\$	165,459	\$	281,924
Servicing fees		839		235		2,811		944
Other income		2,662		3,775		8,922		10,930
		52,821		91,716		177,192		293,798
Expenses:								
Employee costs		8,224		12,455		26,466		38,824
General and administrative		5,569		7,497		18,021		22,417
Interest		26,175		40,963		87,278		120,952
Provision for credit losses		15,258		25,961		49,836		91,764
Loss on sale of receivables		-		13,963		-		13,963
Marketing		895		2,417		2,975		8,658
Occupancy		825		972		2,859		3,011
Depreciation and amortization		182		126		526		364
		57,128		104,354		187,961		299,953
Income (loss) before income tax expense (benefit)		(4,307)		(12,638)		(10,769)		(6,155)
Income tax expense (benefit)		-		(6,312)		-		(3,432)
Net income (loss)	\$	(4,307)	\$	(6,326)	\$	(10,769)	\$	(2,723)
Earnings (loss) per share:								
Basic	\$	(0.23)	¢	(0.32)	¢	(0.57)	¢	(0.14)
Diluted	Ψ	(0.23)	Ψ	(0.32)	Ψ	(0.57)	Ψ	(0.14)
		(1.20)		(3.32)		(3.37)		(3.2.)
Number of shares used in computing								
earnings (loss) per share:								
Basic		18,583		19,693		18,776		19,275
Diluted		18,583		19,693		18,776		19,275

 $See\ accompanying\ Notes\ to\ Unaudited\ Condensed\ Consolidated\ Financial\ Statements.$

CONSUMER PORTFOLIO SERVICES, INC. AND SUBSIDIARIES UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands)

		nths Ended
		nber 30,
Cook flows from an arcting activities	2009	2008
Cash flows from operating activities: Net income (loss)	\$ (10,769)	\$ (2,723)
Adjustments to reconcile net income (loss) to net cash	\$ (10,709)	(2,723)
provided by operating activities:		
Amortization of deferred acquisition fees	(5,825)	(12,583)
Amortization of discount on securitization notes	9,327	10,117
Amortization of discount on senior secured debt, related party	737	-
Depreciation and amortization	526	364
Amortization of deferred financing costs	2,199	8,367
Provision for credit losses	49,836	91,764
Stock-based compensation expense	1,027	953
Interest income on residual assets	(1,211)	(493)
Loss on sale of receivables	` -	13,963
Change in market value of warrants	77	555
Changes in assets and liabilities:		
Accrued interest receivable	4,711	7,257
Other assets	24,669	(2,541)
Tax assets	-	4,968
Accounts payable and accrued expenses	2,694	(2,471)
Tax liabilities	-	(8,860)
Net cash provided by operating activities	77,998	108,637
	<u> </u>	
Cash flows from investing activities:		
Purchases of finance receivables held for investment	(2,539)	(289,760)
Proceeds received on finance receivables held for investment	321,821	487,396
Proceeds received on sale of receivables	-	159,855
Increases (decreases) in restricted cash and equivalents	21,243	2,567
Purchase of furniture and equipment	(532)	
Net cash provided by investing activities	339,993	359,911
recease provided by investing dedivines	553,555	555,511
Cash flows from financing activities:		
Proceeds from issuance of securitization trust debt	_	285,389
Proceeds from issuance of subordinated renewable notes	527	3,808
Proceeds from issuance of senior secured debt	527	25,000
Payments on subordinated renewable notes	(5,016)	
Net proceeds from (repayments to) warehouse lines of credit	(4,748)	
Proceeds from (repayments of) residual interest financing debt	(7,450)	
Repayment of securitization trust debt	(400,751)	
Payment of financing costs	(1,722)	
Repurchase of common stock	(519)	
Exercise of options and warrants	(513)	144
Net cash used in financing activities	(419,679)	
Net Cash used in inidicing activities	(413,079)	(400,130)
Increase (decrease) in cash and cash equivalents	(1,688)	2,350
Cash and cash equivalents at beginning of period	22,084	20,880
Cash and cash equivalents at end of period	\$ 20,396	\$ 23,230
Cash and Cash equivalents at end of period	φ 20,390	\$ 23,230
Supplemental disclosure of cash flow information:		
Cash paid (received) during the period for:	ф	Ф
Interest	\$ 76,840	
Income taxes	\$ (12,374)	\$ 460
Non-cash financing activities:	ф.	ф 1001
Common stock issued in connection with new senior secured debt	\$ -	\$ 1,801
Warrants issued in connection with new senior secured debt	\$ -	\$ 6,122
Warrants issued in connection with new warehouse line of credit	\$ 822	\$ -
Non-cash investing and operating activities:	ф	ф 2.452
Residual interest in structured sale of receivables	\$ -	, -
Notes received upon structured sale of receivables	\$ -	\$ 8,542

(1) Summary of Significant Accounting Policies

Description of Business

We were formed in California on March 8, 1991. We specialize in purchasing and servicing retail automobile installment sale contracts ("automobile contracts" or "finance receivables") originated by licensed motor vehicle dealers located throughout the United States ("dealers") in the sale of new and used automobiles, light trucks and passenger vans. Through our purchases, we provide indirect financing to dealer customers for borrowers with limited credit histories, low incomes or past credit problems ("sub-prime customers"). We serve as an alternative source of financing for dealers, allowing sales to customers who otherwise might not be able to obtain financing. In addition to purchasing installment purchase contracts directly from dealers, we have also (i) acquired installment purchase contracts in three merger and acquisition transactions, (ii) purchased immaterial amounts of vehicle purchase money loans from non-affiliated lenders, and (iii) lent money directly to consumers for an immaterial amount of vehicle purchase money loans. In this report, we refer to all of such contracts and loans as "automobile contracts."

Basis of Presentation

Our Unaudited Condensed Consolidated Financial Statements have been prepared in conformity with accounting principles generally accepted in the United States of America, with the instructions to Form 10-Q and with Article 8 of Regulation S-X of the Securities and Exchange Commission, and include all adjustments that are, in our opinion, necessary for a fair presentation of the results for the interim period presented. All such adjustments are, in the opinion of management, of a normal recurring nature. In addition, certain items in prior period financial statements may have been reclassified for comparability to current period presentation. Results for the nine-month period ended September 30, 2009 are not necessarily indicative of the operating results to be expected for the full year.

Certain information and footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States of America have been condensed or omitted from these Unaudited Condensed Consolidated Financial Statements. These Unaudited Condensed Consolidated Financial Statements should be read in conjunction with the Consolidated Financial Statements and Notes to Consolidated Financial Statements included in our Annual Report on Form 10-K for the year ended December 31, 2008.

Use of Estimates

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities as of the date of the financial statements, as well as the reported amounts of income and expenses during the reported periods. Specifically, a number of estimates were made in connection with determining an appropriate allowance for finance credit losses, valuing residual interest in securitizations, accreting net acquisition fees, amortizing deferred costs, the recording of deferred tax assets and reserves for uncertain tax positions. These are material estimates that could be susceptible to changes in the near term and, accordingly, actual results could differ from those estimates.

Other Income

Other income consists primarily of gains recognized on our residual interest in securitizations, recoveries on previously charged off contracts from previously unconsolidated trusts, convenience fees charged to obligors for certain types of payment transaction methods and fees paid to us by dealers for certain direct mail services we provide. The gain recognized related to the residual interest was \$635,000 and \$178,000 for the nine months ended September 30, 2009 and 2008, respectively. The recoveries on the charged-off contracts relate to contracts from previously unconsolidated trusts were \$1.2 million and \$1.7 million for the nine months ended

September 30, 2009 and 2008, respectively. The convenience fees charged to obligors, which can be expected to increase or decrease in rough proportion to increases or decreases in our managed portfolio, were \$3.5 million and \$4.3 million for the same periods, respectively. The direct mail revenues were \$2.1 million and \$4.2 million for the nine months ended September 30, 2009 and 2008, respectively.

Stock-based Compensation

We recognize compensation costs in the financial statements for all share-based payments granted subsequent to January 1, 2006 based on the grant date fair value estimated in accordance with the provisions of FASB ASC 718 "Accounting for Stock Based Compensation".

For the nine months ended September 30, 2009 and 2008, we recorded stock-based compensation costs in the amount of \$1.0 million and \$953,000, respectively. As of September 30, 2009, unrecognized stock-based compensation costs to be recognized over future periods equaled \$3.7 million. This amount will be recognized as expense over a weighted-average period of 3.4 years.

The following represents stock option activity for the nine months ended September 30, 2009:

			Weighted
	Number of	Weighted	Average
	Shares	Average	Remaining
			Contractual
	(in thousands)	Exercise Price	Term
Options outstanding at the beginning of period	6,320	\$ 4.35	N/A
Granted	1,395	0.77	N/A
Exercised	-	-	N/A
Forfeited	(520)	4.45	N/A
Options outstanding at the end of period	7,195	\$ 3.65	5.82 years
Options exercisable at the end of period	4,576	\$ 4.01	4.35 years

At September 30, 2009, the aggregate intrinsic value of options outstanding and exercisable was \$6,000 and \$600,000, respectively. The total intrinsic value of options exercised was \$50,000 for the nine months ended September 30, 2008. New shares were issued for all options exercised during the nine-month period ended September 30, 2008. There were no options exercised for the nine months ended September 30, 2009. There were 1.3 million shares available for future stock option grants under existing plans as of September 30, 2009.

At the annual shareholders meeting in July 2009, our shareholders approved an amendment of the 2006 Long-term Equity Incentive Plan which will permit an exchange and repricing of eligible outstanding options to purchase approximately 3.96 million shares with exercise prices ranging from \$2.50 to \$7.18 per share. Under this Option Exchange Program, eligible employees would be able to elect to exchange outstanding eligible options for new options with an exercise price of \$1.50. This transaction will result in some additional compensation expense. Assuming a market price for our stock of \$1.00 per share on the date of the exchange, if all the options eligible for the Option Exchange Program are exchanged, we will incur additional compensation expense, in accordance with applicable accounting rules, of approximately \$457,000 at the time of the exchange and approximately \$232,000 over the remaining vesting periods of the exchanged options. The actual amounts of additional compensation expense will vary to the extent the market price for our stock is more or less than \$1.00 per share at the time of the exchange.

We use the Black-Scholes option valuation model to estimate the fair value of each option on the date of grant, using the assumptions noted in the following table. The expected term of options granted is computed as the mid-point between the vesting date and the end of the contractual term. The risk-free rate is based on U.S. Treasury instruments in effect at the time of grant whose terms are consistent with the expected term of our stock options. Expected volatility is based on historical volatility of our stock. The dividend yield is based on historical experience and the lack of any expected future changes.

	Months Ended otember, 30
	2009
Risk-free interest rate	\$ 1.98%
Expected term, in years	\$ 5.4
Expected volatility	\$ 79.28%
Dividend yield	\$ 0%

Purchases of Company Stock

During the nine-month periods ended September 30, 2009 and 2008, we purchased 695,350 and 1,244,098 shares, respectively, of our common stock, at average prices of \$0.75 and \$2.71, respectively.

New Accounting Pronouncements

In June 2009, the FASB issued SFAS No. 166, Accounting for Transfers of Financial Assets – an amendment of FASB Statement No. 140, ("SFAS 166") (ASC 860 10 65-2). SFAS 166 (ASC 860 10 65-2) limits the circumstances in which a financial asset should be derecognized when the transferor has not transferred the entire financial asset by taking into consideration the transferor's continuing involvement. The standard requires that a transferor recognize and initially measure at fair value all assets obtained (including a transferor's beneficial interest) and liabilities incurred as a result of a transfer of financial assets accounted for as a sale. The concept of a qualifying special-purpose entity is removed from SFAS No. 140, Accounting for Transfers and Servicing of Financial Assets and Extinguishments of Liabilities along with the exception from applying Financial Accounting Standards Board Interpretation ("FIN") 46(R) Consolidation of Variable Interest Entities ("FIN 46(R)") (ASC 810 10 25-20). The standard is effective for us beginning with the first quarter in 2010. We are currently evaluating the impact that SFAS 166 (ASC 860 10 65-2) will have on our consolidated financial position, results of operations or cash flows.

In June 2009, the FASB issued SFAS No. 167, Amendments to FASB Interpretation No. 46(R), ("SFAS 167") (ASC 810 10 25-20). The standard amends FIN 46(R) (ASC 810 10 25-20) to require a company to analyze whether its interest in a variable interest entity ("VIE") gives it a controlling financial interest. A company must assess whether it has an implicit financial responsibility to ensure that the VIE operates as designed when determining whether it has the power to direct the activities of the VIE that significantly impact its economic performance. Ongoing reassessment of whether a company is the primary beneficiary is also required by the standard. SFAS 167 (ASC 810 10 25-20) amends the criteria to qualify as a primary beneficiary as well as how to determine the existence of a VIE. The standard also eliminates certain exceptions that were available under FIN 46(R) (810 10 25-20). SFAS 167 (ASC 810 10 25-20) is effective for us beginning with the first quarter in 2010. Comparative disclosures will be required for periods after the effective date. We are currently evaluating the impact that SFAS 167 (ASC 810 10 25-20) will have on our consolidated financial position, results of operations or cash flows.

In June 2009, the FASB issued SFAS No. 168, The FASB Accounting Standards CodificationTM and the Hierarchy of Generally Accepted Accounting Principles – a replacement of FASB Statement No. 162 ("SFAS 168") (ASC 105 10 65-1) as the source of authoritative generally accepted accounting principles for nongovernmental entities. SFAS 168 (ASC 105 10 65-1) is effective for interim and annual periods ending after September 15, 2009 and did not have any impact on our consolidated financial statements.

Recent Developments

Uncertainty of Capital Markets and General Economic Conditions

Historically, we have depended upon the availability of warehouse credit facilities and access to long-term financing through the issuance of asset-backed securities collateralized by our automobile contracts. Since

1994, we have completed 49 term securitizations of approximately \$6.6 billion in contracts. We conducted four term securitizations in 2006, four in 2007, and two in 2008. From July 2003 through April 2008 all of our securitizations were structured as secured financings. The second of our two securitization transactions in 2008 (completed in September 2008) was in substance a sale of the related contracts, and is treated as a sale for financial accounting purposes.

Since the fourth quarter of 2007, we have observed unprecedented adverse changes in the market for securitized pools of automobile contracts. These changes include reduced liquidity, and reduced demand for asset-backed securities, particularly for securities carrying a financial guaranty and for securities backed by sub-prime automobile receivables. Moreover, many of the firms that previously provided financial guarantees, which were an integral part of our securitization program, are no longer offering such guarantees. From December 2008 until September 25, 2009 we had no available warehouse credit facilities. On September 25, 2009 we entered into a \$50 million two-year credit facility. Unlike our recent former credit facilities, which limited the timeframe for eligible collateral to 180 days, collateral can remain eligible in this facility for the entire two-year term. Consequently, we have no immediate plans to complete a term securitization. The adverse changes that have taken place in the market have caused us to seek to conserve liquidity by reducing our purchases of automobile contracts to nominal levels. If the current adverse circumstances that have affected the capital markets should continue or worsen, we may curtail further or cease our purchases of new automobile contracts, which could lead to a material adverse effect on our operations.

Current economic conditions have negatively affected many aspects of our industry. First, as stated above, there is reduced demand for asset-backed securities secured by consumer finance receivables, including sub-prime automobile receivables. Second, lenders who previously provided short-term warehouse financing for sub-prime automobile finance companies such as ours are reluctant to provide such short-term financing due to the uncertainty regarding the prospects of obtaining long-term financing through the issuance of asset-backed securities. In addition, many capital market participants such as investment banks, financial guaranty providers and institutional investors who previously played a role in the sub-prime auto finance industry have withdrawn from the industry, or in some cases, have ceased to do business. Finally, the broad economic weakness and increasing unemployment has made many of the obligors under our receivables less willing or able to pay, resulting in higher delinquency, charge-offs and losses. Each of these factors has adversely affected our results of operations. Should existing economic conditions worsen, both our ability to purchase new contracts and the performance of our existing managed portfolio may be impaired, which, in turn, could have a further material adverse effect on our results of operations.

Financial Covenants

Certain of our securitization transactions and our warehouse credit facility contain various financial covenants requiring certain minimum financial ratios and results. Such covenants include maintaining minimum levels of liquidity and net worth and not exceeding maximum leverage levels and maximum financial losses. In addition, certain securitization and non-securitization related debt contain cross-default provisions that would allow certain creditors to declare a default if a default occurred under a different facility.

The agreements under which we receive periodic fees for servicing automobile contracts in securitizations are terminable by the respective financial guaranty insurance companies (also referred to as note insurers) upon defined events of default, and, in some cases, at the will of the insurance company. Without the waivers we have received from the related note insurers, we would have been in violation of certain financial and operating covenants relating to minimum net worth and maintenance of active warehouse facilities with respect to eight of our 17 currently outstanding securitization transactions with this filing. Upon such an event of default, and subject to the right of the related note insurers to waive such terms, the agreements governing the securitizations call for payment of a default insurance premium, ranging from 25 to 100 basis points per annum on the aggregate outstanding balance of the related insured senior notes, and for the diversion of all excess cash generated by the assets of the respective securitization pools into the related spread accounts to increase the credit enhancement associated with those transactions. The cash so diverted into the spread accounts would otherwise be used to make principal payments on the subordinated notes in each related

securitization or would be released to us. In addition, upon an event of default, the note insurers have the right to terminate us as servicer. Although the diversion of such cash and our termination as servicer have been waived, we are paying default premiums, or their equivalent, with respect to insured notes representing \$593.6 million of the \$1,012.8 million of securitization trust debt outstanding at September 30, 2009. It should be noted that the principal amount of such securitization trust debt is not increased, but that the increased insurance premium is reflected as increased interest expense. Furthermore, such waivers are temporary, and there can be no assurance as to their future extension. We do, however, believe that we will obtain such future extensions of our servicing agreements because it is generally not in the interest of any party to the securitization transaction to transfer servicing. Nevertheless, there can be no assurance as to our belief being correct. Were an insurance company in the future to exercise its option to terminate such agreements or to pursue other remedies, such remedies could have a material adverse effect on our liquidity and results of operations, depending on the number and value of the affected transactions. Our note insurers continue to extend our term as servicer on a monthly and/or quarterly basis, pursuant to the servicing agreements.

(2) Finance Receivables

The following table presents the components of Finance Receivables, net of unearned interest and deferred acquisition fees and originations costs:

	Sep	otember 30,	D	ecember 31,
		2009		2008
Finance Receivables		(In thou	ısan	ds)
Automobile finance receivables, net of unearned interest	\$	1,016,398	\$	1,430,227
Less: Unearned acquisition fees and originations costs		(7,319)		(12,884)
Finance Receivables	\$	1,009,079	\$	1,417,343

The following table presents a summary of the activity for the allowance for credit losses for the nine-month periods ended September 30, 2009 and 2008:

	Sept			2008
		(=== ====		,
Balance at beginning of period	\$	78,036	\$	100,138
Provision for credit losses on finance receivables		49,836		91,764
Charge offs		(116,949		(142,702)
Recoveries		22,143		23,590
Allowance attributable to receivables sold		-		(5,871)
Balance at end of period	\$	33,066	\$	66,919

We have excluded from finance receivables those contracts that we previously classified as finance receivables, but which we reclassified as other assets because we have repossessed the vehicle securing the contract. The following table presents a summary of such repossessed inventory, together with the valuation adjustment for losses in repossessed inventory that is not included in the allowance for credit losses. This valuation adjustment results in the repossessed inventory being valued at the estimated fair value less selling costs.

	Sept	September 30,		ecember 31,
		2009		2008
		(In thousands)		
Gross balance of repossessions in inventory	\$	32,841	\$	47,452
Adjustment for losses on repossessed inventory		(25,009)		(32,690)
Net repossessed inventory included in other assets	\$	7,832	\$	14,762

(3) Securitization Trust Debt

We have completed a number of securitization transactions that are structured as secured borrowings for financial accounting purposes. The debt issued in these transactions is shown on our Unaudited Condensed Consolidated Balance Sheets as "Securitization trust debt," and the components of such debt are summarized in the following table:

						Weighted
						Average
	Final	Receivables		Outstanding	Outstanding	Contractual
						Interest Rate
	Scheduled	Pledged at		Principal at	Principal at	at
	Payment	September 30,	Initial	September 30,	December 31,	September 30,
Series	Date (1)	2009	Principal	2009	2008	2009
			(Dollars in tho	usands)		
CPS 2003-C	March 2010	\$ -	\$ 87,500	\$ -	\$ 1,023	-
CPS 2003-D	October 2010	-	75,000	-	1,704	-
CPS 2004-A	October 2010	700	82,094	791	3,277	4.32%
CPS 2004-B	February 2011	1,753	96,369	1,950	6,192	4.17%
CPS 2004-C	April 2011	2,830	100,000	3,151	8,223	4.24%
CPS 2004-D	December 2011	-	120,000	-	12,395	-
CPS 2005-A	October 2011	9,489	137,500	9,225	17,586	5.30%
CPS 2005-B	February 2012	13,116	130,625	12,570	21,991	4.67%
CPS 2005-C	March 2012	24,642	183,300	23,908	39,478	5.13%
CPS 2005-TFC	July 2012	6,686	72,525	6,686	12,333	5.76%
CPS 2005-D	July 2012	22,973	145,000	23,073	36,548	5.73%
CPS 2006-A	November 2012	48,529	245,000	48,346	73,257	5.33%
CPS 2006-B	January 2013	62,978	257,500	64,413	92,106	6.46%
CPS 2006-C	June 2013	70,376	247,500	72,653	101,716	5.76%
CPS 2006-D	August 2013	76,958	220,000	77,663	105,687	5.65%
CPS 2007-A	November 2013	120,008	290,000	119,033	160,122	5.58%
CPS 2007-TFC	December 2013	35,228	113,293	35,221	51,115	5.77%
CPS 2007-B	January 2014	149,344	314,999	149,414	195,800	6.10%
CPS 2007-C	May 2014	173,789	327,499	175,967	228,478	6.20%
CPS 2008-A	October 1, 2014	195,712	310,359	188,723	235,180	7.09%
		\$ 1,015,111	\$ 3,556,063	\$ 1,012,787	\$ 1,404,211	

⁽¹⁾ The Final Scheduled Payment Date represents final legal maturity of the securitization trust debt. Securitization trust debt is expected to become due and to be paid prior to those dates, based on amortization of the finance receivables pledged to the Trusts. Expected payments, which will depend on the performance of such receivables, as to which there can be no assurance, are \$158.1 million in 2009, \$470.3 million in 2010, \$262.6 million in 2011, \$103.1 million in 2012 and \$18.7 million in 2013.

All of the securitization trust debt was sold in private placement transactions to qualified institutional buyers. The debt was issued through our wholly-owned bankruptcy remote subsidiaries and is secured by the assets of

such subsidiaries, but not by our other assets. Principal of \$874.3 million, and the related interest payments, are guaranteed by financial guaranty insupplicies issued by third party financial institutions.	urance

The terms of the various securitization agreements related to the issuance of the securitization trust debt and the warehouse credit facilities require that we meet certain delinquency and credit loss criteria with respect to the collateral pool, and require that we maintain minimum levels of liquidity and net worth and not exceed maximum leverage levels and maximum financial losses. In addition, certain securitization and non-securitization related debt contain cross-default provisions, which would allow certain creditors to declare a default if a default were declared under a different facility. We have received waivers regarding the potential breach of certain such covenants relating to minimum net worth and maintenance of active warehouse credit facilities.

We are responsible for the administration and collection of the automobile contracts. The securitization agreements also require certain funds be held in restricted cash accounts to provide additional collateral for the borrowings or to be applied to make payments on the securitization trust debt. As of September 30, 2009, restricted cash under the various agreements totaled approximately \$132.2 million. Interest expense on the securitization trust debt consists of the stated rate of interest plus amortization of additional costs of borrowing. Additional costs of borrowing include facility fees, insurance and amortization of deferred financing costs and discounts on notes sold. Deferred financing costs and discounts on notes sold related to the securitization trust debt are amortized using a level yield method. Accordingly, the effective cost of the securitization trust debt is greater than the contractual rate of interest disclosed above.

Our wholly-owned bankruptcy remote subsidiaries were formed to facilitate the above asset-backed financing transactions. Similar bankruptcy remote subsidiaries issue the debt outstanding under our warehouse line of credit. Bankruptcy remote refers to a legal structure in which it is expected that the applicable entity would not be included in any bankruptcy filing by its parent or affiliates. All of the assets of these subsidiaries have been pledged as collateral for the related debt. All such transactions, treated as secured financings for accounting and tax purposes, are treated as sales for all other purposes, including legal and bankruptcy purposes. None of the assets of these subsidiaries are available to pay other creditors of ours.

(4) Interest Income

The following table presents the components of interest income:

	Three Mor Septem			Nine Mon Septem		
	 2009		2008	 2009		2008
	 (In the	ousand	ls)	(In tho	usano	ls)
erest on Finance Receivables	\$ 48,813	\$	86,628	\$ 163,737	\$	278,102
esidual interest income	346		134	1,012		493
ther interest income	161		944	710		3,329
nterest income	\$ 49,320	\$	87,706	\$ 165,459	\$	281,924

(5) Earnings (Loss) Per Share

Earnings (loss) per share for the three and nine-month periods ended September 30, 2009 and 2008 were calculated using the weighted average number of shares outstanding for the related period. The following table reconciles the number of shares used in the computations of basic and diluted earnings (loss) per share for the three-month and nine-month periods ended September 30, 2009 and 2008:

	Three Month Septembe		Nine Months Septembe	
	2009	2009 2008		2008
	(In thous	ands)	(In thousa	ands)
Weighted average number of common shares outstanding during				
the period used to compute basic earnings (loss) per share	18,583	19,693	18,776	19,275
Incremental common shares attributable to exercise of				
outstanding options and warrants	-	-	-	-
Weighted average number of common shares used to compute				
diluted earnings (loss) per share	18,583	19,693	18,776	19,275

If the anti-dilutive effects of common stock equivalents were considered, shares included in the diluted earnings (loss) per share calculation for the three-month and nine-month periods ended September 30, 2009 would have included an additional 2.8 million shares attributable to the exercise of outstanding options and warrants. For the three- and nine-month periods ended September 30, 2008, an additional 1.6 million and 949,000 shares, respectively, would have been included that are attributable to outstanding options and warrants.

(6) Income Taxes

We file numerous consolidated and separate income tax returns with the United States and with many states. With few exceptions, we are no longer subject to United States federal income tax examinations for years before 2005 and are no longer subject to state and local income tax examinations by tax authorities for years before 2003.

We have subsidiaries in various states that are currently under audit for years ranging from 1998 through 2005. To date, no material adjustments have been proposed as a result of these audits.

We recognize potential interest and penalties related to unrecognized tax benefits in income tax expense. During the nine months ended September 30, 2009, we recognized \$53,000 of potential interest and penalties, net of a reversal of interest and penalties previously accrued relating to periods no longer subject to examination by taxing authorities. To the extent interest and penalties are not assessed with respect to uncertain tax positions, portions of the amounts accrued will be reversed, and reflected as a reduction of the overall income tax provision.

We do not anticipate that total unrecognized tax benefits will significantly change due to any settlements of audits or expirations of statutes of limitations prior to December 31, 2009.

The Company and its subsidiaries file a consolidated federal income tax return and combined or stand-alone state franchise tax returns for certain states. We utilize the asset and liability method of accounting for income taxes, under which deferred income taxes are recognized for the future tax consequences attributable to the differences between the financial statement values of existing assets and liabilities and their respective tax bases. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred taxes of a change in tax rates is recognized in income in the period that includes the enactment date. We have estimated a valuation allowance against that portion of the deferred tax asset whose utilization in

future periods is not more than likely. Our net deferred tax asset of \$52.7 million as of September 30, 2009 is net of a valuation allowance of \$4.5 million as of September 30, 2009.

On a quarterly basis, we determine whether a valuation allowance is necessary for our deferred tax asset. In performing this analysis, we consider all evidence currently available, both positive and negative, in determining whether, based on the weight of that evidence, the deferred tax asset will be realized. We establish a valuation allowance when it is more likely than not that a recorded tax benefit will not be realized. The expense to create the valuation allowance is recorded as additional income tax expense in the period the valuation allowance is established. During the first nine months of 2009, we increased our valuation allowance by \$3.5 million.

Our estimate of the required valuation allowance is highly dependent upon our estimate of projected levels of future taxable income. Projections carry a degree of uncertainty, particularly for longer-term forecasts. In our case, the amount of uncertainty is increased due to the effects of the economic downturn and estimates for the timing and magnitude of economic recovery, as well as the level of provision for loan losses we have experienced during the economic downturn. Should the actual amount of taxable income be less than what is projected, it may be necessary for us to increase our valuation allowance, either by not reflecting additional deferred tax benefits or by reducing the existing asset. Given the risk of additional deterioration in our loan portfolios not included in our projections, we stress tested those projections and considered the results of those tests in establishing our federal income tax valuation allowance.

Deferred tax assets are recognized subject to management's judgment that realization is more likely than not. Although realization is not assured, we believe that the realization of the recognized net deferred tax asset of \$52.7 million is more likely than not, based on expectations as to future taxable income in the jurisdictions in which we operate and available tax planning strategies that could be implemented if necessary to prevent a carryforward from expiring. Our net deferred tax asset of \$52.7 million is net of a valuation allowance of \$4.5 million and consists of approximately \$48.8 million of net U.S. federal deferred tax assets and \$3.9 million of net state deferred tax assets. The major components of the deferred tax asset are \$27.4 million in net operating loss carryforwards and built in losses, and \$22.2 million in net deductions that have not yet been taken on a tax return. We estimate that we would need to generate approximately \$127.5 million of taxable income during the applicable carryforward periods to realize fully our federal and state deferred tax assets. The federal net operating losses begin to expire in 2028. The state net operating losses begin to expire in 2013.

As a result of the losses incurred in 2008 and 2009, we are in a three-year cumulative pretax loss position as of September 30, 2009. A cumulative loss position is considered significant negative evidence in assessing the realizability of a deferred tax asset. However, we have concluded that there is sufficient positive evidence to overcome this negative evidence. First, relatively recently, from 2003 through 2007, we generated approximately \$107.0 million in taxable income. In addition, we recognized a \$14.0 million loss on the September 2008 securitization that was structured as a sale for financial accounting purposes. Since our inception in 1991, we have completed 49 securitizations of approximately \$6.6 billion in contracts and had never recognized a loss until the September 2008 securitization. We view this securitization as an anomaly created by the unusual and adverse market conditions at the time. Furthermore, our managed portfolio at September 30, 2009 includes approximately \$1.0 billion of finance receivables, virtually all of which are financed with fixed-rate, full term structured securitizations. As such, we believe our managed portfolio can be considered a backlog of future interest income revenue.

Our core business has produced earnings in the past, even with intermittent loss periods resulting from economic cycles not unlike, although not as severe as, the current economic downturn. We believe that our recent losses are largely attributable to the current recessionary economy and heightened unemployment together with temporary disruptions in the capital markets, rather than to a fundamental flaw in our business model. We forecast sufficient taxable income in the carryforward period to realize fully our deferred tax assets, exclusive of tax planning strategies, even under stressed scenarios. Regarding the estimate of future taxable income, we have projected pretax earnings based upon our core business that we intend to conduct going forward. We have taken steps to reduce our cost structure and have adjusted the contract interest rates and purchase prices applicable to our purchases of automobile contracts from dealers. We appear to be able to

increase our acquisition fees and reduce our purchase prices because of lessened competition for our services. Our forecasts do assume, however, that we are able to obtain warehouse and long-term financing so that we may again originate significant volumes of new receivables. Since December 2008, we have seen a significant recovery in the asset-backed securitization market, which has been stimulated by the government's Term Asset-Backed Securities Loan Facility ("TALF"). There can be no assurance, however, that we will in fact be able to obtain such financing. Taking these items into account, we project generating sufficient pretax earnings within the carryforward period to realize our deferred tax assets.

Nevertheless, the amount of the deferred tax asset considered realizable, however, could be significantly reduced in the future if adverse developments cause us to lower our estimates of future taxable income during the carryforward period. Based upon the foregoing discussion, as well as tax planning opportunities and other factors discussed below, we have concluded that the U.S. and state net operating loss carryforward periods provide enough time to utilize the deferred tax assets pertaining to the existing net operating loss carryforwards and any net operating loss that would be created by the reversal of the future net deductions which have not yet been taken on a tax return. We have also examined tax planning strategies available to us in accordance with FASB ASC 740 which would be employed, if necessary, to prevent a carryforward from expiring. Our projection of sufficient earnings is a forward-looking statement, and there can be no assurance that our projections of such earnings will be correct.

(7) Legal Proceedings

Stanwich Litigation. We were for some time a defendant in a class action (the "Stanwich Case") brought in the California Superior Court, Los Angeles County. The original plaintiffs in that case were persons entitled to receive regular payments (the "Settlement Payments") under out-of-court settlements reached with third party defendants. Stanwich Financial Services Corp. ("Stanwich"), then an affiliate of our former chairman of the board of directors, is the entity that was obligated to pay the Settlement Payments. Stanwich had defaulted on its payment obligations to the plaintiffs and in September 2001 filed for reorganization under the Bankruptcy Code, in the federal Bankruptcy Court of Connecticut. By February 2005, we had settled all claims brought against us in the Stanwich Case.

In November 2001, one of the defendants in the Stanwich Case, Jonathan Pardee, asserted claims for indemnity against us in a separate action, which is now pending in federal district court in Rhode Island. We have filed counterclaims in the Rhode Island federal court against Mr. Pardee, and have filed a separate action against Mr. Pardee's Rhode Island attorneys, in the same court. Each of these actions in the court in Rhode Island is stayed, awaiting resolution of an adversary action brought against Mr. Pardee in the bankruptcy court, which is hearing the bankruptcy of Stanwich.

We had reached an agreement in principle with the representative of creditors in the Stanwich bankruptcy to resolve the adversary action. Under the agreement in principle, we were to pay the bankruptcy estate \$625,000 and abandon our claims against the estate, while the estate would abandon its adversary action against Mr. Pardee. The bankruptcy court has rejected that proposed settlement, the representative of creditors has appealed that rejection, and the appeal was denied on April 8, 2009. There can be no assurance as to the ultimate outcome of these matters.

The reader should consider that any adverse judgment against us in these cases for indemnification, in an amount materially in excess of any liability already recorded in respect thereof, could have a material adverse effect on our financial position.

Other Litigation.

We are routinely involved in various legal proceedings resulting from our consumer finance activities and practices, both continuing and discontinued. We believe that there are substantive legal defenses to such claims, and intend to defend them vigorously. There can be no assurance, however, as to the outcome.

We have recorded a liability as of September 30, 2009 that we believe represents a sufficient allowance for legal contingencies, including those described above. Any adverse judgment against us, if in an amount materially in excess of the recorded liability, could have a material adverse effect on our financial position.

(8) Employee Benefits

We sponsor the MFN Financial Corporation Benefit Plan (the "Plan"). Plan benefits were frozen September 30, 2001. The table below sets forth the Plan's net periodic benefit cost for the three-month and nine-month periods ended September 30, 2009 and 2008.

	Three Months Ended			Nine Months Ended			ed
	Septem	ber 30,			September 30,		
	2009	20)8	2009		20	800
	(In thou	isands)		(In thousands)			
Components of net periodic cost (benefit)							
Service cost	\$ -	\$	-	\$	-	\$	-
Interest Cost	237		237		711		710
Expected return on assets	(175)		(305)		(525)		(915)
Amortization of transition (asset)/obligation	-		-		-		-
Amortization of net (gain) / loss	169		41		507		123
Net periodic cost (benefit)	\$ 231	\$	(27)	\$	693	\$	(82)

We did not make any contributions to the Plan during the nine-month period ended September 30, 2009 and we do not anticipate making any contributions for the remainder of 2009.

(9) Fair Value Measurements

In September 2006, the FASB issued SFAS No. 157, "Fair Value Measurements" ("SFAS No. 157") (ASC 820 10 65). SFAS No. 157 (ASC 820 10 65) clarifies the principle that fair value should be based on the assumptions market participants would use when pricing an asset or liability and establishes a fair value hierarchy that prioritizes the information used to develop those assumptions. Under the standard, fair value measurements would be separately disclosed by level within the fair value hierarchy.

SFAS No. 157 (ASC 820 10 65) defines fair value, establishes a framework for measuring fair value, establishes a three-level valuation hierarchy for disclosure of fair value measurement and enhances disclosure requirements for fair value measurements. The three levels are defined as follows: level 1 - inputs to the valuation methodology are quoted prices (unadjusted) for identical assets or liabilities in active markets; level 2 - inputs to the valuation methodology include quoted prices for similar assets and liabilities in active markets, and inputs that are observable for the asset or liability, either directly or indirectly, for substantially the full term of the financial instrument; and level 3 - inputs to the valuation methodology are unobservable and significant to the fair value measurement.

In September 2008 we sold automobile contracts in a securitization that was structured as a sale for financial accounting purposes. In that sale, we retained certain assets that are measured at fair value. We describe below the valuation methodologies we use for the securities retained and the residual interest in the cash flows of the transaction, as well as the general classification of such instruments pursuant to the valuation hierarchy. The securities retained are \$8.5 million of notes, which are classified as level 2 because we sold similar assets in the transaction. We use the price at which those similar notes were sold to value the securities retained. The residual interest in such securitization is \$2.5 million and is classified as level 3. We determine the value of that residual interest using a discounted cash flow model that includes estimates for prepayments and losses. We use a discount rate of 33% per annum. The assumptions we use are based on historical performance of automobile contracts we have originated and serviced in the past, adjusted for current market conditions.

Repossessed vehicle inventory, which is included in Other Assets on our balance sheet, is measured at fair value using Level 2 assumptions based on our actual loss experience on sale of repossessed vehicles. At September 30, 2009, the finance receivables related to the repossessed vehicles in inventory totaled \$32.8 million. We have applied a valuation adjustment of \$25.0 million, resulting in an estimated fair value and carrying amount of \$7.8 million.

The table below presents a reconciliation for Level 3 assets measured at fair value on a recurring basis using significant inputs::

	Inte	sidual erest in itizations
Balance at January 1, 2009	\$	3,582
Transfers into Level 3		-
Included in earnings		562
Balance at September 30, 2009	\$	4,144

The following summary presents a description of the methodologies and assumptions used to estimate the fair value of our financial instruments. Much of the information used to determine fair value is highly subjective. When applicable, readily available market information has been utilized. However, for a significant portion of our financial instruments, active markets do not exist. Therefore, considerable judgments were required in estimating fair value for certain items. The subjective factors include, among other things, the estimated timing and amount of cash flows, risk characteristics, credit quality and interest rates, all of which are subject to change. Since the fair value is estimated as of September 30, 2009 and December 31, 2008, the amounts that will actually be realized or paid at settlement or maturity of the instruments could be significantly different. The estimated fair values of financial assets and liabilities at September 30, 2009 and December 31, 2008, were as follows:

	September 30, 2009		Decembe	er 31, 2008		
	C	arrying	Fair	Carrying		Fair
Financial Instrument		Value	Value	Value		Value
			(In tho	usands)		
Cash and cash equivalents	\$	20,396	\$ 20,396	\$ 22,084	\$	22,084
Restricted cash and equivalents		132,236	132,236	153,479		153,479
Finance receivables, net		976,013	937,872	1,339,307		1,312,148
Residual interest in securitizations		4,144	4,144	3,582		3,582
Accrued interest receivable		10,193	10,193	14,903		14,903
Warehouse lines of credit		5,171	5,171	9,919		9,919
Accrued interest payable		4,517	4,517	5,605		5,605
Residual interest financing		59,850	59,850	67,300		67,300
Securitization trust debt		1,012,787	1,006,632	1,404,211		1,378,271
Senior secured debt		20,842	20,842	20,105		20,105
Subordinated renewable notes		21,232	21,232	25,721		25,721

Cash, Cash Equivalents and Restricted Cash

The carrying value equals fair value.

Finance Receivables, net

The fair value of finance receivables is estimated by discounting future cash flows expected to be collected using current rates at which similar receivables could be originated.

Residual Interest in Securitizations

The fair value is estimated by discounting future cash flows using credit and discount rates that we believe reflect the estimated credit, interest rate and prepayment risks associated with similar types of instruments.

Accrued Interest Receivable and Payable

The carrying value approximates fair value because the related interest rates are estimated to reflect current market conditions for similar types of instruments.

Warehouse Lines of Credit, Residual Interest Financing, and Senior Secured Debt and Subordinated Renewable Notes

The carrying value approximates fair value because the related interest rates are estimated to reflect current market conditions for similar types of secured instruments.

Securitization Trust Debt

The fair value is estimated by discounting future cash flows using interest rates that we believe reflects the current market rates.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Overview

We are a specialty finance company. Our business is to purchase and service retail automobile contracts originated primarily by franchised automobile dealers and, to a lesser extent, by select independent dealers in the United States in the sale of new and used automobiles, light trucks and passenger vans. Through our automobile contract purchases, we provide indirect financing to the customers of dealers who have limited credit histories, low incomes or past credit problems, who we refer to as sub-prime customers. We serve as an alternative source of financing for dealers, facilitating sales to customers who otherwise might not be able to obtain financing from traditional sources, such as commercial banks, credit unions and the captive finance companies affiliated with major automobile manufacturers. In addition to purchasing installment purchase contracts directly from dealers, we have also (i) acquired installment purchase contracts in three merger and acquisition transactions, (ii) purchased immaterial amounts of vehicle purchase money loans from non-affiliated lenders, and (iii) originated ourselves an immaterial amount of vehicle purchase money loans by lending money directly to consumers. In this report, we refer to all of such contracts and loans as "automobile contracts."

We were incorporated and began our operations in March 1991. From inception through September 30, 2009, we have purchased a total of approximately \$8.7 billion of automobile contracts from dealers. In addition, we obtained a total of approximately \$605.0 million of automobile contracts in mergers and acquisitions we made in 2002, 2003 and 2004. Unlike recent prior years, our managed portfolio decreased in 2009 from the previous year due to our strategy of decreasing contract purchases to conserve our liquidity in response to adverse economic conditions as discussed further below. Our total managed portfolio, net of unearned interest on pre-computed automobile contracts, was approximately \$1,197.3 million at September 30, 2009 compared to \$1,829.5 million at September 30, 2008.

We are headquartered in Irvine, California, where most operational and administrative functions are centralized. All credit and underwriting functions are performed in our California headquarters, and we service our automobile contracts from our California headquarters and from three servicing branches in Virginia, Florida and Illinois.

We purchase contracts in our own name ("CPS") and, until July 2008, also in the name of our wholly-owned subsidiary, TFC. Programs marketed under the CPS name are intended to serve a wide range of sub-prime customers, primarily through franchised new car dealers. Our TFC program served vehicle purchasers enlisted in the U.S. Armed Forces, primarily through independent used car dealers. In July 2008, we suspended contract purchases under our TFC program.

We purchase automobile contracts with the intention of financing them on a long-term basis through securitizations. Securitizations are transactions in which we sell a specified pool of contracts to a special purpose entity of ours, which in turn issues asset-backed securities to fund the purchase of the pool of contracts from us. Depending on the structure of the securitization, the transaction may be treated, for financial accounting purposes, as a sale of the contracts or as a secured financing.

Securitization and Warehouse Credit Facilities

Throughout the period for which information is presented in this report, we have purchased automobile contracts with the intention of financing them on a long-term basis through securitizations, and on an interim basis through our warehouse credit facilities. All such financings have involved identification of specific automobile contracts, sale of those automobile contracts (and associated rights) to one of our special-purpose subsidiaries, and issuance of asset-backed securities to fund the transactions. Depending on the structure, these transactions may properly be accounted for under generally accepted accounting principles as sales of the automobile contracts or as secured financings.

When structured to be treated as a secured financing for accounting purposes, the subsidiary is consolidated with us. Accordingly, the sold automobile contracts and the related debt appear as assets and liabilities, respectively, on our consolidated balance sheet. We then periodically (i) recognize interest and fee income on

the contracts, (ii) recognize interest expense on the securities issued in the transaction and (iii) record as expense a provision for credit losses on the contracts.

Since the third quarter of 2003, we have conducted 24 term securitizations. Of these 24, 19 were periodic (generally quarterly) securitizations of automobile contracts that we purchased from automobile dealers under our regular programs. In addition, in March 2004 and November 2005, we completed securitizations of our retained interests in other securitizations that we and our affiliates previously sponsored. The debt from the March 2004 transaction was repaid in August 2005, and the debt from the November 2005 transaction was repaid in May 2007. Also, in June 2004, we completed a securitization of automobile contracts purchased in the SeaWest asset acquisition and under our TFC programs. Further, in December 2005 and May 2007 we completed securitizations that included automobile contracts purchased under the TFC programs, automobile contracts purchased under the CPS programs and automobile contracts we repurchased upon termination of prior securitizations of our MFN and TFC subsidiaries. Since July 2003 all such securitizations have been structured as secured financings, except that our September 2008 securitization was in substance a sale of the underlying receivables, and is treated as a sale for financial accounting purposes.

Uncertainty of Capital Markets and General Economic Conditions

Historically, we have depended upon the availability of warehouse credit facilities and access to long-term financing through the issuance of asset-backed securities collateralized by our automobile contracts. Since 1994, we have completed 49 term securitizations of approximately \$6.6 billion in contracts. We conducted four term securitizations in 2006, four in 2007, and two in 2008. From July 2003 through April 2008 all of our securitizations were structured as secured financings. The second of our two securitization transactions in 2008 (completed in September 2008) was in substance a sale of the related contracts, and is treated as a sale for financial accounting purposes.

Since the fourth quarter of 2007, we have observed unprecedented adverse changes in the market for securitized pools of automobile contracts. These changes include reduced liquidity, and reduced demand for asset-backed securities, particularly for securities carrying a financial guaranty and for securities backed by sub-prime automobile receivables. Moreover, many of the firms that previously provided financial guarantees, which were an integral part of our securitization program, are no longer offering such guarantees. From December 2008 until September 25, 2009 we had no available funding facilities. On September 25, 2009 we entered into a \$50 million two-year credit facility. Unlike our recent former credit facilities which limited the timeframe for eligible collateral to 180 days, collateral can remain eligible in this facility for the entire two-year term. Consequently, we have no immediate plans to complete a term securitization. The adverse changes that have taken place in the market have caused us to seek to conserve liquidity by reducing our purchases of automobile contracts to nominal levels. If the current adverse circumstances that have affected the capital markets should continue or worsen, we may curtail further or cease our purchases of new automobile contracts, which could lead to a material adverse effect on our operations.

Current economic conditions have negatively affected many aspects of our industry. First, as stated above, there is reduced demand for asset-backed securities secured by consumer finance receivables, including sub-prime automobile receivables. Second, lenders who previously provided short-term warehouse financing for sub-prime automobile finance companies such as ours are reluctant to provide such short-term financing due to the uncertainty regarding the prospects of obtaining long-term financing through the issuance of asset-backed securities. In addition, many capital market participants such as investment banks, financial guaranty providers and institutional investors who previously played a role in the sub-prime auto finance industry have withdrawn from the industry, or in some cases, have ceased to do business. Finally, the broad economic weakness and increasing unemployment has made many of the obligors under our receivables less willing or able to pay, resulting in higher delinquency, charge-offs and losses. Each of these factors has adversely affected our results of operations. Should existing economic conditions worsen, both our ability to purchase new contracts and the performance of our existing managed portfolio may be impaired, which, in turn, could have a further material adverse effect on our results of operations.

Financial Covenants

Certain of our securitization transactions and our warehouse credit facility contain various financial covenants requiring certain minimum financial ratios and results. Such covenants include maintaining minimum levels of liquidity and net worth and not exceeding maximum leverage levels and maximum financial losses. In addition, certain securitization and non-securitization related debt contain cross-default provisions that would allow certain creditors to declare a default if a default occurred under a different facility.

The agreements under which we receive periodic fees for servicing automobile contracts in securitizations are terminable by the respective financial guaranty insurance companies (also referred to as note insurers) upon defined events of default, and, in some cases, at the will of the insurance company. Without the waivers we have received from the related note insurers, we would have been in violation of certain financial and operating covenants relating to minimum net worth and maintenance of active warehouse facilities with respect to eight of our 17 currently outstanding securitization transactions. Upon such an event of default, and subject to the right of the related note insurers to waive such terms, the agreements governing the securitizations call for payment of a default insurance premium, ranging from 25 to 100 basis points per annum on the aggregate outstanding balance of the related insured senior notes, and for the diversion of all excess cash generated by the assets of the respective securitization pools into the related spread accounts to increase the credit enhancement associated with those transactions. The cash so diverted into the spread accounts would otherwise be used to make principal payments on the subordinated notes in each related securitization or would be released to us. In addition, upon an event of default, the note insurers have the right to terminate us as servicer. Although the diversion of such cash and our termination as servicer have been waived, we are paying default premiums, or their equivalent, with respect to insured notes representing \$593.6 million of the \$1,012.8 million of securitization trust debt outstanding at September 30, 2009. It should be noted that the principal amount of such securitization trust debt is not increased, but that the increased insurance premium is reflected as increased interest expense. Furthermore, such waivers are temporary, and there can be no assurance as to their future extension. We do, however, believe that we will obtain such future extensions of our servicing agreements because it is generally not in the interest of any party to the securitization transaction to transfer servicing. Nevertheless, there can be no assurance as to our belief being correct. Were an insurance company in the future to exercise its option to terminate such agreements or to pursue other remedies, such remedies could have a material adverse effect on our liquidity and results of operations, depending on the number and value of the affected transactions. Our note insurers continue to extend our term as servicer on a monthly and/or quarterly basis, pursuant to the servicing agreements.

Results of Operations

Comparison of Operating Results for the three months ended September 30, 2009 with the three months ended September 30, 2008

Revenues. During the three months ended September 30, 2009, revenues were \$52.8 million, a decrease of \$38.9 million, or 42.4%, from the prior year revenue of \$91.7 million. The primary reason for the decrease in revenues is a decrease in interest income. Interest income for the three months ended September 30, 2009 decreased \$38.4 million, or 43.8%, to \$49.3 million from \$87.7 million in the prior year. The primary reason for the decrease in interest income is the decrease in finance receivables held by consolidated subsidiaries.

Servicing fees totaling \$839,000 in the three months ended September 30, 2009 increased \$604,000, or 257.0%, from \$235,000 in the prior year. The increase in servicing fees is the result of our September 2008 securitization that was structured as a sale for financial accounting purposes and on which we earn a base servicing fee. During 2008 we also earned base servicing fees on a portfolio which we have serviced for SeaWest Financial Corporation since April 2004, which has declined to an immaterial amount as of September 30, 2009. As of September 30, 2009 and 2008, our managed portfolio owned by consolidated vs. non-consolidated subsidiaries and other third parties was as follows:

		September 3	30, 2009		September	30, 2008	
	A	mount	%		Amount	%	
Total Managed Portfolio			(\$ in mil	lion	s)		
Owned by Consolidated Subsidiaries	\$	1,049.3	87.6%	\$	1,633.8	89.3%	
Owned by Non-Consolidated Subsidiaries		148.0	12.4%		195.6	10.7%	
Third Party Portfolio		-	0.0%		0.1	0.0%	
Total	\$	1,197.3	100.0%	\$	1,829.5	100.0%	

At September 30, 2009, we were generating income and fees on a managed portfolio with an outstanding principal balance of \$1,197.3 million (this amount includes \$148.0 million of automobile contracts on which we earn servicing fees, own 5.0% of the asset-backed notes issued by the related trust, and own a residual interest), compared to a managed portfolio with an outstanding principal balance of \$1,829.5 million as of September 30, 2008. At September 30, 2009 and 2008, the managed portfolio composition was as follows:

		September 3	0, 2009	September 30, 2008		
	A	mount	%	Amount	%	
Originating Entity			(\$ in million	s)		
CPS	\$	1,169.4	97.7% \$	1,777.1	97.1%	
TFC		27.9	2.3%	52.1	2.8%	
MFN		-	0.0%	-	0.0%	
SeaWest		-	0.0%	0.2	0.0%	
Third Party Portfolio		-	0.0%	0.1	0.0%	
Total	\$	1,197.3	100.0% \$	1,829.5	100.0%	

Other income decreased by \$1.1 million, or 29.5%, to \$2.7million in the three months ended September 30, 2009 from \$6.8 million during the prior year. Other income consists primarily of convenience fees charged to our borrowers for certain electronic payments, fees paid to us by dealers for training, potential customer targeting and certain direct mail products that we offer, and recoveries on portfolios that we previously acquired through acquisitions.

Expenses. Our operating expenses consist largely of provision for credit losses, interest expense, employee costs and general and administrative expenses. Provision for credit losses and interest expense are significantly affected by the volume of automobile contracts we purchased during a period and by the outstanding balance of finance receivables held by consolidated subsidiaries. Employee costs and general and administrative expenses are incurred as applications and automobile contracts are received, processed and serviced. Factors that affect margins and net income (loss) include changes in the automobile and automobile finance market environments, and macroeconomic factors such as interest rates and the unemployment level.

Employee costs include base salaries, commissions and bonuses paid to employees, and certain expenses related to the accounting treatment of outstanding stock options, and are one of our most significant operating expenses. These costs (other than those relating to stock options) generally fluctuate with the level of applications and automobile contracts processed and serviced.

Other operating expenses consist largely of facilities expenses, telephone and other communication services, credit services, computer services, marketing and advertising expenses, and depreciation and amortization.

Total operating expenses were \$57.1 million for the three months ended September 30, 2009, compared to \$104.4 million for the prior year, a decrease of \$47.2 million, or 45.3%. The decrease is primarily due to the continued decline in the balance of our outstanding managed portfolio and the related costs to service it, plus reduced expenses associated with significantly lower volumes of new contract purchases compared to the prior period. Operating expenses for the three-month period ended September 30, 2008 included a loss of \$14.0 million resulting from the sale of \$198.7 million of our receivables to a third party.

Employee costs decreased by \$4.2 million, or 34.0%, to \$8.2 million during the three months ended September 30, 2009, representing 14.4% of total operating expenses, from \$12.5 million for the prior year, or 11.9% of total operating expenses. Since January 2008, we have reduced staff through attrition and reductions in force as a result of the uncertainty in capital markets and the related limited access to financing for new purchases of automobile contracts. At September 30, 2009 we had 489 employees compared to 753 employees at September 30, 2008.

General and administrative expenses include costs associated with purchasing and servicing our portfolio of finance receivables including expenses for facilities, credit services, and telecommunications. General and administrative expenses were \$5.6 million, a decrease of 25.7%, compared to the previous year and represented 9.7% of total operating expenses.

Interest expense for the three months ended September 30, 2009 decreased \$14.8 million, or 36.1%, to \$26.2 million, compared to \$41.0 million in the previous year. The decrease is primarily the result of changes in the amount and composition of securitization trust debt carried on our consolidated balance sheet. Interest on securitization trust debt decreased by \$11.1 million in the three months ended September 30, 2009 compared to the prior year. Interest expense on residual interest financing decreased \$893,000 in the three months ended September 30, 2009 compared to the prior year. The outstanding balances of our aggregate subordinated debt and our residual interest debt have been decreased through repayments since the prior year period. Interest expense on warehouse debt decreased by \$2.6 million for the three months ended September 30, 2009 compared to the prior year. In the prior year period, we had access to two warehouse credit facilities totaling \$400 million in financing capacity. During that period we were actively purchasing new contracts and financing such purchases with these facilities that had, in the aggregate significant outstanding balances during the three-month period ended September 30, 2008. As of September 30, 2009, our newly established credit facility had an outstanding balance of \$5.2 million.

Provision for credit losses was \$15.3 million for the three months ended September 30, 2009, a decrease of \$10.7 million, or 70.1% compared to the prior year and represented 26.7% of total operating expenses. The provision for credit losses maintains the allowance for loan losses at levels that we feel are adequate for probable credit losses that can be reasonable estimated. The decrease in provision expense is the result of the decrease in the size of the portfolio and the smaller volumes of new contract purchases in the current period compared to the prior period.

Marketing expenses consist primarily of commission-based compensation paid to our employee marketing representatives. Our marketing representatives earn salary and commissions based on our volume of contract purchases and also on sales of training programs, potential customer targeting, and direct mail products that we offer our dealers. Marketing expenses decreased by \$1.5 million, or 63.0%, to \$895,000, compared to \$2.4 million in the previous year, and represented 1.6% of total operating expenses. The decrease is primarily due to the decrease in automobile contracts we purchased during the three months ended September 30, 2009 as compared to the prior year. During the three months ended September 30, 2009, we purchased 37 automobile contracts aggregating \$506,000, compared to 2,227 automobile contracts aggregating \$33.6 million in the prior year. The adverse changes that have taken place in the securitization market since the fourth quarter of 2007 have caused us to curtail our purchases of automobile contracts in order to preserve liquidity.

Occupancy expenses decreased by \$147,000 or 15.1%, to \$825,000 compared to \$972,000 in the previous year and represented 1.5% of total operating expenses.

Depreciation and amortization expenses increased \$55,000, or 43.9%, to \$182,000 from \$126,000 in the previous year.

For the three months ended September 30, 2009, we recorded no tax provision or benefit. As of September 30, 2009, our net deferred tax asset of \$52.7 million is net of a valuation allowance of \$4.5 million and consists of approximately \$48.8 million of net U.S. federal deferred tax assets and \$3.9 million of net state deferred tax assets. The major components of the deferred tax asset are \$27.4 million in net operating loss carryforwards and built in losses and \$22.2 million in net deductions which have not yet been taken on a tax return. We have considered the circumstances that may affect the ultimate realization of our deferred tax assets and have

concluded that the valuation allowance is appropriate at this time. However, if future events change our expected realization of our deferred tax assets, we may be required to increase the valuation allowance against that asset in the future.

Comparison of Operating Results for the nine months ended September 30, 2009 with the nine months ended September 30, 2008

Revenues. During the nine months ended September 30, 2009, revenues were \$177.2 million, a decrease of \$116.6 million, or 39.7%, from the prior year revenue of \$293.8 million. The primary reason for the decrease in revenues is a decrease in interest income. Interest income for the nine months ended September 30, 2009 decreased \$116.5 million, or 41.3%, to \$165.5 million from \$281.9 million in the prior year. The primary reason for the decrease in interest income is the decrease in finance receivables held by consolidated subsidiaries.

Servicing fees totaling \$2.8 million in the nine months ended September 30, 2009 increased \$1.9 million, or 197.9%, from \$943,000 in the prior year. The increase in servicing fees is the result of our September 2008 securitization that was structured as a sale for financial accounting purposes and on which we earn a base servicing fee. During 2008 we also earned base servicing fees on a portfolio which we have serviced for SeaWest Financial Corporation since April 2004, which has declined to an immaterial amount as of September 30, 2009. As of September 30, 2009 and 2008, our managed portfolio owned by consolidated vs. non-consolidated subsidiaries and other third parties was as follows:

	September 30, 2009			Septembe	er 30, 2008
	A	mount	%	Amount	%
Total Managed Portfolio			(\$ in mi	illions)	
Owned by Consolidated Subsidiaries	\$	1,049.3	87.6%	\$ 1,633.8	89.3%
Owned by Non-Consolidated Subsidiaries		148.0	12.4%	195.6	10.7%
Third Party Portfolio		-	0.0%	0.1	0.0%
Total	\$	1,197.3	100.0%	\$ 1,829.5	100.0%

At September 30, 2009, we were generating income and fees on a managed portfolio with an outstanding principal balance of \$1,197.3 million (this amount includes \$148.0 million of automobile contracts on which we earn servicing fees, own 5.0% of the asset-backed notes issued by the related trust, and own a residual interest), compared to a managed portfolio with an outstanding principal balance of \$1,979.5 million as of September 30, 2008. At September 30,

		September 3	0, 2009	September	30, 2008
	F	Amount	%	Amount	%
Originating Entity			(\$ in millio	ons)	
CPS	\$	1,169.4	97.7% \$	1,777.1	97.1%
TFC		27.9	2.3%	52.1	2.8%
MFN		-	0.0%	-	0.0%
SeaWest		-	0.0%	0.2	0.0%
Total	\$	1,197.3	100.0% \$	1,829.5	100.0%

Other income decreased by \$2.0 million, or 18.4%, to \$8.9 million in the nine months ended September 30, 2009 from \$10.9 million during the prior year. Other income consists primarily of convenience fees charged to our borrowers for certain electronic payments, fees paid to us by dealers for training, potential customer targeting and certain direct mail products that we offer, and recoveries on portfolios that we previously acquired through acquisitions.

Expenses. Our operating expenses consist largely of provision for credit losses, interest expense, employee costs and general and administrative expenses. Provision for credit losses and interest expense are significantly affected by the volume of automobile contracts we purchased during a period and by the outstanding balance of finance receivables held by consolidated subsidiaries. Employee costs and general and administrative expenses are incurred as applications and automobile contracts are received, processed and serviced. Factors that affect margins and net income (loss) include changes in the automobile and automobile finance market environments, and macroeconomic factors such as interest rates and the unemployment level.

Employee costs include base salaries, commissions and bonuses paid to employees, and certain expenses related to the accounting treatment of outstanding stock options, and are one of our most significant operating expenses. These costs (other than those relating to stock options) generally fluctuate with the level of applications and automobile contracts processed and serviced.

Other operating expenses consist largely of facilities expenses, telephone and other communication services, credit services, computer services, marketing and advertising expenses, and depreciation and amortization.

Total operating expenses were \$188.0 million for the nine months ended September 30, 2009, compared to \$300.0 million for the prior year, a decrease of \$112.0 million, or 37.3%. The decrease is primarily due to the continued decline in the balance of our outstanding managed portfolio and the related costs to service it, plus reduced expenses associated with significantly lower volumes of new contract purchases compared to the prior period.

Employee costs decreased by \$12.4 million, or 31.8%, to \$26.5 million during the nine months ended September 30, 2009, representing 14.1% of total operating expenses, from \$38.8 million for the prior year, or 12.9% of total operating expenses. Since January 2008, we have reduced staff through attrition and reductions in force as a result of the uncertainty in capital markets and the related limited access to financing for new purchases of automobile contracts. At September 30, 2009 we had 489 employees compared to 753 employees at September 30, 2008.

General and administrative expenses include costs associated with purchasing and servicing our portfolio of finance receivables including expenses for facilities, credit services, and telecommunications. General and administrative expenses were \$18.0 million, a decrease of 19.6%, compared to the previous year and represented 9.6% of total operating expenses.

Interest expense for the nine months ended September 30, 2009 decreased \$33.7 million, or 27.8%, to \$87.3 million, compared to \$121.0 million in the previous year. The decrease is primarily the result of changes in the amount and composition of securitization trust debt carried on our consolidated balance sheet. Interest on securitization trust debt decreased by \$25.3 million in the nine months ended September 30, 2009 compared to the prior year. Interest expense on senior secured and subordinated debt increased by \$2.5 million as a result of our issuance in June 2008 and July 2008 of senior secured notes of \$10.0 million \$15.0 million, respectively. Interest expense on residual interest financing decreased \$1.2 million in the nine months ended September 30, 2009 compared to the prior year. Interest expense on warehouse debt decreased by \$9.6 million for the nine months ended September 30, 2009 compared to the prior year. In the prior year period, we had access to two warehouse credit facilities totaling \$400 million in financing capacity. During that period we were actively purchasing new contracts and financing such purchases with these facilities that had, in the aggregate, significant outstanding balances during the nine-month period ended September 30, 2008. As of September 30, 2009, our newly established credit facility had an outstanding balance of \$5.2 million.

As stated above, we have issued \$25.0 million in new senior secured debt since June 2008. In addition, in July 2008 we amended our existing residual financing facility resulting in a higher interest rate. As a result, we can expect that our interest expense on these components of debt will increase in future periods although such increases may be somewhat offset by decreases in our securitization trust and warehouse debt should our level of contract purchases and our portfolio of managed receivables continue to decline.

Provision for credit losses was \$49.8 million for the nine months ended September 30, 2009, a decrease of \$41.9 million, or 45.7% compared to the prior year and represented 26.5% of total operating expenses. The provision for credit losses maintains the allowance for loan losses at levels that we feel are adequate for

probable credit losses that can be reasonable estimated. The decrease in provision expense is the result of the decrease in the size of the portfolio and the smaller volumes of new contract purchases in the current period compared to the prior period.

Marketing expenses consist primarily of commission-based compensation paid to our employee marketing representatives. Our marketing representatives earn salary and commissions based on our volume of contract purchases and also on sales of training programs, potential customer targeting, and direct mail products that we offer our dealers. Marketing expenses decreased by \$5.7 million, or 65.6%, to \$3.0 million, compared to \$8.7 million in the previous year, and represented 1.6% of total operating expenses. The decrease is primarily due to the decrease in automobile contracts we purchased during the nine months ended September 30, 2009 as compared to the prior year. During the nine months ended September 30, 2009, we purchased 191 automobile contracts aggregating \$2.5 million, compared to 19,278 automobile contracts aggregating \$289.6 million in the prior year. The adverse changes that have taken place in the securitization market since the fourth quarter of 2007 have caused us to curtail our purchases of automobile contracts in order to preserve liquidity.

Occupancy expenses were \$2.8 million for the nine months ended September 30, 2009 and were basically unchanged from the prior period.

Depreciation and amortization expenses increased \$162,000, or 44.6%, to \$526,000 from \$364,000 in the previous year.

For the nine months ended September 30, 2009, we recorded no tax provision or benefit. As of September 30, 2009, our net deferred tax asset of \$52.7 million is net of a valuation allowance of \$4.5 million and consists of approximately \$48.8 million of net U.S. federal deferred tax assets and \$3.9 million of net state deferred tax assets. The major components of the deferred tax asset are \$27.4 million in net operating loss carryforwards and built in losses and \$22.2 million in net deductions which have not yet been taken on a tax return. We have considered the circumstances that may affect the ultimate realization of our deferred tax assets and have concluded that the valuation allowance is appropriate at this time. However, if future events change our expected realization of our deferred tax assets, we may be required to increase the valuation allowance against that asset in the future.

Credit Experience

Our financial results are dependent on the performance of the automobile contracts in which we retain an ownership interest. The table below documents the delinquency, repossession and net credit loss experience of all automobile contracts that we were servicing as of the respective dates shown. Credit experience for CPS, MFN, TFC and SeaWest (the three acquisitions described above involved MFN Financial Corporation, TFC Enterprises, Inc. and SeaWest Financial Corporation and were completed in 2002, 2003 and 2004, respectively) is shown on a combined basis in the table below. While the broad economic weakness and increasing unemployment over the last year has resulted in higher delinquencies and net charge-offs compared to the same period last year, the increase in the percentage levels for 2009 is also partially attributable to the decrease in the size and the increase in the average age of our managed portfolio.

Delinquency Experience (1) CPS, MFN, TFC and SeaWest Combined

	Septeml	er 3	80, 2009	September 30, 2008			Decemb	er 3	1, 2008
	Number of			Number of			Number of		
	Contracts		Amount	Contracts		Amount	Contracts		Amount
			_	(Dollars	in th	nousands)			
Delinquency Experience									
Gross servicing portfolio (1)	119,721	\$	1,197,706	154,516	\$	1,830,727	145,564	\$	1,665,036
Period of delinquency (2)									
31-60 days	3,266		30,347	4,971		55,373	3,733		39,798
61-90 days	2,102		20,860	2,127		23,379	2,376		26,549
91+ days	1,754		18,301	1,558		16,521	2,424		27,243
Total delinquencies (2)	7,122		69,508	8,656		95,273	8,533		93,590
Amount in repossession (3)	4,062		36,204	3,963		45,312	4,262		49,357
Total delinquencies and amount in repossession (2)	11,184	\$	105,712	12,619	\$	140,585	12,795	\$	142,947
Delinquencies as a percentage of gross servicing portfolio	. 5.9%		5.8%	5.6%		5.2%	5.9%		5.6%
Total deliquencies and amoun in repossession as									
a perscentage of gross sevicing portfolio	9.3%		8.8%	8.2%		7.7%	8.8%		8.6%

⁽¹⁾ All amounts and percentages are based on the amount remaining to be repaid on each automobile contract, including, for pre-computed automobile contracts, any unearned interest. The information in the table represents the gross principal amount of all automobile contracts purchased by us, including automobile contracts subsequently sold by us in securitization transactions that we continue to service. The table does not include automobile contracts from the portfolio we have serviced for SeaWest Financial Corporation since 2004.

Net Charge-Off Experience (1) CPS, MFN, TFC and SeaWest Combined

	September 30, 2009	September 30, 2008	December 31, 2008
		(Dollars in thousands)	
Average servicing portfolio outstanding	\$ 1,390,959	2,005,504	\$ 1,934,003
Annualized net charge-offs as a percentage of			
average servicing portfolio (2)	\$ 10.4%	7.1%	7.7%

⁽¹⁾ All amounts and percentages are based on the principal amount scheduled to be paid on each automobile contract, net of unearned income on precomputed automobile contracts.

⁽²⁾ We consider an automobile contract delinquent when an obligor fails to make at least 90% of a contractually due payment by the following due date, which date may have been extended within limits specified in the Servicing Agreements. The period of delinquency is based on the number of days payments are contractually past due. Automobile contracts less than 31 days delinquent are not included.

⁽³⁾ Amount in repossession represents financed vehicles that have been repossessed but not yet liquidated.

⁽²⁾ Net charge-offs include the remaining principal balance, after the application of the net proceeds from the liquidation of the vehicle (excluding accrued and unpaid interest) and amounts collected subsequent to the date of charge-off, including some recoveries which have been classified as other income in the accompanying interim financial statements. September 30, 2009 and September 30, 2008 percentage represents nine months ended September 30, 2009 and September 30, 2008 annualized. December 31, 2008 represents 12 months ended December 31, 2008.

Liquidity and Capital Resources

Our business requires substantial cash to support our purchases of automobile contracts and other operating activities. Our primary sources of cash have been cash flows from operating activities, including proceeds from sales of automobile contracts, amounts borrowed under various revolving credit facilities (also sometimes known as warehouse credit facilities), servicing fees on portfolios of automobile contracts previously sold in securitization transactions or serviced for third parties, customer payments of principal and interest on finance receivables, and releases of cash from securitized pools of automobile contracts in which we have retained a residual ownership interest and from the spread account associated with such pools. Our primary uses of cash have been the purchases of automobile contracts, repayment of amounts borrowed under lines of credit and otherwise, operating expenses such as employee, interest, occupancy expenses and other general and administrative expenses, the establishment of spread account and initial overcollateralization, and the increase of credit enhancement to required levels in securitization transactions, and income taxes. There can be no assurance that internally generated cash will be sufficient to meet our cash demands. The sufficiency of internally generated cash will depend on the performance of securitized pools (which determines the level of releases from those pools and their related spread accounts), the rate of expansion or contraction in our managed portfolio, and the terms upon which we are able to acquire, sell, and borrow against automobile contracts.

Net cash provided by operating activities for the nine-month period ended September 30, 2009 was \$78.0 million compared to net cash provided by operating activities for the nine-month period ended September 30, 2008 of \$108.6 million. Cash provided by operating activities is affected by our net earnings, or loss, before provisions for credit losses.

Net cash provided by investing activities for the nine-month period ended September 30, 2009 was \$340.0 million compared to net cash provided by investing activities of \$359.9 million in the prior year period. Cash flows from investing activities has primarily related to purchases of automobile contracts less principal amortization on our consolidated portfolio of automobile contracts. The adverse changes that have taken place in the securitization market since the fourth quarter of 2007 have caused us to curtail our purchases of automobile contracts in order to preserve liquidity.

Net cash used by financing activities for the nine months ended September 30, 2009 was \$419.7 million compared to net cash used by financing activities of \$466.2 million in the prior year period. Cash used by financing activities is generally related to the repayment of securitization trust debt net of any new issuances of securitization trust debt. We repaid \$400.8 million in securitization trust debt in the nine months ended September 30, 2009 compared to \$543.1 million in the prior year period. However, our securitization trust debt repayments during the nine months ended September 30, 2008 were offset by proceeds of \$285.4 million from the issuance of new securitization trust debt. We have not issued any new securitization trust debt thus far in 2009.

We purchase automobile contracts from dealers for a cash price computed by reference to their principal amounts, reduced by an acquisition fee. Those automobile contracts generate cash flow, however, over a period of years. As a result, we have been dependent on warehouse credit facilities to purchase automobile contracts, and on the availability of cash from outside sources in order to finance our continuing operations, as well as to fund the portion of automobile contract purchase prices not financed under revolving warehouse credit facilities. At January 1, 2008, we had \$425 million in warehouse credit capacity, in the form of two \$200 million senior facilities, and one \$25 million subordinated facility. One \$200 million facility provided funding for automobile contracts purchased under the TFC programs while both warehouse facilities provided funding for automobile contracts purchased under the CPS programs. The subordinated facility was established on January 12, 2007 and expired by its terms in April 2008. From December 2008 until September 25, 2009 we had no available funding facilities and reduced our volume of contract purchases to nominal levels. On September 25, 2009, we entered into a \$50 million, two-year revolving credit agreement with Fortress Credit Corp. ("Fortress"). The Fortress agreement allows advances to us of up to 75% of the principal balance of eligible collateral that are secured by automobile receivables that we now hold or may purchase in the future from dealers.

Loans under the Fortress agreement bear interest at a floating rate equal to one-month LIBOR plus 12.00%, but in all events no less than 14.00% per year. As of September 30, 2009, \$5.2 million was outstanding under this facility. As part of the consideration given to Fortress for committing to make loans under this facility, we issued a 10-year warrant to purchase up to 1,158,087 of our common shares, at an exercise price of \$0.879 per share (we refer to this as the Fortress Warrant). Issuance of the Fortress Warrant required an adjustment to the terms of an existing outstanding warrant regarding 1,564,324 shares, reducing the exercise price of that other warrant from \$1.44 per share to \$1.407 per share and increasing the number of shares available for purchase to 1,600,991.

The first of two warehouse facilities in place at January 1, 2008 as mentioned above was provided by an affiliate of Bear, Stearns and was structured to allow us to fund a portion of the purchase price of automobile contracts by drawing against a floating rate variable funding note issued by our consolidated subsidiary Page Three Funding, LLC. This facility was established on November 15, 2005, and expired on November 6, 2008. On November 8, 2006 the facility was increased from \$150 million to \$200 million and the maximum advance rate was increased to 83% from 80% of eligible contracts, subject to collateral tests and certain other conditions and covenants. On January 12, 2007 the facility was amended to allow for the issuance of subordinated notes resulting in an increase of the maximum advance rate to 93%. The advance rate was subject to the lender's valuation of the collateral which, in turn, was affected by factors such as the credit performance of our managed portfolio and the terms and conditions of our term securitizations, including the expected yields required for bonds issued in our term securitizations. Senior notes under this facility accrued interest at a rate of one-month LIBOR plus 2.50% per annum while the subordinated notes accrued interest at a rate of one-month LIBOR plus 5.50% per annum.

The second of two warehouse facilities in place at January 1, 2008 was provided by an affiliate of UBS AG and was similarly structured to allow us to fund a portion of the purchase price of automobile contracts by drawing against a floating rate variable funding note issued by our consolidated subsidiary Page Funding LLC. This facility was entered into on June 30, 2004. On June 30, 2005 the facility was increased from \$100 million to \$125 million and further amended to provide for funding for automobile contracts purchased under the TFC programs, in addition to our CPS programs. The available credit under the facility was increased again to \$200 million on August 31, 2005. In April 2006, the terms of this facility were amended to allow advances to us of up to 80% of the principal balance of automobile contracts that we purchase under our CPS programs, and of up to 70% of the principal balance of automobile contracts that we purchase under our TFC programs, in all events subject to collateral tests and certain other conditions and covenants. On June 30, 2006, the terms of this facility were amended to allow advances to us of up to 83% of the principal balance of automobile contracts that we purchase under our CPS programs, in all events subject to collateral tests and certain other conditions and covenants. In February 2007 the facility was amended to allow for the issuance of subordinated notes resulting in an increase of the maximum advance rate to 93%. The advance rate was subject to the lender's valuation of the collateral which, in turn, was affected by factors such as the credit performance of our managed portfolio and the terms and conditions of our term securitizations, including the expected yields for bonds issued in our term securitizations. Senior notes under this facility accrue interest at a rate of one-month LIBOR plus 9.85% per annum. The facility was amended in December 2008 to eliminate future advances and provide for repayment of the notes from proceeds collected on the underlying pledged rece

We securitized \$509.0 million in auto contracts in two private placement transactions in 2008. Our April 2008 transaction was structured as a secured financing and, therefore, resulted in no gain or loss on sale. The September 2008 transaction was structured as a sale for financial accounting purposes and resulted in a loss on sale of \$14.0 million. We did not securitize any contracts during the nine-month period ended September 30, 2009.

In July 2007, we established a combination term and revolving residual credit facility, and have used eligible residual interests in securitizations as collateral for floating rate borrowings. The amount that we were able to borrow was computed using an agreed valuation methodology of the residuals, subject to an overall maximum

principal amount of \$120 million, represented by (i) a \$60 million Class A-1 variable funding note (the "revolving note"), and (ii) a \$60 million Class A-2 term note (the "term note"). The term note was fully drawn in July 2007 and was due in July 2009. As of July 2008, we had drawn \$26.8 million on the revolving note. The facility's revolving feature expired in July 2008. On July 10, 2008 we amended the terms of the combination term and revolving residual credit facility, (i) eliminating the revolving feature and increasing the interest rate, (ii) consolidating the amounts then owing on the Class A-1 note with the Class A-2 note, (iii) establishing an amortization schedule for principal reductions on the Class A-2 note, and (iv) providing for an extension, at our option if certain conditions were met, of the Class A-2 note maturity from June 2009 to June 2010. In June 2009 we met all the conditions and extended the maturity of the term note. In conjunction with the July 2008 amendment, we reduced the principal amount outstanding to \$70 million by delivering to the lender (i) warrants valued as being equivalent to 2,500,000 common shares, or \$4,071,429 and (ii) cash of \$12,765,244. The warrants represent the right to purchase 2,500,000 CPS common shares at a nominal exercise price, at any time prior to July 10, 2018. As of September 30, 2009 the aggregate indebtedness under this facility was \$59.9 million.

On June 30, 2008, we entered into a series of agreements pursuant to which a lender purchased a \$10 million five-year, fixed rate, senior secured note from us. The indebtedness is secured by substantially all of our assets, though not by the assets of our special-purpose financing subsidiaries. In July 2008, in conjunction with the amendment of the combination term and revolving residual credit facility as discussed above, the lender purchased an additional \$15 million note with substantially the same terms as the \$10 million note. Pursuant to the June 30, 2008 securities purchase agreement, we issued to the lender 1,225,000 shares of common stock. In addition, we issued the lender two warrants: (i) warrants that we refer to as the FMV Warrants, which are exercisable for 1,600,991 shares of our common stock, at an exercise price of \$1.407 per share; and (ii) warrants that we refer to as the N Warrants, which are exercisable for 283,985 shares of our common stock, at a nominal exercise price. Both the FMV Warrants and the N Warrants are exercisable in whole or in part and at any time up to and including June 30, 2018. We valued the warrants using the Black-Scholes valuation model and recorded their value as a liability on our balance sheet because the terms of the warrants also included a provision whereby the lender could require us to purchase the warrants for cash. That provision was eliminated by mutual agreement in September 2008. The FMV Warrants were initially exercisable to purchase 1,500,000 shares for \$2.573 per share, were adjusted in connection with the July 2008 issuance of other warrants to become exercisable to purchase 1,564,324 shares at \$2.4672 per share, and were further adjusted in connection with a July 2009 amendment of our option plan to become exercisable at \$1.44 per share. Upon issuance in September 2009 of the Fortress Warrant, the the FMV Warrant was further adjusted to become exercisable to purchase 1,600,991 shares at an exercise price of \$1.407 per share.

The acquisition of automobile contracts for subsequent sale in securitization transactions, and the need to fund spread accounts and initial overcollateralization, if any, and increase credit enhancement levels when those transactions take place, results in a continuing need for capital. The amount of capital required is most heavily dependent on the rate of our automobile contract purchases, the required level of initial credit enhancement in securitizations, and the extent to which the previously established trusts and their related spread accounts either release cash to us or capture cash from collections on securitized automobile contracts. Of those, the factor most subject to our control is the rate at which we purchase automobile contracts.

We are and may in the future be limited in our ability to purchase automobile contracts due to limits on our capital. As of September 30, 2009, we had unrestricted cash of \$20.4 million. We have \$44.8 million available under our Fortress facility (subject to available eligible collateral) and no immediate plans to complete a securitization. Our plans to manage our liquidity include maintaining our rate of automobile contract purchases at a merely nominal level, and, wherever appropriate, reducing our operating costs. There can be no assurance that we will be able to obtain additional future funding facilities or to complete securitizations on favorable economic terms or that we will be able to complete securitizations at all. If we are unable to complete such securitizations, we may be unable to increase our rate of automobile contract purchases, in which case our interest income and other portfolio related income would continue to decrease.

Our liquidity will also be affected by releases of cash from the trusts established with our securitizations. While the specific terms and mechanics of each spread account vary among transactions, our securitization

agreements generally provide that we will receive excess cash flows, if any, only if the amount of credit enhancement has reached specified levels and/or the delinquency, defaults or net losses related to the automobile contracts in the pool are below certain predetermined levels. In the event delinquencies, defaults or net losses on the automobile contracts exceed such levels, the terms of the securitization: (i) may require increased credit enhancement to be accumulated for the particular pool; (ii) may restrict the distribution to us of excess cash flows associated with other pools; or (iii) in certain circumstances, may permit the insurers to require the transfer of servicing on some or all of the automobile contracts to another servicer. There can be no assurance that collections from the related trusts will continue to generate sufficient cash. Moreover, most of our spread account balances are pledged as collateral to our residual interest financing. As such, most of the current releases of cash from our securitization trusts are directed to pay the obligations of our residual interest financing.

Certain of our securitization transactions and our warehouse credit facility contain various financial covenants requiring certain minimum financial ratios and results. Such covenants include maintaining minimum levels of liquidity and net worth and not exceeding maximum leverage levels and maximum financial losses. In addition, certain securitization and non-securitization related debt contain cross-default provisions that would allow certain creditors to declare a default if a default occurred under a different facility.

The agreements under which we receive periodic fees for servicing automobile contracts in securitizations are terminable by the respective note insurers upon defined events of default, and, in some cases, at the will of the insurer. We have received waivers regarding the potential breach of certain such covenants relating to minimum net worth and maintenance of active warehouse credit facilities. Without such waivers, certain credit enhancement providers would have had the right to terminate us as servicer with respect to certain of our outstanding securitization pools. Although such rights have been waived, such waivers are temporary, and there can be no assurance as to their future extension. We do, however, believe that we will obtain such future extensions because it is generally not in the interest of any party to the securitization transaction to transfer servicing. Nevertheless, there can be no assurance as to our belief being correct. Were a note insurer in the future to exercise its option to terminate such agreements, such a termination could have a material adverse effect on our liquidity and results of operations, depending on the number and value of the terminated agreements. Our note insurers continue to extend our term as servicer on a monthly and/or quarterly basis, pursuant to the servicing agreements.

Critical Accounting Policies

(a) Allowance for Finance Credit Losses

In order to estimate an appropriate allowance for losses incurred on finance receivables held on our Unaudited Condensed Consolidated Balance Sheet, we use a loss allowance methodology commonly referred to as "static pooling," which stratifies our finance receivable portfolio into separately identified pools. Using analytical and formula-driven techniques, we estimate an allowance for finance credit losses, which management believes is adequate for probable credit losses that can be reasonably estimated in our portfolio of finance receivable automobile contracts. Provision for losses is charged to our Unaudited Consolidated Statement of Operations. Net losses incurred on finance receivables are charged to the allowance. Management evaluates the adequacy of the allowance by examining current delinquencies, the characteristics of the portfolio and the value of the underlying collateral. As conditions change, our level of provisioning and/or allowance may change as well.

(b) Contract acquisition fees and originations costs

Upon purchase of a contract from a dealer, we generally charge or advance the dealer an acquisition fee. For contracts securitized in pools which were structured as sales for financial accounting purposes, the acquisition fees associated with contract purchases were deferred until the contracts were securitized, at which time the deferred acquisition fees were recognized as a component of the gain on sale.

For contracts purchased and securitized in pools which are structured as secured financings for financial accounting purposes, dealer acquisition fees and deferred originations costs are applied against the carrying value of finance receivables and are accreted into earnings as an adjustment to the yield over the estimated life of the contract using the interest method, in accordance with Statement of Financial Accounting Standard No. 91, Accounting for Nonrefundable Fees and Costs Associated with Originating or Acquiring Loans and Initial Direct Costs of Leases.

(c) Income taxes

We and our subsidiaries file consolidated federal income and combined state franchise tax returns. We utilize the asset and liability method of accounting for income taxes, under which deferred income taxes are recognized for the future tax consequences attributable to the differences between the financial statement values of existing assets and liabilities and their respective tax bases. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred taxes of a change in tax rates is recognized in income in the period that includes the enactment date. A valuation allowance is recorded against that portion of the deferred tax asset whose utilization in future period is not more than likely.

In determining the possible realization of deferred tax assets, future taxable income from the following sources are considered: (a) the reversal of taxable temporary differences; (b) future operations exclusive of reversing temporary differences; and (c) tax planning strategies that, if necessary, would be implemented to accelerate taxable income into a period in which net operating losses might otherwise expire.

(d) Stock-based compensation

For the nine months ended September 30, 2009 and 2008, we recorded \$1.0 million and \$953,000 respectively, in stock-based compensation costs, resulting from grants of options during the period and vesting of previously granted options. As of September 30, 2009, there were \$3.7 million in unrecognized stock-based compensation costs to be recognized over future periods.

Forward Looking Statements

This report on Form 10-Q includes certain "forward-looking statements." Forward-looking statements may be identified by the use of words such as "anticipates," "expects," "plans," "estimates," or words of like meaning. Our provision for credit losses is a forward-looking statement, as it is dependent on our estimates as to future chargeoffs and recovery rates. Factors that could affect charge-offs and recovery rates include changes in the general economic climate, which could affect the willingness or ability of obligors to pay pursuant to the terms of automobile contracts, changes in laws respecting consumer finance, which could affect our ability to enforce rights under automobile contracts, and changes in the market for used vehicles, which could affect the levels of recoveries upon sale of repossessed vehicles. Factors that could affect our revenues in the current year include the levels of cash releases from existing pools of automobile contracts, which would affect our ability to purchase automobile contracts, the terms on which we are able to finance such purchases, the willingness of dealers to sell automobile contracts to us on the terms that we offer, and the terms on which we are able to complete term securitizations once automobile contracts are acquired. Factors that could affect our expenses in the current year include competitive conditions in the market for qualified personnel and interest rates (which affect the rates that we pay on notes issued in our securitizations).

Item 3. Quantitative and Qualitative Disclosures about Market Risk

Interest Rate Risk

During periods where we purchase significant automobile contracts and pledge them for borrowings under warehouse credit facilities, we are subject to interest rate risk during the period between when the contracts are purchased from dealers and when they become part of a term securitization. Specifically, the interest rates on the warehouse facilities are adjustable while the interest rates on the automobile contracts are fixed. Historically, our term securitization facilities have had fixed rates of interest. To mitigate some of this risk, we have in the past, and intend to continue to, structure certain of our securitization transactions to include pre-funding structures, in which the amount of notes issued exceeds the amount of automobile contracts initially sold to the trusts. In pre-funding, the proceeds from the pre-funded portion are held in an escrow account until we sell the additional automobile contracts to the trust in amounts up to the balance of the pre-funded escrow account. In pre-funded securitizations, we lock in the borrowing costs with respect to the automobile contracts it subsequently delivers to the trust. However, we incur an expense in pre-funded securitizations equal to the difference between the money market yields earned on the proceeds held in escrow prior to subsequent delivery of automobile contracts and the interest rate paid on the notes outstanding, as to the amount of which there can be no assurance.

During 2009, we have not purchased a significant volume of automobile contracts and neither have we pledged any new contracts for borrowings under any warehouse facilities.

Item 4T. Controls and Procedures

We maintain a system of internal controls and procedures designed to provide reasonable assurance as to the reliability of our published financial statements and other disclosures included in this report. As of the end of the period covered by this report, we evaluated the effectiveness of the design and operation of such disclosure controls and procedures. Based upon that evaluation, the principal executive officer (Charles E. Bradley, Jr.) and the principal financial officer (Jeffrey P. Fritz) concluded that the disclosure controls and procedures are effective in recording, processing, summarizing and reporting, on a timely basis, material information relating to us that is required to be included in our reports filed under the Securities Exchange Act of 1934. There have been no changes in our internal controls over financial reporting during our most recently completed fiscal quarter that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

PART II — OTHER INFORMATION

Item 1. Legal Proceedings

The information provided under the caption "Legal Proceedings," Note 7 to the Unaudited Condensed Consolidated Financial Statements, included in Part I of this report, is incorporated herein by reference.

Item 1A. Risk Factors

We remind the reader that risk factors are set forth in Item 1A of our report on Form 10-K, filed with the U.S. Securities and Exchange Commission on March 31, 2009. Where we are aware of material changes to such risk factors as previously disclosed, we set forth below an updated discussion of such risks. The reader should note that the other risks identified in our report on Form 10-K remain applicable to us.

We require a substantial amount of cash to service our substantial debt.

To service our existing substantial indebtedness, we require a significant amount of cash. Our ability to generate cash depends on many factors, including our successful financial and operating performance. Our financial and operational performance depends upon a number of factors, many of which are beyond our control. These factors include, without limitation:

- · the economic and competitive conditions in the asset-backed securities market;
- · the performance of our current and future automobile contracts;
- \cdot the performance of our residual interests from our securitizations and warehouse credit facilities;
- · any operating difficulties or pricing pressures we may experience;
- · our ability to obtain credit enhancement for our securitizations;
- · our ability to establish and maintain dealer relationships;
- · the passage of laws or regulations that affect us adversely;
- $\cdot\,$ our ability to compete with our competitors; and
- · our ability to acquire and finance automobile contracts.

Depending upon the outcome of one or more of these factors, we may not be able to generate sufficient cash flow from operations or obtain sufficient funding to satisfy all of our obligations. We note in particular that the market for asset-backed securities is, as of the date of this report, severely adverse to issuers such as ourselves. We also note that credit enhancement in the form of financial guaranty insurance policies does not appear to be available. There can be no assurance as to when, whether or on what terms we may be able to securitize pools of automobile contracts in the future. If the unavailability of securitization transactions were to cause us to be unable to pay our debts, we would be required to pursue one or more alternative strategies, such as selling assets, refinancing or restructuring our indebtedness or selling additional equity capital. These alternative strategies might not be feasible at the time, might prove inadequate or could require the prior consent of our lenders.

We need substantial liquidity to operate our business.

We have historically funded our operations principally through internally generated cash flows, sales of debt and equity securities, including through securitizations and warehouse credit facilities, borrowings

under senior secured debt agreements and sales of subordinated notes. However, we may not be able to obtain sufficient funding for our future operations from such sources. As of the date of this report, our access to the capital markets is impaired, with respect to both short-term and long-term debt. As a result, our results of operations, financial condition and cash flows have been and may continue to be materially and adversely affected. We require a substantial amount of cash liquidity to operate our business. Among other things, we have used such cash liquidity to:

- · acquire automobile contracts;
- · fund overcollateralization in warehouse credit facilities and securitizations;
- · pay securitization fees and expenses;
- · fund spread accounts in connection with securitizations;
- · satisfy working capital requirements and pay operating expenses;
- · pay taxes; and
- · pay interest expense.

To mitigate the effects of our difficulties in obtaining financing on acceptable terms, we have materially reduced our acquisitions of automobile contracts, and have refrained from attempting to conduct securitization transactions on terms that we believe would be too burdensome to be prudent. We continue to pay our operating expenses, taxes and interest expense, and to satisfy our working capital requirements. However, there can be no assurance of our continued ability to do so.

We have substantial indebtedness.

We have and will continue to have a substantial amount of indebtedness. At September 30, 2009 and December 31, 2008, we had approximately \$1,119.9 million and \$1,527.3 million, respectively, of debt outstanding. Such debt consisted, as of December 31, 2008, primarily of \$1,404.2 million of securitization trust debt, and also included \$9.9 million of warehouse indebtedness, \$67.3 million of residual interest financing, \$20.1 million of securitization trust debt, and also included a subordinated notes program. At September 30, 2009, such debt consisted primarily of \$1,012.8 million of securitization trust debt, and also included \$5.2 million of warehouse indebtedness, \$59.9 million of residual interest financing, \$20.8 million of senior secured debt, and \$21.2 million owed under a subordinated notes program. We are currently offering the subordinated notes to the public on a continuous basis, and such notes have maturities that range from three months to 10 years.

Our substantial indebtedness could adversely affect our financial condition by, among other things:

- · increasing our vulnerability to general adverse economic and industry conditions;
- · requiring us to dedicate a substantial portion of our cash flow from operations to payments on our indebtedness, thereby reducing amounts available for working capital, capital expenditures and other general corporate purposes;
- · limiting our flexibility in planning for, or reacting to, changes in our business and the industry in which we operate;
- · placing us at a competitive disadvantage compared tour competitors that have less debt; and
- · limiting our ability to borrow additional funds.

Although we believe we are able to service and repay such debt, there is no assurance that we will be able to do so. If we do not generate sufficient operating profits, our ability to make required payments on our debt would be impaired. Failure to pay our indebtedness when due could have a material adverse effect.

If an increase in interest rates results in a decrease in our cash flow from excess spread, our results of operations may be impaired.

Our profitability is largely determined by the difference, or "spread," between (i) the interest rates payable under our warehouse credit facilities and on the asset-backed securities issued in our securitizations, or payable in any alternate permanent financing transactions, and (ii) the effective interest rate received by us on the automobile contracts that we acquire. Disruptions in the market for asset-backed securities observed over the past year have resulted in an increase in the interest rates we paid on the asset-backed securities that we issued in our most recent securitization, as compared with prior transactions, and may result in any future transactions involving comparably high (or higher) interest rates payable by us. Although we have attempted to offset increases in our cost of funds by increasing fees we charge to dealers when purchasing contracts and by requiring higher interest rates on contracts we purchase, there can be no assurance that such price increases on our part will fully offset increases in interest we pay to finance our managed portfolio.

In addition to the interest rates payable in our financing transactions, there are other factors that affect our ability to manage interest rate risk. Specifically, we are subject to interest rate risk during the period between when automobile contracts are purchased from dealers and when such contracts are sold and financed in a securitization or any alternate permanent financing transaction. Interest rates on our warehouse credit facilities are adjustable while the interest rates on the automobile contracts are fixed. Therefore, if interest rates increase, the interest we must pay to the lenders under our warehouse credit facilities is likely to increase while the interest realized by us from those warehoused automobile contracts remains the same, and thus, during the warehousing period, the excess spread cash flow received by us would likely decrease. Additionally, contracts warehoused and then securitized during a rising interest rate environment may result in less excess spread cash flow realized by us under those securitizations as, historically, our securitization facilities pay interest to security holders on a fixed rate basis set at prevailing interest rates at the time of the closing of the securitization, which may be several months after the securitized contracts were originated and entered the warehouse, while our customers pay fixed rates of interest on the contracts, set at the time they purchase the underlying vehicles. A decrease in excess spread cash flow could adversely affect our earnings and cash flow.

To mitigate, but not eliminate, the short-term risk relating to interest rates payable by us under the warehouse facilities, we have generally held automobile contracts in the warehouse credit facilities for less than four months. The disruptions in the market for asset-backed securities issued in securitizations have caused us to lengthen that period, which has reduced the effectiveness of this mitigation strategy. To mitigate, but not eliminate, the long-term risk relating to interest rates payable by us in securitizations, we have in the past, and intend to continue to, structure some of our securitization transactions to include prefunding structures, whereby the amount of securities issued exceeds the amount of contracts initially sold into the securitization. In pre-funding, the proceeds from the pre-funded portion are held in an escrow account until we sell the additional contracts into the securitization in amounts up to the balance of the pre-funded escrow account. In pre-funded securitizations, we effectively lock in our borrowing costs with respect to the contracts we subsequently sell into the securitization. However, we incur an expense in pre-funded securitizations equal to the difference between the money market yields earned on the proceeds held in escrow prior to subsequent delivery of contracts and the interest rate paid on the securities issued in the securitization. The amount of such expense may vary. Despite these mitigation strategies, an increase in prevailing interest rates would cause us to receive less excess spread cash flows on automobile contracts, and thus could adversely affect our earnings and cash flows.

Forward-Looking Statements

Discussions of certain matters contained in this report may constitute forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended (the "Securities Act") and Section 21E of the Exchange Act, and as such, may involve risks and uncertainties. These forward-looking statements relate to, among other things, expectations of the business environment in which we operate, projections of future performance, perceived opportunities in the market and statements regarding our mission and vision. You can generally identify forward-looking statements as statements containing the words "will," "would," "believe," "may," "could," "expect," "anticipate," "intend," "estimate," "assume" or other similar expressions. Our actual results, performance and achievements may differ materially from the results, performance and achievements expressed or implied in such forward-looking statements. The discussion under "Risk Factors" identifies some of the factors that might cause such a difference, including the following:

- · changes in general economic conditions;
- · our ability or inability to obtain necessary financing
- · changes in interest rates;
- · our ability to generate sufficient operating and financing cash flows;
- · competition;
- · level of future provisioning for receivables losses; and
- regulatory requirements.

Forward-looking statements are not guarantees of performance. They involve risks, uncertainties and assumptions. Actual results may differ from expectations due to many factors beyond our ability to control or predict, including those described herein, and in documents incorporated by reference in this report. For these statements, we claim the protection of the safe harbor for forward-looking statements contained in the Private Securities Litigation Reform Act of 1995.

We undertake no obligation to publicly update any forward-looking information. You are advised to consult any additional disclosure we make in our periodic reports filed with the SEC. See "Where You Can Find More Information" and "Documents Incorporated by Reference."

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

During the three months ended September 30, 2009, we purchased a total of 234,807 shares of our common stock, as described in the following table:

Issuer Purchases of Equity Securities

	Total			Total Number of Shares Purchased as		oproximate Dollar llue of Shares that	
	Number of Shares			Part of Publicly Announced Plans or		May Yet be Purchased Under the Plans or	
Period(1)	Purchased	per Share		per Share Programs		Programs	
July 2009	-	\$	-	-	\$	1,980,373	
August 2009	108,812		0.91	108,812	\$	1,881,210	
September 2009	125,995		0.92	125,995	\$	1,765,147	
Total	234,807	\$	0.92	234,807			

⁽¹⁾ Each monthly period is the calendar month.

Item 4. Submission of Matters to a Vote of Security Holders

Our annual meeting of shareholders was held on July 17, 2009. At the meeting, each of the six nominees to the Board of Directors was elected for a one-year term by the shareholders, with votes cast as follows:

Nominee	Votes For	Votes Withheld	
Charles E. Bradley, Jr.	16,181,115	378,279	
Chris A. Adams	16,203,311	356,080	
Brian J. Rayhill	16,191,163	368,228	
William B. Roberts	15,929,663	629,728	
Gregory S. Washer	15,931,235	628,156	
Daniel S. Wood	15,930,335	629,056	

The shareholders also approved the two other proposals placed before the annual meeting. Those proposals were (i) to ratify the appointment of Crowe Horwath, LLP as independent auditors of the Company for the fiscal year ending December 31, 2009, and (ii) to approve an amendment to our 2006 Long-Term Equity Incentive Plan, which amendment permits an exchange and repricing of outstanding options. Votes on the proposals were cast as follows:

	Ratification of Selection of Independent	Amendment of our 2006 Long-Term Equity		
	Auditors	Incentive Plan		
For	16,381,267	6,630,936		
Against	156,236	3,179,603		
Abstain	21,888	54,153		
Broker Non-votes	0	6,694,699		

Item 5. Other Information

The discussion of partially waived events of default under insurance agreements related to our securitization pools appearing under the caption "Financial Covenants" in Note 1 to the interim unaudited financial statements included in Part I of this quarterly report is incorporated herein by reference.

Item 6. Exhibits

The Exhibits listed below are filed with this report.

- 4.14 Instruments defining the rights of holders of long-term debt of certain consolidated subsidiaries of the registrant are omitted pursuant to the exclusion set forth in subdivisions (b)(iv)(iii)(A) and (b)(v) of Item 601 of Regulation S-K (17 CFR 229.601). The registrant agrees to provide copies of such instruments to the United States Securities and Exchange Commission upon request.
- 10.26 Revolving Loan Note dated September 25, 2009
- 10.27 Warrant dated September 25, 2009 to purchase 1,158,087 common shares.

- 10.28 Revolving Credit Agreement dated September 25, 2009
- 31.1 Rule 13a-14(a) Certification of the Chief Executive Officer of the registrant.
- 31.2 Rule 13a-14(a) Certification of the Chief Financial Officer of the registrant.
- 32 Section 1350 Certifications.*

^{*} These Certifications shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, or otherwise subject to the liability of that section. These Certifications shall not be deemed to be incorporated by reference into any filing under the Securities Act of 1933, as amended, or the Exchange Act, except to the extent that the registration statement specifically states that such Certifications are incorporated therein.

SIGNATURES

Pursuar	nt to the requirements of th	ne Securities Exchange A	ct of 1934, the	registrant duly of	caused this report to	o be signed on its b	ehalf by the ui	ndersigned
thereunto du	ıly authorized.							

CONSUMER PORTFOLIO SERVICES, INC.

(Registrant)

Date: November 16, 2009

By: /s/ CHARLES E. BRADLEY, JR.

Charles E. Bradley, Jr.

President and Chief Executive Officer

(Principal Executive Officer)

Date: November 16, 2009

By: /s/ JEFFREY P. FRITZ

Jeffrey P. Fritz

Senior Vice President and Chief Financial Officer

(Principal Financial Officer)

REVOLVING LOAN NOTE

\$50,000,000 September 25, 2009

FOR VALUE RECEIVED, PAGE FOUR FUNDING LLC, a Delaware limited liability company ("Company"), promises to pay FORTRESS CREDIT CORP., a Delaware corporation ("Payee"), or its registered assigns, on or before September 25, 2011, the lesser of (a) FIFTY MILLION DOLLARS (\$50,000,000) and (b) the outstanding principal amount of all advances made by Payee to Company as Revolving Loans under the Credit Agreement referred to below.

Company also promises to pay interest on the unpaid principal amount hereof, from the date hereof until paid in full, at the rates and at the times which shall be determined in accordance with the provisions of that certain Revolving Credit Agreement, dated as of September 25, 2009 (as amended, restated, replaced, supplemented or otherwise modified from time to time, the "Credit Agreement"; the terms defined therein and not otherwise defined herein being used herein as therein defined), by and among Consumer Portfolio Services, Inc., Company, as Borrower and Payee, as Administrative Agent (in such capacity, "Administrative Agent"), Collateral Agent and Lender.

This Revolving Loan Note is one of the "Revolving Loan Notes" in the aggregate principal amount of \$50,000,000 and is issued pursuant to and entitled to the benefits of the Credit Agreement, to which reference is hereby made for a more complete statement of the terms and conditions under which the Loans evidenced hereby were made and are to be repaid.

All payments of principal and interest in respect of this Revolving Loan Note shall be made in lawful money of the United States of America in same day funds at the Principal Office of Administrative Agent or at such other place as shall be designated in writing for such purpose in accordance with the terms of the Credit Agreement. Unless and until an Assignment Agreement effecting the assignment or transfer of the obligations evidenced hereby shall have been defined in accordance with the Credit Agreement and accepted by Administrative Agent and recorded in the Register, Company shall be entitled to deem and treat Payee as the owner and holder of this Revolving Loan Note and the obligations evidenced hereby. Payee hereby agrees, by its acceptance hereof, that before disposing of this Revolving Loan Note or any part hereof it will make a notation hereon of all principal payments previously made hereunder and of the date to which interest hereon has been paid; provided, the failure to make a notation of any payment made on this Revolving Loan Note shall not limit or otherwise affect the obligations of Company hereunder with respect to payments of principal of or interest on this Revolving Loan Note.

This Revolving Loan Note is subject to mandatory prepayment and to prepayment at the option of Company, each as provided in the Credit Agreement.

THIS REVOLVING LOAN NOTE AND THE RIGHTS AND OBLIGATIONS OF COMPANY AND PAYEE HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE INTERNAL LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO CONFLICTS OF LAWS PRINCIPLES THEREOF.

Upon the occurrence of an Event of Default (as defined in the Credit Agreement), the unpaid balance of the principal amount of this Revolving Loan Note, together with all accrued and unpaid interest thereon, may become, or may be declared to be, due and payable in the manner, upon the conditions and with the effect provided in the Credit Agreement.

The terms of this Revolving Loan Note are subject to amendment only in the manner provided in the Credit Agreement.

No reference herein to the Credit Agreement and no provision of this Revolving Loan Note or the Credit Agreement shall alter or impair the obligations of Company, which are absolute and unconditional, to pay the principal of and interest on this Revolving Loan Note at the place, at the respective times, and in the currency herein prescribed.

Company promises to pay all reasonable, out-of-pocket costs and expenses, including reasonable attorneys' fees, all as provided in the Credit Agreement, incurred in the collection and enforcement of this Revolving Loan Note. Company and any endorsers of this Revolving Loan Note hereby consent to renewals and extensions of time at or after the maturity hereof, without notice, and hereby waive diligence, presentment, protest, demand notice of every kind and, to the full extent permitted by law, the right to plead any statute of limitations as a defense to any demand hereunder.

IN WITNESS WHEREOF, Company has caused this Revolving Loan Note to be duly executed and delivered by its officer thereunto duly authorized as of the date and at the place first written above.

PAGE FOUR FUNDING LLC

By: /s/ JEFFREY P. FRITZ

Name: Jeffrey P. Fritz Title: V.P. and CFO

TRANSACTIONS ON REVOLVING LOAN NOTE

Amount of Amount of Outstanding
Revolving Loan Principal Paid Principal Notation
Date Made This Date This Date Balance This Date Made By

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY APPLICABLE STATE SECURITIES LAWS, AND MAY NOT BE SOLD, TRANSFERRED, PLEDGED, HYPOTHECATED OR OTHERWISE ASSIGNED EXCEPT IN COMPLIANCE WITH THE REGISTRATION REQUIREMENTS OF SUCH ACT AND SUCH STATE SECURITIES LAWS OR PURSUANT TO AN EXEMPTION OR QUALIFICATION THEREFROM.

WARRANT NO. DBSO-1 September 25, 2009

WARRANT TO PURCHASE 1,158,087 SHARES OF COMMON STOCK

FOR VALUE RECEIVED, CONSUMER PORTFOLIO SERVICES, INC., a California corporation (the "COMPANY"), hereby certifies that DRAWBRDIGE SPECIAL OPPORTUNITIES FUND LP, a Delaware limited partnership, or its assigns (the "HOLDER"), is entitled to purchase, on the terms and subject to the conditions contained herein, One Million, One Hundred Fifty Eight Thousand, Eighty Seven (1,158,087) shares (the "WARRANT SHARES") of the Company's common stock, no par value per share ("COMMON STOCK"), at the exercise price of \$0.879 per Warrant Share (the "WARRANT PURCHASE PRICE") at any time and from time to time during the Exercise Period (as such term is defined below). The number of Warrant Shares and the Warrant Purchase Price shall be subject to adjustment as set forth in SECTION 3.

This Warrant (this "WARRANT") is the "Warrant" referred to in, and is being issued in connection with the consummation of the transactions contemplated by, that certain Revolving Credit Agreement dated as of September 25, 2009, by and among the Company, Page Four Funding LLC and Fortress Credit Corp., an affiliate of the initial holder of this Warrant (as amended or modified from time to time, the "CREDIT AGREEMENT"), and is subject to the following terms and conditions:

1. DEFINITIONS. For the purposes of this Warrant, the following terms shall have the respective meanings set forth below:

"APPLICABLE LAW" means all provisions of statutes, rules and regulations, interpretations and orders of any Governmental Authority applicable to a Person, and all orders and decrees of all courts and arbitrators in proceedings or actions in which the Person in question is a party including applicable federal, state and local laws and regulations thereunder.

"BUSINESS DAY" means any day excluding Saturday, Sunday and any day which is a legal holiday under the laws of the State of New York or the State of California or is a day on which banking institutions located in either such state are authorized or required by law or other governmental action to close.

"COMMON STOCK" has the meaning set forth in the preamble of this Warrant.

"COMPANY" has the meaning set forth in the preamble of this Warrant.

"CONVERTIBLE SECURITIES" means, with respect to any Person, any securities or other obligations issued or issuable by such Person or any other Person that are exercisable or exchangeable for, or convertible into, any Equity Interests of the Company.

"CURRENT MARKET PRICE" per share of Common Stock means, as of any specified date on which the Common Stock is publicly traded, the average of the daily market prices of the Common Stock over the twenty (20) consecutive trading days immediately preceding (and not including) such date. The 'daily market price' for each such trading day shall be (i) the closing sales price on such day on the principal stock exchange on which the Common Stock is then listed or admitted to trading or on Nasdaq, as applicable, (ii) if no sale takes place on such day on any such exchange or system, the average of the closing bid and asked prices, regular way, on such day for the Common Stock is not then listed or admitted to trading on any stock exchange or system, the last reported sale price, regular way, on such day for the Common Stock, or if no sale takes place on such day, the average of the closing bid and asked prices for the Common Stock on such day, as reported by Nasdaq or the National Quotation Bureau, or (iv) if the Common Stock is not then listed or admitted to trading on any securities exchange and if no such reported sale price or bid and asked prices are available, the average of the reported high bid and low asked prices on such day, as reported by a reputable quotation service, or a newspaper of general circulation in the City of Los Angeles customarily published on each Business Day. If the daily market price cannot be determined for the twenty (20) consecutive trading days immediately preceding such date in the manner specified in the foregoing sentence, then the Common Stock shall not be deemed to be publicly traded as of such date.

"DESIGNATED OFFICE" has the meaning set forth in SECTION 2.1.

"DILUTIVE ISSUANCE" has the meaning set forth in SECTION 3.9.

"DISTRIBUTION" has the meaning set forth in SECTION 3.3.

"DISTRIBUTION AMOUNT" has the meaning set forth in SECTION 3.3.

"DOJ" has the meaning set forth in SECTION 2.4.

"EQUITY INTERESTS" means, with respect to any Person, (i) if such Person is a corporation, any and all shares of capital stock, participations in profits or other equivalents (however designated) or other equity interests of such Person, including any preferred stock of such Person, (ii) if such Person is a limited liability company, any and all membership interests, or (iii) if such Person is a partnership or other entity, any and all partnership or entity interests or other units.

"EQUITY RIGHTS" means any warrants, options or other rights to subscribe for or purchase, or obligations to issue, any Equity Interests of the Company, or any Convertible Securities, or any stock appreciation rights, including any options or similar rights issued or issuable under any employee stock option plan, pension plan or other employee benefit plan of the Company.

"EXCLUDED SHARES" means Common Stock, Equity Rights or Convertible Securities issued in any of the transactions described in SECTIONS 3.1, 3.2, 3.3 or 3.5, (ii) shares of Common Stock issued after the Issuance Date upon conversion, exercise or exchange of (A) Equity Rights or Convertible Securities outstanding on the Issuance Date or (B) Equity Rights issued in connection with an exchange and re-pricing of outstanding options issued under the Company's 2006 Long Term Equity Incentive Plan as described in the Company's proxy statement filed with the Securities and Exchange Commission on June 24, 2009, and (iii) this Warrant or any securities issued upon exercise hereof. Notwithstanding the foregoing, shares of Common Stock, Equity Rights or Convertible Securities, and the shares of Common Stock issuable in respect thereof described in clauses (i) or (ii) shall not constitute "Excluded Shares" if the terms thereof are amended or revised after the Issuance Date.

"EXERCISE NOTICE" has the meaning set forth in SECTION 2.1.

"EXERCISE PERIOD" means the period commencing on the Issuance Date and ending on (and including) the Expiration Date.

"EXPIRATION DATE" means September 25, 2019.

"FAIR MARKET VALUE" per share of Common Stock as of any specified date means (i) if the Common Stock is publicly traded on such date, the Current Market Price per share, or (ii) if the Common Stock is not publicly traded (or deemed not to be publicly traded) on such date, the fair market value per share of Common Stock as determined in good faith by the Board of Directors of the Company and set forth in a written notice to the Holder, subject to the Holder's right to dispute such determination under SECTION 3.8(E).

"FTC" has the meaning set forth in SECTION 2.4.

"GOVERNMENTAL AUTHORITY" means any nation or government, any state or other political subdivision thereof, any agency, authority, instrumentality, regulatory body, court, administrative tribunal, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government, and any corporation or other entity owned or controlled, through stock or capital ownership or otherwise, by any of the foregoing.

"HOLDER" has the meaning set forth in the preamble of this Warrant.

"HSR ACT" has the meaning set forth in SECTION 2.4.

"ISSUANCE DATE" means September 25, 2009.

"NASDAQ" means the Nasdaq Stock Market or any successor reporting system thereof.

"OTHER PROPERTY" has the meaning set forth in SECTION 3.5.

"PERSON" means any individual, corporation, partnership, limited liability company, joint venture, association, joint stock company, trust (including any beneficiary

thereto), unincorporated organization or government or any agency or political subdivision thereof.

"REFERENCE PRICE" means the Warrant Purchase Price.

"REGISTRATION RIGHTS AGREEMENT" shall mean a Registration Rights Agreement dated as of the date hereof, in form and substance satisfactory to the initial Holder, among the Company, the initial Holder and Charles E. Bradley, Jr.

"SECURITIES ACT" means the Securities Act of 1933, as amended.

"WARRANT" means this Warrant, any amendment of this Warrant, and any warrants issued upon transfer, division or combination of, or in substitution for, this Warrant or any other such warrant. All such Warrants shall at all times be identical as to terms and conditions and date, except as to the number of Warrant Shares for which they may be exercised.

"WARRANT PURCHASE PRICE" has the meaning set forth in the preamble of this Warrant (as adjusted in accordance with the terms of this Warrant).

"WARRANT SHARES" has the meaning set forth in the preamble of this Warrant.

2. EXERCISE.

- 2.1. EXERCISE; DELIVERY OF CERTIFICATES. Subject to the provisions of SECTION 2.4, this Warrant may be exercised, in whole or in part, at the option of the Holder, at any time and from time to time during the Exercise Period, by (a) delivering to the Company at its principal executive office (the "DESIGNATED OFFICE") (i) a notice of exercise, in substantially the form attached hereto (the "EXERCISE NOTICE"), duly completed and signed by the Holder, and (ii) this Warrant, and (b) paying the Warrant Purchase Price pursuant to SECTION 2.2 for the number of Warrant Shares proposed to be purchased in the Exercise Notice. Subject to the provisions of SECTION 2.4, the Warrant Shares being purchased under this Warrant will be deemed to have been issued to the Holder, as the record owner of such Warrant Shares, as of the close of business on the date on which payment therefor is made by the Holder pursuant to SECTION 2.2. Stock certificates representing the Warrant Shares so purchased shall be delivered to the Holder within three (3) Business Days after this Warrant has been exercised (or, if applicable, immediately after the conditions set forth in SECTION 2.4 have been satisfied); PROVIDED, HOWEVER, that in the case of a purchase of less than all of the Warrant Shares issuable upon exercise of this Warrant, the Company shall cancel this Warrant and, within three (3) Business Days after this Warrant has been surrendered, execute and deliver to the Holder a new Warrant of like tenor for the number of unexercised Warrant Shares. Each stock certificate representing the number of Warrant Shares purchased pursuant to this Warrant shall be registered in the name of the Holder or, subject to compliance with Applicable Laws, such other name as designated by the Holder.
- 2.2. PAYMENT OF WARRANT PRICE. Payment of the Warrant Purchase Price shall be made, at the option of the Holder, by (i) check from the Holder, (ii) wire transfer, (iii) instructing the Company to withhold and cancel a number of Warrant Shares then issuable

upon exercise of this Warrant with respect to which the excess, if any, of the Fair Market Value over the Warrant Purchase Price for such canceled Warrant Shares is at least equal to the Warrant Purchase Price for the Warrant Shares being purchased, (iv) surrendering to the Company of shares of Common Stock previously acquired by the Holder with a Fair Market Value equal to the Warrant Purchase Price for the Warrant Shares then being purchased or (v) any combination of the foregoing.

- 2.3. NO FRACTIONAL SHARES. The Company shall not be required to issue fractional shares of Common Stock upon the exercise of this Warrant. If any fraction of a share of Common Stock would, except for the provisions of this paragraph, be issuable on the exercise of this Warrant (or specified portion thereof), the Company shall pay to the Holder an amount in cash calculated by it to be equal to the then Fair Market Value per share of Common Stock multiplied by such fraction computed to the nearest whole cent.
- 2.4. ANTITRUST NOTIFICATION. If the Holder determines, in its sole judgment upon the advice of counsel, that an exercise of this Warrant pursuant to the terms hereof is subject to the provisions of the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR ACT"), the Holder shall notify the Company, and the Company and the Holder shall, prior to the payment of the Warrant Purchase Price, file with the United States Federal Trade Commission (the "FTC") and the United States Department of Justice (the "DOJ") the notification and report form, if any, required in connection with the exercise of this Warrant, and any supplemental information required in connection therewith, pursuant to the HSR Act. Any information required to be provided by or with respect to the Company shall be provided by the Company within seven (7) Business Days after receiving notification from the Holder of the applicability of the HSR Act. Any such notification, report form and supplemental information will be in full compliance with the requirements of the HSR Act. The Company will furnish to the Holder such necessary information and such assistance as the Holder may reasonably request in connection with the preparation of any filing or submission which is necessary under the HSR Act.

The Company shall respond promptly after receiving any inquiries or requests for additional information from the FTC or the DOJ (and in no event more than five (5) Business Days after receipt of such inquiry or request). The Company shall keep the Holder apprised of the status of any communications with, and any inquiries or requests for additional information from, the FTC or the DOJ, and shall respond promptly to any such inquiries or requests. The Company shall bear all filing or other fees required to be paid by the Company and the Holder (or the "ultimate parent entity" of the Holder, if any) under the HSR Act or any other Applicable Law and the Company shall bear all costs and expenses (including, without limitation, attorneys' fees) incurred by the Company and the Holder (or the "ultimate parent entity" of the Holder, if any) in connection with the preparation of such filings, responses to inquiries or requests and compliance with the HSR Act and other Applicable Laws. In the event that this SECTION 2.4 is applicable to any exercise of this Warrant, the purchase of the Warrant Shares subject to the Exercise Notice, and the payment of the Warrant Purchase Price, will be subject to the expiration or earlier termination of the waiting period under the HSR Act.

3. ADJUSTMENTS TO THE NUMBER OF WARRANT SHARES AND TO THE WARRANT PURCHASE PRICE. The number of Warrant Shares for which this Warrant is

exercisable and the Warrant Purchase Price shall be subject to adjustment from time to time as set forth in this SECTION 3.

- 3.1. STOCK DIVIDENDS, SUBDIVISIONS AND COMBINATIONS. If at any time the Company:
- (a) pays a dividend or other distribution on its Common Stock in shares of Common Stock or shares of any other class or series of Capital Stock;
- (b) subdivides its outstanding shares of Common Stock (by stock split, reclassification or otherwise) into a larger number of shares of Common Stock; or
- (c) combines (by reverse stock split or otherwise) its outstanding shares of Common Stock into a smaller number of shares of Common Stock, then the number of Warrant Shares purchasable upon exercise of this Warrant immediately prior to the record date for such dividend or distribution or the effective date of such subdivision or combination shall be adjusted so that the Holder shall thereafter be entitled to receive upon exercise of this Warrant the kind and number of shares of Common Stock that the Holder would have owned or have been entitled to receive immediately after such record date or effective date had this Warrant been exercised immediately prior to such record date or effective date. Any adjustment made pursuant to this SECTION 3.1 shall become effective immediately after the effective date of such event, but be retroactive to the record date, if any, for such event.

Upon any adjustment of the number of Warrant Shares purchasable upon the exercise of this Warrant as herein provided, the Warrant Purchase Price per share shall be adjusted by multiplying the Warrant Purchase Price immediately prior to such adjustment by a fraction, the numerator of which shall be the number of Warrant Shares purchasable upon the exercise of this Warrant immediately prior to such adjustment and the denominator of which shall be the number of Warrant Shares so purchasable immediately thereafter.

- 3.2. RIGHTS; OPTIONS; WARRANTS. If, at any time after the Issuance Date, the Company issues (without payment of any consideration) to all holders of outstanding Common Stock any rights, options or warrants to subscribe for or purchase shares of Common Stock or securities convertible into or exchangeable for Common Stock, then the Company shall also distribute, at the same time as the distribution to such holders, such rights, options, warrants or securities to the Holder as if this Warrant had been exercised immediately prior to the record date for such issuance.
- 3.3. DISTRIBUTION OF ASSETS OR SECURITIES. If at any time the Company makes a distribution (other than a distribution covered by SECTION 3.1 or 3.2) to its shareholders of any asset, including cash or securities (such distribution, a "DISTRIBUTION", and the total of the cash, assets or securities so distributed, the "DISTRIBUTION AMOUNT"), then, at the Holder's option:
- (i) the Holder shall have the right to receive an amount of assets, including cash or securities equal to the number of Warrant Shares for which this Warrant is exercisable immediately prior to the Distribution, multiplied by the Distribution Amount, divided by the number of shares of Common Stock outstanding at such time. Upon the closing of the

Distribution, the Company shall distribute such portion of the Distribution Amount to the Holder; or

(ii) the Warrant Purchase Price shall be adjusted and shall be equal to the Warrant Purchase Price in effect immediately prior to the close of business on the date fixed for the determination of shareholders entitled to receive such distribution, multiplied by a fraction (which shall not be less than zero), the numerator of which shall be the Fair Market Value per share of Common Stock on the date fixed for such determination, less the amount of cash, the then-fair market value of the portion of the assets, or the fair market value of the portion of the securities, as the case may be (as determined in good faith by the Board of Directors of the Company, subject to the Holder's rights under SECTION 3.8(E)), so distributed applicable to one share of Common Stock, and the denominator of which shall be the Fair Market Value per share of Common Stock. Such adjustment to the Warrant Purchase Price shall become effective immediately prior to the opening of business on the day immediately following the date fixed for the determination of shareholders entitled to receive such distribution. Upon any adjustment to the Warrant Purchase Price as provided for in this SECTION 3.3(ii), the number of Warrant Shares issuable upon the exercise of this Warrant shall also be adjusted and shall be equal to the number of Warrant Shares issuable upon exercise of this Warrant immediately prior to such adjustment multiplied by a fraction, the numerator of which is the Warrant Purchase Price in effect immediately prior to such adjustment and the denominator of which is the Warrant Purchase Price as so adjusted.

3.4. ISSUANCE OF EQUITY SECURITIES AT LESS THAN FAIR MARKET VALUE OR WARRANT PURCHASE PRICE.

(a) If, at any time after Issuance Date, the Company shall issue or sell (or, in accordance with SECTION 3.4(b), shall be deemed to have issued or sold) shares of Common Stock, Equity Rights or Convertible Securities representing the right to subscribe for or purchase shares of Common Stock (other than any Excluded Shares) at a price per share of Common Stock that is lower than the Reference Price in effect immediately prior to such sale and issuance, then the Warrant Purchase Price shall be adjusted so that it shall equal the price determined by multiplying the Warrant Purchase Price in effect immediately prior thereto by a fraction, the numerator of which shall be an amount equal to the sum of (A) the number of shares of Common Stock outstanding immediately prior to such sale and issuance plus (B) the number of shares of Common Stock which the aggregate consideration received by the Company (and in the case of adjustments triggered in full or in part by the issuance of Equity Rights or Convertible Securities, the aggregate consideration deemed received in respect of such issuance determined as provided in SECTION 3.4(b) below) for such sale or issuance would purchase at such Reference Price per share, and the denominator of which shall be the total number of shares of Common Stock outstanding (and in the case of adjustments triggered in full or in part by the issuance of Equity Rights or Convertible Securities, the number of shares of Common Stock deemed to be outstanding as a result of such issuance determined as provided in SECTION 3.4(b) below) immediately after such sale or issuance.

Adjustments shall be made successively whenever such an issuance is made. Upon any adjustment in the Warrant Purchase Price as provided in this SECTION 3.4(a), the number of shares of Common Stock purchasable upon the exercise of this Warrant shall also be

adjusted and shall be that number determined by multiplying the number of Warrant Shares issuable upon exercise immediately prior to such adjustment by a fraction, the numerator of which is the Warrant Purchase Price in effect immediately prior to such adjustment and the denominator of which is the Warrant Purchase Price as so adjusted.

- (b) For the purpose of determining the adjusted Warrant Purchase Price under SECTION 3.4(a), the following shall be applicable:
- (i) If the Company in any manner issues or grants any Equity Rights (including Equity Rights to acquire Convertible Securities), and the price per share for which shares of Common Stock are issuable upon the exercise of such Equity Rights or upon conversion or exchange of such Convertible Securities is less than the Reference Price determined as of the date of such issuance or grant of such Equity Rights, then the total maximum number of shares of Common Stock issuable upon the exercise of such Equity Rights (or upon conversion or exchange of the total maximum amount of such Convertible Securities issuable upon the exercise of such Equity Rights) shall be deemed to be outstanding and to have been issued and sold by the Company for such lower price per share. For purposes of this paragraph, the price per share for which shares of Common Stock are issuable upon exercise of Equity Rights or upon conversion or exchange of Convertible Securities issuable upon exercise of Equity Rights shall be determined by dividing (A) the total amount, if any, received or receivable by the Company as consideration for the issuing or granting of such Equity Rights, plus the minimum aggregate amount of additional consideration payable to the Company upon the exercise of all such Equity Rights, plus in the case of such Equity Rights which relate to Convertible Securities, the minimum aggregate amount of additional consideration, if any, payable to the Company upon the issuance or sale of such Equity Rights or upon the conversion or exchange thereof, by (B) the total maximum number of shares of Common Stock issuable upon exercise of such Equity Rights.
- (ii) If the Company in any manner issues or grants any Convertible Securities having an exercise or conversion or exchange price per Share which is less than the Reference Price determined as of the date of issuance or sale, then the maximum number of shares of Common Stock issuable upon the conversion or exchange of such Convertible Securities shall be deemed to be outstanding and to have been issued and sold by the Company for such lower price per share. For purposes of this paragraph, the price per share for which shares of Common Stock are issuable upon conversion or exchange of Convertible Securities is determined by dividing (A) the total amount received by the Company as consideration for the issuance or sale of such Convertible Securities, plus the minimum aggregate amount of additional consideration, if any, payable to the Company upon the conversion or exchange thereof, by (B) the total maximum number of shares of Common Stock issuable upon the conversion or exchange of all such Convertible Securities.
- (iii) If the purchase price provided for in any Equity Rights, the additional consideration, if any, payable upon the issuance, conversion or exchange of any Convertible Securities or the rate at which any Convertible Securities are convertible into or exchangeable for shares of Common Stock decreases at any time, or if the number of shares of Common Stock issuable upon the exercise, issuance, conversion or exchange of any Equity

Rights or Convertible Securities increases at any time, then the Warrant Purchase Price in effect at the time of such decrease (or increase) shall be readjusted to the Warrant Purchase Price which would have been in effect at such time had such Equity Rights or Convertible Securities still outstanding provided for such decreased purchase price, additional consideration, changed conversion rate or increased shares, as the case may be, at the time initially granted, issued or sold and the number of Warrant Shares shall be correspondingly readjusted, by taking the number of Warrant Shares issuable upon the exercise of this Warrant immediately prior to such adjustment multiplied by a fraction, the numerator of which is the Warrant Purchase Price in effect immediately prior to such adjustment and the denominator of which is the Warrant Purchase Price as so adjusted.

- (iv) If at any time the Company sells and issues shares of Common Stock or Equity Rights or Convertible Securities containing the right to subscribe for or purchase shares of Common Stock for a consideration consisting, in whole or in part, of property other than cash or its equivalent, then in determining the "price per share of Common Stock" and the "consideration received by the Company" for purposes of the preceding paragraphs of this SECTION 3.4, the Board of Directors of the Company shall determine, in good faith, the fair market value of property, subject to the Holder's rights under SECTION 3.8(E).
- (v) There shall be no adjustment of the Warrant Purchase Price in respect of the Common Stock pursuant to this SECTION 3.4 if the amount of such adjustment is less than \$0.00001 per share of Common Stock; PROVIDED, HOWEVER, that any adjustments which by reason of this provision are not required to be made shall be carried forward and taken into account in any subsequent adjustment.
- 3.5. REORGANIZATION, RECLASSIFICATION, MERGER, CONSOLIDATION OR DISPOSITION OF ASSETS. If at any time the Company reorganizes its capital, reclassifies its capital stock, consolidates, merges or combines with or into another Person (where the Company is not the surviving corporation or where there is any change whatsoever in, or distribution with respect to, the outstanding Common Stock), or the Company sells, transfers or otherwise disposes of all or substantially all of its property, assets or business to another Person, other than in a transaction provided for in SECTION 3.1, 3.2, 3.3, 3.4 or 3.6, and, pursuant to the terms of such reorganization, reclassification, consolidation, merger, combination, sale, transfer or other disposition of assets, (i) shares of common stock of the successor or acquiring Person or of the Company (if it is the surviving corporation) or (ii) any cash, shares of stock or other securities or property of any nature whatsoever (including warrants or other subscription or purchase rights) in addition to or in lieu of common stock of the successor or acquiring Person or the Company ("OTHER PROPERTY") are to be received by or distributed to the holders of Common Stock who are holders immediately prior to such transaction, then the Holder shall have the right thereafter to receive, upon exercise of this Warrant, the number of shares of Common Stock, common stock of the successor or acquiring Person, and/or Other Property which holder of the number of shares of Common Stock for which this Warrant is exercisable immediately prior to such event would have owned or received immediately after and as a result of such event. In such event, the aggregate Warrant Purchase Price otherwise payable for the Warrant Shares issuable upon exercise of this Warrant shall be allocated among such securities and Other Property in proportion to the respective fair market

values of such securities and Other Property as determined in good faith by the Board of Directors of the Company, subject to the Holder's rights under SECTION 3.8(e).

In case of any such event, the successor or acquiring Person (if other than the Company) shall expressly assume the due and punctual observance and performance of each and every covenant and condition of this Warrant to be performed and observed by the Company and all the obligations and liabilities hereunder, subject to such modifications as the Holder may approve in writing (as determined by resolution of the Board of Directors of the Company) in order to provide for adjustments of any shares of common stock of such successor or acquiring Person for which this Warrant thus becomes exercisable, which modifications shall be as equivalent as practicable to the adjustments provided for in this SECTION 3.5. For purposes of this SECTION 3.5, "common stock of the successor or acquiring Person" shall include stock or other equity securities, or securities that are exercisable or exchangeable for or convertible into equity securities, of such corporation, or other securities if such Person is not a corporation, of any class that is not preferred as to dividends or assets over any other class of stock of such corporation or Person and that is not subject to redemption and shall also include any evidences of indebtedness, shares of stock or other securities that are convertible into or exchangeable for any such stock, either immediately or upon the arrival of a specified date or the happening of a specified event and any warrants or other rights to subscribe for or purchase any such stock. The foregoing provisions of this SECTION 3.5 shall similarly apply to successive reorganizations, reclassifications, consolidations, mergers, sales, transfers and other dispositions of assets.

3.6. DISSOLUTION, TOTAL LIQUIDATION OR WINDING-UP. If at any time there is a voluntary or involuntary dissolution, total liquidation or winding-up of the Company, other than as contemplated by SECTION 3.5, then the Company shall cause to be mailed (by registered or certified mail, return receipt requested, postage prepaid) to the Holder at the Holder's address as shown on the Warrant register, at the earliest practicable time (and, in any event, not less than thirty (30) calendar days before any date set for definitive action) written notice of the date on which such dissolution, liquidation or winding-up shall take place, as the case may be. Such notice shall also specify the date as of which the record holders of shares of Common Stock shall be entitled to exchange their shares for securities, money or other property deliverable upon such dissolution, liquidation or winding-up, as the case may be. On such date, the Holder shall be entitled to receive upon surrender of this Warrant the cash or other property, less the Warrant Purchase Price for this Warrant then in effect, that the Holder would have been entitled to receive had this Warrant been exercised immediately prior to such dissolution, liquidation or winding-up. Upon receipt of the cash or other property, any and all rights of the Holder to exercise this Warrant shall terminate in their entirety. If the cash or other property distributable in the dissolution, liquidation or winding-up has a fair market value (as determined in good faith by the board of directors of the Company and set forth in a written notice to the Holder, subject to the Holder's right to dispute such determination under SECTION 3.8(E)) which is less than the Warrant Purchase Price for this Warrant then in effect, this Warrant shall terminate and be of no further force or effect upon the dissolution, liquidation or winding-up.

3.7. OTHER DILUTIVE EVENTS. If any event occurs as to which the other provisions of this SECTION 3 are not strictly applicable but as to which the failure to make any adjustment would not protect the purchase rights represented by this Warrant in accordance with the intent and principles hereof then, in each such case, the Holder (or if the Warrant has been

divided up, the Holders of Warrants exercisable for the purchase of more than fifty percent (50%) of the aggregate number of Warrant Shares then issuable upon exercise of all of the then exercisable Warrants) may appoint an independent investment bank or firm of independent public accountants which shall give its opinion as to the adjustment, if any, on a basis consistent with the intent and principles established herein, necessary to preserve the purchase rights represented by this Warrant (or such Warrants). Upon receipt of such opinion, the Company will mail (by registered or certified mail, return receipt requested, postage prepaid) a copy thereof to the Holder within three (3) Business Days and shall make the adjustments described therein. The fees and expenses of such investment bank or independent public accountants shall be borne by the Company.

- 3.8. OTHER PROVISIONS APPLICABLE TO ADJUSTMENTS UNDER THIS SECTION. The following provisions shall be applicable to the adjustments provided for pursuant to this SECTION 3:
- (a) WHEN ADJUSTMENTS TO BE MADE. The adjustments required by this SECTION 3 shall be made whenever and as often as any specified event requiring such an adjustment shall occur. For the purpose of any such adjustment, any specified event shall be deemed to have occurred at the close of business on the date of its occurrence.
- (b) RECORD DATE. If the Company fixes a record date of the holders of Common Stock for the purpose of entitling them to (i) receive a dividend or other distribution payable in shares of Common Stock or in shares of any other class or series of capital stock or securities convertible into or exchangeable for Common Stock or shares of any other class or series of capital stock or (ii) subscribe for or purchase shares of Common Stock or such other shares or securities, then all references in this SECTION 3 to the date of the issuance or sale of such shares of Common Stock or such other shares or securities shall be deemed to be references to that record date.
- (c) WHEN ADJUSTMENT NOT REQUIRED. If the Company fixes a record date of the holders of its Common Stock for the purpose of entitling them to receive a dividend or distribution or subscription or purchase rights to which the provisions of SECTION 3.1 would apply, but shall, thereafter and before the distribution to shareholders, legally abandon its plan to pay or deliver such dividend, distribution, subscription or purchase rights, then thereafter no adjustment shall be required by reason of the taking of such record and any such adjustment previously made in respect thereof shall be rescinded and annulled.
- (d) NOTICE OF ADJUSTMENTS. Whenever the number of shares of Common Stock for which this Warrant is exercisable or the Warrant Purchase Price shall be adjusted or recalculated pursuant to this SECTION 3, the Company shall immediately prepare a certificate to be executed by the chief financial officer of the Company setting forth, in reasonable detail, the event requiring the adjustment or recalculation and the method by which such adjustment or recalculation was calculated, specifying the number of shares of Common Stock for which this Warrant is exercisable and (if such adjustment was made pursuant to SECTION 3.5) describing the number and kind of any other shares of stock or Other Property for which this Warrant is exercisable, and any related change in the Warrant Purchase Price, after giving effect to such adjustment, recalculation or change. The Company shall mail (by registered

or certified mail, return receipt requested, postage prepaid) a signed copy of the certificate to be delivered to the Holder within three (3) Business Days of the event which caused the adjustment or recalculation. The Company shall keep at the Designated Office copies of all such certificates and cause them to be available for inspection at the Designated Office during normal business hours by the Holder or any prospective transferee of this Warrant designated by the Holder.

- (e) CHALLENGE TO GOOD FAITH DETERMINATION. Whenever the Board of Directors of the Company is required to make a determination in good faith of the fair market value of any item under this Warrant, or any item that may affect the value of this Warrant, that determination may be challenged or disputed by the Holder (or if the Warrant has been divided up, the Holders of Warrants exercisable for more than fifty percent (50%) of the aggregate number of Warrant Shares then issuable upon exercise of all of the then exercisable Warrants), and any such challenge or dispute shall be resolved promptly, but in no event in more than thirty (30) days, by an investment banking firm of recognized national standing or one of the four (4) largest national accounting firms agreed upon by the Company and the Holders and whose decision shall be binding on the Company and the Holders. If the Company and the Holders cannot agree on a mutually acceptable investment bank or accounting firm, then the Holders, jointly, and the Company shall within five (5) Business Days each choose one investment bank or accounting firm and the respective chosen firms shall within five (5) Business Days jointly select a third investment bank or accounting firm, which shall make the determination promptly, but in no event in more than thirty (30) days, and such determination shall be binding upon all parties thereto. The Company shall bear all costs in connection with such determination, including without limitation, fees of the investment bank(s) or accounting firm(s), unless the change in Warrant Purchase Price determined pursuant to this paragraph is within five percent (5%) of the change in warrant purchase price as computed by the Company, in which case the Holder or Holders challenging the determination shall bear such costs pro rata.
- (f) INDEPENDENT APPLICATION. Except as otherwise provided herein, all subsections of this SECTION 3 are intended to operate independently of one another (but without duplication). If an event occurs that requires the application of more than one subsection, all applicable subsections shall be given independent effect without duplication.
- 3.9. FIDUCIARY DUTIES TO HOLDER. The board of directors of the Company shall at all times, whether or not this Warrant has been exercised, owe the Holder the same fiduciary duties that it would owe to a holder of the Warrant Shares underlying the Warrant.

4. MISCELLANEOUS.

4.1. RESTRICTIVE LEGEND. This Warrant, any Warrant issued upon transfer of this Warrant and, unless registered under the Securities Act, any Warrant Shares issued upon exercise of this Warrant or any portion thereof shall be imprinted with the following legend, in addition to any legend required under applicable state securities laws:

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY APPLICABLE STATE SECURITIES LAWS, AND MAY NOT BE SOLD, TRANSFERRED,

PLEDGED, HYPOTHECATED OR OTHERWISE ASSIGNED EXCEPT IN COMPLIANCE WITH THE REGISTRATION REQUIREMENTS OF SUCH ACT AND SUCH STATE SECURITIES LAWS OR PURSUANT TO AN EXEMPTION OR QUALIFICATION THEREFROM.

The legend shall be appropriately modified upon issuance of certificates for shares of Common Stock.

Upon request of the holder of a Common Stock certificate, the Company shall issue to that holder a new certificate free of the foregoing legend, if, with such request, such holder provides the Company with an opinion of counsel reasonably acceptable to the Company (PROVIDED that Dewey & LeBoeuf LLP shall be deemed to be acceptable to the Company) to the effect that the securities evidenced by such certificate may be sold without restriction under Rule 144 (or any other rule permitting resales of securities without restriction) promulgated under the Securities Act.

- 4.2. HOLDER ENTITLED TO BENEFITS. This Warrant is the "Warrant" referred to in the Credit Agreement. The Holder is entitled to certain rights, benefits and privileges with respect to this Warrant and the Warrant Shares pursuant to the terms of this Warrant and the Registration Rights Agreement.
- 4.3. OTHER COVENANTS. The Company covenants and agrees that, as long as this Warrant remains outstanding or any Warrant Shares are issuable with respect to this Warrant, the Company will perform all of the following covenants for the express benefit of the Holder: (a) the Warrant Shares shall, upon issuance, be duly authorized, validly issued, fully paid and non-assessable shares of Common Stock; (b) the Holder shall, upon the exercise hereof in accordance with the terms hereof, receive good and marketable title to the Warrant Shares, free and clear of all voting and other trust arrangements to which the Company is a party or by which it is bound, preemptive rights of any shareholder, liens, encumbrances, equities and claims whatsoever, including, but not limited to, all Taxes, Liens and other charges with respect to the issuance thereof; (c) at all times prior to the Expiration Date, the Company shall have reserved for issuance a sufficient number of authorized but unissued shares of Common Stock, or other securities or property for which this Warrant may then be exercisable, to permit this Warrant (or if this Warrant has been divided, all outstanding Warrants) to be exercised in full; (d) the Company shall deliver to the Holder the information and reports delivered to any other holder of the Company's common stock; and (e) the Company shall provide the Holder with written notice of all corporate actions, including, without limitation, prior written notice of any dividends in the same manner and to the same extent as the shareholders of the Company.
- 4.4. ISSUE TAX. The issuance of shares of Common Stock upon the exercise of this Warrant shall be made without charge to the Holder for any stamp, documentary, issue or similar tax in respect thereof.
- 4.5. CLOSING OF BOOKS. The Company will at no time close its transfer books against the transfer of this Warrant or of any Warrant Shares in any manner which interferes with the timely exercise hereof.

- 4.6. NO VOTING RIGHTS; LIMITATION OF LIABILITY. Except as expressly set forth in this Warrant or the Registration Rights Agreement, nothing contained in this Warrant shall be construed as conferring upon the Holder (a) the right to vote or to consent as a shareholder in respect of meetings of shareholders for the election of directors of the Company or any other matter, or (b) the right to receive dividends except as set forth in Section 3. No provisions hereof, in the absence of affirmative action by the Holder to purchase shares of Common Stock, and no mere enumeration herein of the rights or privileges of the Holder, shall give rise to any liability of the Holder for the Warrant Purchase Price or as a shareholder of the Company, whether such liability is asserted by the Company or by its creditors.
- 4.7. MODIFICATION AND WAIVER. This Warrant and any provision hereof may be changed, waived, discharged or terminated only by an instrument in writing signed by the party against which enforcement is sought.
- 4.8. NOTICES. All notices, requests, demands and other communications which are required or may be given under this Warrant shall be in writing and shall be deemed to have been duly given if transmitted by telecopier with receipt acknowledged, or upon delivery, if delivered personally or by recognized commercial courier with receipt acknowledged, or upon receipt, if mailed by registered or certified mail, return receipt requested, postage prepaid, addressed as follows:

If to the Company: Consumer Portfolio Services, Inc.

19500 Jamboree Road Irvine, California 92612 Attention: General Counsel Telecopier No.: (888) 577-7923 Telephone No.: (888) 785-6691

If to the initial Holder: Drawbridge Special Opportunities Fund LP

1345 Avenue of the Americas

46th Floor

New York, New York 10105

Attention: Constantine Dakolias

Facsimile No.: (212) 798-6131

Confirmation No.: (212) 798-6100

or at such other address or addresses as the Holder or the Company, as the case may be, may specify by written notice given in accordance with this SECTION 4.8.

4.9. SUCCESSORS AND ASSIGNS. The Company may not assign any of its rights, or delegate any of its obligations, under this Warrant without the prior written consent of the Holder (which consent may be withheld for any reason or no reason at all). The Holder may make an Assignment of this Warrant, in whole or in part, at any time or from time to time, without the consent of the Company. Each Assignment of this Warrant, in whole or in part, shall be registered on the books of the Company to be maintained for such purpose, upon surrender of this Warrant at the Designated Office, together with appropriate instruments of assignment, duly

completed and executed. Upon such surrender, the Company shall, at its own expense, within three (3) Business Days of surrender, execute and deliver a new Warrant or Warrants in the name of the assignee or assignees specified in such assignment and in the denominations specified therein and this Warrant shall promptly be canceled. If any portion of this Warrant is not being assigned, the Company shall, at its own expense, within three (3) Business Days issue to the Holder a new Warrant evidencing the part not so assigned. If the Holder makes an Assignment of this Warrant to one or more Persons, any decisions that the Holder is entitled to make at any time hereunder shall be made by the Holders holding more than fifty percent (50%) of the aggregate number of Warrant Shares issuable upon exercise of all of the then exercisable Warrants.

In addition, the Holder may, without notice to or the consent of the Company, grant or sell Participations to one or more participants in all or any part of its right, title and interest in and to this Warrant.

This Warrant shall be binding upon and inure to the benefit of the Company, the Holder and their respective successors and permitted assigns, and shall include, with respect to the Company, any Person succeeding the Company by merger, consolidation, combination or acquisition of all or substantially all of the Company's assets, and in such case, except as expressly provided herein, all of the obligations of the Company hereunder shall survive such merger, consolidation, combination or acquisition.

- 4.10. CONSTRUCTION AND INTERPRETATION. The headings of the paragraphs of this Warrant are for convenience of reference only and do not constitute a part of this Warrant and are not to be considered in construing or interpreting this Warrant. No party, nor its counsel, shall be deemed the drafter of this Warrant for purposes of construing the provisions of this Warrant, and all provisions of this Warrant shall be construed in accordance with their fair meaning, and not strictly for or against any party.
- 4.11. LOST WARRANT OR CERTIFICATES. Upon receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant or of a stock certificate evidencing Warrant Shares and, in the case of any such loss, theft or destruction, upon receipt of an indemnity reasonably satisfactory to the Company or, in the case of any such mutilation, upon surrender and cancellation of the Warrant or stock certificate, the Company shall make and deliver to the Holder, within three (3) Business Days of receipt by the Company of such documentation, a new Warrant or stock certificate, of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant or stock certificate.
- 4.12. NO IMPAIRMENT. The Company shall not by any action, including, without limitation, amending its charter documents or regulations or through any reorganization, reclassification, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such actions as may be necessary or appropriate to protect the rights of the Holder against impairment. Without limiting the generality of the foregoing, the Company will (i) not increase the par value (if any) of any shares of Common Stock receivable upon the exercise of this Warrant above the amount payable therefor upon such exercise immediately prior to such increase in par value, (ii) take all such action as may be necessary or appropriate in

order that the Company may validly and legally issue fully paid and nonassessable shares of Common Stock upon the exercise of this Warrant, free and clear of all liens, encumbrances, equities and claims, and (iii) use its best efforts to obtain all such authorizations, exemptions or consents from any public regulatory body having jurisdiction thereof as may be necessary to enable the Company to perform its obligations under this Warrant.

- 4.13. NO IMPLIED WAIVER. Nothing in this Warrant, including any reference herein to an act or transaction, shall be construed as, or imply, a waiver of any provision of the Credit Agreement or the Registration Rights Agreement, including any prohibition therein against such an act or transaction.
- 4.14. GOVERNING LAW. IN ALL RESPECTS, INCLUDING ALL MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE, THIS WARRANT AND THE RIGHTS AND OBLIGATIONS ARISING HEREUNDER SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE INTERNAL LAWS OF THE STATE OF NEW YORK APPLICABLE TO CONTRACTS MADE AND PERFORMED IN THAT STATE (WITHOUT REGARD TO THE CHOICE OF LAW OR CONFLICTS OF LAW PROVISIONS THEREOF) AND ANY APPLICABLE LAWS OF THE UNITED STATES OF AMERICA.
- 4.15. CONSENT TO JURISDICTION AND VENUE. ANY SUIT, LEGAL ACTION OR SIMILAR PROCEEDING WITH RESPECT TO THIS WARRANT SHALL BE BROUGHT IN THE COURTS OF THE STATE OF CALIFORNIA SITTING IN THE CITY OF LOS ANGELES, STATE OF CALIFORNIA OR OF THE UNITED STATES FOR THE CENTRAL DISTRICT OF CALIFORNIA SITTING IN THE CITY OF LOS ANGELES, STATE OF CALIFORNIA, AND BY EXECUTION AND DELIVERY OF THIS WARRANT, EACH OF THE COMPANY AND THE HOLDER (BY ACCEPTANCE HEREOF) CONSENTS, FOR ITSELF AND IN RESPECT OF ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF SUCH COURTS. EACH OF THE COMPANY AND THE HOLDER (BY ACCEPTANCE HEREOF) IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY OBJECTION, INCLUDING ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS, WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY ACTION OR PROCEEDING IN SUCH JURISDICTION IN RESPECT OF THIS WARRANT. EACH OF THE COMPANY AND THE HOLDER (BY ACCEPTANCE HEREOF) HEREBY WAIVES PERSONAL SERVICE OF ANY SUMMONS, COMPLAINT OR OTHER PROCESS, WHICH MAY BE MADE BY ANY OTHER MEANS PERMITTED BY THE LAW OF THE STATE OF THE NEW YORK.
- 4.16. REMEDIES. If the Company fails to perform, comply with or observe any covenant or agreement to be performed, complied with or observed by it under this Warrant, the Holder may proceed to protect and enforce its rights by suit in equity or action at law, whether for specific performance of any term contained in this Warrant or for an injunction against the breach of any such term or in aid of the exercise of any power granted in this Warrant or to enforce any other legal or equitable right, or to take any one or more of such actions. The Company hereby agrees that the Holder shall not be required or otherwise obligated to, and hereby waives any right to demand that the Holder, post any performance or other bond in

connection with the enforcement of its rights and remedies hereunder. The Company agrees to pay all fees, costs, and expenses, including, without limitation, fees and expenses of attorneys, accountants and other experts retained by the Holder, and all fees, costs and expenses of appeals, incurred or expended by the Holder in connection with the enforcement of this Warrant or the collection of any sums due hereunder, whether or not suit is commenced. None of the rights, powers or remedies conferred under this Warrant shall be mutually exclusive, and each right, power or remedy shall be cumulative and in addition to any other right, power or remedy whether conferred by this Warrant or now or hereafter available at law, in equity, by statute or otherwise.

- 4.17. ENTIRE AGREEMENT. This Warrant and the Registration Rights Agreement constitutes the full and entire agreement and understanding between the Holder and the Company with respect to the subject matter hereof and supersede all prior oral and written, and all contemporaneous oral, agreements and understandings relating to the subject matter hereof.
- 4.18. WAIVER OF JURY TRIAL. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAWS, THE COMPANY HEREBY KNOWINGLY, INTENTIONALLY AND VOLUNTARILY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING UNDER THIS WARRANT OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE COMPANY AND THE HOLDER WITH RESPECT TO THE WARRANT, OR THE TRANSACTIONS RELATED THERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE; AND THE COMPANY HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT THE COMPANY OR ANY HOLDER MAY FILE AN ORIGINAL COPY OF THIS SECTION 4.18 WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE COMPANY HERETO TO THE WAIVER OF ITS RIGHT TO TRIAL BY JURY.

IN WITNESS WHEREOF, the Company has caused this Warrant to be executed and issued by its duly authorized representative on the date first written above.

CONSUMER PORTFOLIO SERVICES, INC., a California corporation

By: <u>/s/ CHARLES E. BRADLEY, JR.</u>
Charles E. Bradley, Jr.
President and Chief Executive Officer

REVOLVING CREDIT AGREEMENT

dated as of September 25, 2009

	n		ì	
a	11	Ц	"	17

PAGE FOUR FUNDING LLC,

CONSUMER PORTFOLIO SERVICES, INC.,

and

FORTRESS CREDIT CORP.,

as Administrative Agent, Collateral Agent and Lender,

\$50,000,000 Secured Revolving Credit Facility

REVOLVING CREDIT AGREEMENT

This **REVOLVING CREDIT AGREEMENT**, dated as of September 25, 2009, is entered into by and among **PAGE FOUR FUNDING LLC**, a Delaware limited liability company (the "**Borrower**"), **CONSUMER PORTFOLIO SERVICES, INC.**, a California corporation, ("**CPS**"), and **FORTRESS CREDIT CORP**. ("**Fortress**"), as Administrative Agent (in such capacity, the "**Administrative Agent**"), Collateral Agent (in such capacity, the "**Collateral Agent**") and as Lender.

RECITALS:

WHEREAS, the Lender has agreed to extend a credit facility (the "Facility") to the Borrower, consisting of up to \$50,000,000 aggregate principal amount of Revolving Commitments, the proceeds of which will be used by the Borrower to acquire Receivables and to pay fees and expenses related to the foregoing; and

WHEREAS, the Borrower has agreed to secure all of its Obligations by granting to the Collateral Agent, for the benefit of the Secured Parties, a first priority Lien on all of its assets;

NOW, THEREFORE, in consideration of the premises and the agreements, provisions and covenants herein contained, the parties hereto agree as follows:

SECTION DEFINITIONS AND INTERPRETATION

1.

1.1. Definitions. The following terms used herein, including in the preamble, recitals, exhibits and schedules hereto, shall have the following meanings:

"Act" as defined in Section 4.22.

"Adjusted LIBOR Rate" means, for any Interest Period for any LIBOR Rate Loan made or continued during such Interest Period, the per annum rate equal to the greater of (i) 2.00% per annum and (ii)(x) the rate per annum (rounded to the nearest 1/100 of 1%) equal to the rate determined by the Administrative Agent to be the offered rate which appears on the page of the Reuters Screen which displays an average British Bankers Association Interest Settlement Rate (such page currently being Reuters Screen LIBOR01 Page) for deposits (for delivery on the first day of such period) for a one-month period in Dollars, determined as of approximately 11:00 a.m. (London, England time) on the related Interest Rate Reset Date, or (y) in the event the rate referenced in the preceding clause (x) does not appear on such page or service or if such page or service shall cease to be available, the rate per annum (rounded to the nearest 1/100 of 1%) equal to the rate determined by the Administrative Agent to be the offered rate on such other page or other service which displays an average British Bankers Association Interest Settlement Rate for deposits (for delivery on the first day of such period) for a one-month period in Dollars, determined as of approximately 11:00 a.m. (London, England time) on the related Interest Rate Reset Date, or (z) in the event the rates referenced in the preceding clauses (x) and (y) are not available, the rate per annum (rounded to the nearest 1/100 of 1%) equal to the offered quotation rate to first class banks in the London interbank market for deposits (for delivery on the first day of the relevant period) in Dollars of amounts in same day funds comparable to the principal amount of the applicable Revolving Loan for which the Adjusted LIBOR Rate is then being

determined with maturities equal to a one-month period as of approximately 11:00 a.m. (London, England time) on such Interest Rate Reset Date.

"Adjusted Tangible Net Worth" means, with respect to any fiscal quarter, the total shareholders' equity of the Servicer and its consolidated Subsidiaries that, in accordance with GAAP, is reflected on the consolidated balance sheet of the Servicer and its consolidated Subsidiaries as of the end of such fiscal quarter, minus the amount equal to the net deferred tax assets of the Servicer and its consolidated Subsidiaries reflected on such consolidated balance sheet, plus the amount equal to the net deferred tax assets of the Servicer and its consolidated Subsidiaries reflected on the consolidated balance sheet of the Servicer and its consolidated Subsidiaries as of December 31, 2008 (which amount is \$52,727,000), minus the aggregate amount of the Servicer's and its consolidated Subsidiaries' intangible assets, including without limitation, goodwill, franchises, licenses, patents, trademarks, tradenames, copyrights and service marks.

"**Administrative Agent**" as defined in the preamble hereto.

"Adverse Proceeding" means, with respect to any Person, any action, suit, proceeding (whether administrative, judicial or otherwise), governmental investigation or arbitration (whether or not purportedly on behalf of such Person) at law or in equity, or before or by any Governmental Authority, domestic or foreign, whether pending or, to the knowledge of such Person, threatened against or affecting such Person or its properties.

"Affected Loans" as defined in Section 2.14(b).

"Affiliate" means, as applied to any Person, any other Person directly or indirectly controlling (including any member of senior management of such Person), controlled by, or under common control with, that Person. For the purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as applied to any Person, means the possession, directly or indirectly, of the power (a) to vote 20% or more of the Securities having ordinary voting power for the election of directors of such Person or (b) to direct or cause the direction of the management and policies of that Person, whether through the ownership of voting securities or by contract or otherwise.

"Agent" means each of the Administrative Agent and the Collateral Agent.

"Agreement" means this Revolving Credit Agreement, dated as of September 25, 2009, as it may be amended, supplemented or otherwise modified from time to time in accordance with the terms hereof.

"Allowed Financing" as defined in Section 9.22.

"Annualized Third Party Fee Rate" means, with respect to any Collection Period, a per annum percentage equal to a fraction (a) the numerator of which is equal to sum of (i) the aggregate amount paid to the Custodian with respect to the Custodian Fees for such Collection Period, (ii) the aggregate amount paid to the Backup Servicer with respect to the Backup Servicing Fee for such Collection Period and (iii) the aggregate amount paid to the Controlled

Account Bank with respect to the Controlled Account Bank Fee, and (b) the denominator of which is the aggregate principal balance of the Eligible Receivables as of the first day of such Collection Period.

"Applicable Margin" as defined in the Fee Letter.

- "APR" means, with respect to a Receivable, the annual percentage rate of finance charges stated in such Receivable; <u>provided</u> that if the annual percentage rate with respect to such Receivable, is reduced as a result of (i) an insolvency proceeding involving the related Obligor or (ii) pursuant to the Servicemembers Civil Relief Act, the APR shall refer to such reduced rate.
- "Assignment Agreement" means an Assignment and Assumption Agreement substantially in the form of Exhibit D, with such amendments or modifications as may be approved by the Administrative Agent.
- "Authorized Officer" means, as applied to any Person, any individual holding the position of chairman of the board (if an officer), chief executive officer, president or one of its vice presidents (or the equivalent thereof), and such Person's chief financial officer or treasurer.
- "Average Delinquency Rate" means, with respect to any Vintage Pool and any Reporting Date, the arithmetic average of the Delinquency Rate for each of the three (3) Collection Periods immediately preceding the month in which such Reporting Date occurs.
- "Average Elapsed Period" means, with respect to any Vintage Pool and any Reporting Date, the number of months elapsed between (x) the first day of the second month of the applicable calendar quarter of origination with respect to such Vintage Pool and (y) the first day of the month in which such Reporting Date occurs.
- "Average Maximum Advance Rate" means, with respect to any date of determination, the weighted average of the Maximum Advance Rates for the Eligible Receivables as of the last calculation of the Borrowing Base.
- "Average Repossession Rate" means, with respect to any Reporting Date, the arithmetic average of the Repossession Rate for each of the three (3) Collection Periods immediately preceding the month in which such Reporting Date occurs.
- "Backup Servicer" means Wells Fargo Bank, National Association, or any independent third party selected by the Administrative Agent, in its reasonable discretion, to perform monitoring functions with respect to the Receivables.
- "Backup Servicing Agreement" means that certain Backup Servicing Agreement, dated as of September 25, 2009, by and among the Backup Servicer, the Servicer and Fortress, as Controlling Party, as may be further amended, modified or supplemented from time to time.

"Backup Servicing Fees" as defined in the Backup Servicing Agreement.

"Bankruptcy Code" means Title 11 of the United States Code entitled "Bankruptcy," as now and hereafter in effect, or any successor statute.

"Base Rate" means, for any day, a rate per annum equal to the greater of (i) the Prime Rate in effect on such day, and (ii) the Federal Funds Effective Rate in effect on such day plus ½ of 1%. Any change in the Base Rate due to a change in the Prime Rate or the Federal Funds Effective Rate shall be effective day of such change in the Prime Rate or the Federal Funds Effective Rate, respectively.

"Base Rate Loan" means a Revolving Loan bearing interest at a rate determined by reference to the Base Rate.

"Borrower" as defined in the preamble hereto.

"Borrowing Base" means, subject to Section 2.18(a), as of any date of determination, the sum of:

- (a) the Maximum Advance for such date; plus
- (b) the aggregate amount then on deposit in the Collection Account; minus
- (c) all interest, fees, expenses (exclusive of legal fees and expenses) and indemnification amounts with respect to the Facility payable on the immediately succeeding Settlement Date.

"Borrowing Base Certificate" means a certificate, substantially in the form of Exhibit C, executed by an Authorized Officer of the Borrower and delivered to the Collateral Agent and the Administrative Agent, which sets forth the calculation of the Borrowing Base, including a calculation of each component thereof.

"Borrowing Base Deficiency" means the amount (if any) by which the Total Utilization of Revolving Commitments exceeds the lesser of (a) the Revolving Commitments then in effect and (b) the Borrowing Base. For the avoidance of doubt, a "Borrowing Base Deficiency" shall be deemed to exist at any time following the occurrence and continuance of an Event of Default, to the extent that any deficiency balance exists with respect to the Obligations remaining due and payable by the Borrower.

"Borrowing Base Guaranty" means that certain Guaranty dated as of September 25, 2009 by CPS in favor of the Administrative Agent, on behalf of the Lender, as may be further amended, modified or supplemented from time to time in accordance with the terms thereof.

"Business Day" means (a) any day excluding Saturday, Sunday and any day which is a legal holiday under the laws of the State of New York, the State of Minnesota or the State of California or is a day on which banking institutions located in either such state are authorized or required by law or other governmental action to close, and (b) with respect to all notices, determinations, fundings and payments in connection with the Adjusted LIBOR Rate or any LIBOR Rate Loans, the term "Business Day" shall mean any day which is a Business Day described in clause (a) and which is also a day for trading by and between banks in Dollar deposits in the London interbank market.

"Capital Lease" means, as applied to any Person, any lease of any property (whether real, personal or mixed) by that Person (a) as lessee that, in conformity with GAAP, is or should be accounted for as a capital lease on the balance sheet of that Person or (b) as lessee which is a transaction of a type commonly known as a "synthetic lease" (i.e., a transaction that is treated as an operating lease for accounting purposes but with respect to which payments of rent are intended to be treated as payments of principal and interest on a loan for Federal income tax purposes).

"Capital Stock" means any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all equivalent ownership interests in a Person (other than a corporation), including, without limitation, partnership interests and membership interests, and any and all warrants, rights or options to purchase or other arrangements or rights to acquire any of the foregoing.

"Cash" means money, currency or a credit balance in any demand or deposit account.

"Cash Equivalents" means, as at any date of determination, (a) marketable securities (i) issued or directly and unconditionally guaranteed as to interest and principal by the United States Government, or (ii) issued by any agency of the United States the obligations of which are backed by the full faith and credit of the United States, in each case maturing within one (1) year after such date; (b) marketable direct obligations issued by any state of the United States of America or any political subdivision of any such state or any public instrumentality thereof, in each case maturing within one (1) year after such date and having, at the time of the acquisition thereof, a rating of at least A-1 from S&P or at least P-1 from Moody's; (c) commercial paper maturing no more than one (1) year from the date of creation thereof and having, at the time of the acquisition thereof, a rating of at least A-1 from S&P or at least P-1 from Moody's; (d) certificates of deposit or bankers' acceptances maturing within one (1) year after such date and issued or accepted by the Lender or by any commercial bank organized under the laws of the United States of America or any state thereof or the District of Columbia that (i) is at least "adequately capitalized" (as defined in the regulations of its primary Federal banking regulator), and (ii) has Tier 1 capital (as defined in such regulations) of not less than \$100,000,000; and (e) shares of any money market mutual fund that (i) has substantially all of its assets invested continuously in the types of investments referred to in clauses (a) and (b) above, (ii) has net assets of not less than \$500,000,000, and (iii) has the highest rating obtainable from either S&P or Moody's.

"Change of Control" means, at any time, (i) with respect to the Borrower, CPS shall cease to beneficially own and control 100% on a fully diluted basis of the economic and voting interest in the Capital Stock of the Borrower and (ii) with respect to CPS, the acquisition by any Person, or two (2) or more Persons acting in concert, of beneficial ownership (within the meaning of Rule 13d-3 of the Securities and Exchange Commission under the Securities Exchange Act of 1934) of outstanding shares of voting stock of CPS, as applicable, at any time if after giving effect to such acquisition such Person or Persons owns fifty percent (50%) or more of such outstanding voting stock.

"Charge-Off Receivable" means any Receivable (or any other automobile receivable acquired or originated by the Originator) with respect to which the earlier of any of the following

shall have occurred (without duplication): (i) the Receivable has been liquidated by the Servicer through the sale of the Financed Vehicle, (ii) the related Obligor has failed to make a Scheduled Receivable Payment by its due date and such failure continues for one hundred and twenty (120) days (or, if the related Financed Vehicle has been repossessed, two hundred and ten (210) days), (iii) ninety (90) days following the repossession of the related Financed Vehicle by the Servicer, (iv) the related Obligor is subject to a proceeding under the Bankruptcy Code or other applicable Debtor Relief Laws and the related Receivable is not a performing contract, (v) the related Obligor is deceased, (vi) proceeds have been received which, in the Servicer's good faith judgment, constitute the final amounts recoverable in respect of such Receivable, or (vii) the Servicer has otherwise determined, in accordance with its Collection Policy, that the related Receivable should be charged-off.

"Closing Date" means the date on which the initial Revolving Loans are made.

"Closing Date Certificate" means a Closing Date Certificate substantially in the form of Exhibit E-1.

"Closing Date Material Adverse Change" means a material adverse change in (i) the business operations, assets, condition (financial or otherwise), liabilities or prospects of any Credit Party or the Originator, since August 31, 2009; (ii) the ability of the Borrower to fully and timely perform its material Obligations under any of the Credit Documents to which it is a party, or the legality, validity, binding effect, or enforceability against the Borrower of any such Credit Documents; or (iii) the ability of CPS to fully and timely perform its material obligations under the Credit Documents to which it is a party, or the legality, validity, binding effect, or enforceability against CPS of any such Credit Documents.

"Collateral" means, collectively, all of the real, personal and mixed property in which Liens are purported to be granted pursuant to the Collateral Documents as security for the Obligations.

"Collateral Agent" as defined in the preamble hereto.

"Collateral Documents" means the Security Agreement, the Control Agreements and all other instruments, documents and agreements delivered by any Credit Party pursuant to this Agreement or any of the other Credit Documents in order to grant to the Collateral Agent, for the benefit of Secured Parties, a Lien on any real, personal or mixed property of that Credit Party, as the case may be, as security for the Obligations.

"Collateral Receipt and Exception Report" as defined in the Custodial Agreement.

"Collection Account" as defined in the Security Agreement.

"Collection Period" means, (i) with respect to the initial Settlement Date, the period beginning on the Closing Date and ending on the last day of the immediately preceding calendar month and (ii) with respect to any other Settlement Date, the immediately preceding calendar month.

"Collection Policy" as defined in the Servicing Agreement.

"Collections" means all collections on the Receivables, including, without limitation, all Scheduled Receivable Payments, all non-scheduled payments, all prepayments, all late fees, all other fees, all insurance proceeds, all Liquidation Proceeds, all Recoveries, all payments received under any Interest Rate Agreement, investment earnings, rental payments, residual proceeds, payments received under any personal guaranty with respect to a Receivable and all other payments received with respect to the Receivables, but excluding sales and property tax payments.

"Contractual Obligation" means, as applied to any Person, any provision of any Security issued by that Person or of any indenture, mortgage, deed of trust, contract, undertaking, agreement or other instrument to which that Person is a party or by which it or any of its properties is bound or to which it or any of its properties is subject.

"Control Agreements" means, collectively, the Lockbox Control Agreement and the Controlled Account Control Agreement.

"Controlled Account" as defined in Section 2.11(a).

"Controlled Account Bank" means Wells Fargo Bank, National Association, in its capacity as account bank under the Controlled Account Control Agreement, and its successors and assigns.

"Controlled Account Bank Fee" as defined in the Controlled Account Control Agreement.

"Controlled Account Control Agreement" as defined in the Security Agreement.

"CPS" means Consumer Portfolio Services, Inc., a California corporation.

"Credit Date" means the date of a Credit Extension.

"Credit Document" means any of (a) this Agreement, the Fee Letter, the Revolving Loan Notes, if any, the Collateral Documents, the Related Agreements, the Warrants, the Borrowing Base Guaranty and the Limited Guaranty and (b) all other documents, instruments or agreements executed and delivered by a Credit Party for the benefit of any Agent or the Lender in connection herewith.

"Credit Extension" means the making of a Revolving Loan.

"Credit Party" means the Borrower and CPS.

"Credit Score" means the applicable credit score, for each primary Obligor, as determined by Equifax, Inc. or a comparable credit bureau.

"Cumulative Net Loss Rate" means, as of any Reporting Date and with respect to any Vintage Pool, a rate, expressed as a percentage equal to a fraction, (I) the numerator of which is the Cumulative Net Losses with respect to all automobile receivables acquired or originated by the Originator in the related Vintage Pool and (II) the denominator of which is the aggregate

principal balance of all automobile receivables acquired or originated by the Originator in the related Vintage Pool at the time of origination or acquisition by the Originator; provided that, if any sale or securitization of receivables by the Originator occurs on a servicing released basis, the Originator and the Administrative Agent shall negotiate in good faith to amend the calculation of Cumulative Net Loss Rate if such sale or securitization results in an inability to calculate the Cumulative Net Loss Rate because of lack of information upon which to make such calculation.

"Cumulative Net Losses" means, as of any date of determination and with respect to any Vintage Pool, the aggregate cumulative principal amount of automobile receivables acquired or originated by the Originator that have become Charge-Off Receivables during the period beginning on the applicable date of origination through the end of the Collection Period immediately preceding the month in which such date of determination occurs, net of all Net Liquidation Proceeds with respect to such receivables as of the end of the Collection Period immediately preceding the month in which such date of determination occurs.

"Custodial Agreement" means that certain Custodial and Collateral Agency Agreement dated as of September 25, 2009 by and among the Borrower, the Servicer, Custodian and the Collateral Agent, as may be further amended, modified or supplemented from time to time in accordance with the terms thereof.

"Custodian" means Wells Fargo Bank, National Association, in its capacity as custodian under the Custodial Agreement, or any successor thereto acceptable to the Administrative Agent in its sole discretion.

"Custodian Fee" as defined in the Custodial Agreement.

"Custodian Fee and Expenses" as defined in the Custodial Agreement.

"Dealer" means, with respect to a Receivable, the seller of the related Financed Vehicle, who originated and assigned such Receivable to the Originator pursuant to a Dealer Agreement.

"Dealer Agreement" means each agreement between a Dealer and the Originator pursuant to which such Dealer assigned, sold or otherwise conveyed a Receivable to the Originator.

"Debtor Relief Laws" means the Bankruptcy Code, and all other applicable liquidation, conservatorship, bankruptcy, moratorium, rearrangement, receivership, insolvency, reorganization, suspension of payments, readjustment of debt, marshalling of assets, assignment for the benefit of creditors or similar debtor relief laws of the United States, any state or any foreign country from time to time in effect, affecting the rights of creditors generally or the rights of creditors of banks.

"Default" means a condition or event that, after notice or lapse of time or both, would constitute an Event of Default.

"**Default Rate**" means the interest rate provided in Section 2.6.

"Defaulted Receivable" means, with respect to any date of determination, a Receivable with respect to which the earliest of any of the following shall have occurred: (i) the related Obligor has failed to make the first periodic payment within thirty (30) days following the last day of the calendar month during which such Receivable's first payment was due under the terms of such Receivable, (ii) the related Obligor has failed to make any Scheduled Receivable Payment by its due date and such failure continues for one hundred and twenty (120) days, (iii) the Servicer has repossessed the related Financed Vehicle (and any applicable redemption period has expired) (iv) such Receivable is in default and the Servicer has determined in good faith that payments thereunder are not likely to be resumed or (v) such Receivable is a Charge-off Receivable.

"Delinquency Rate" means, with respect to any date of determination and any Vintage Pool, a rate, expressed as a percentage, equal to a fraction (I) the numerator of which is the aggregate outstanding principal balance of all automobile receivables acquired or originated by the Originator in such Vintage Pool that are Delinquent Receivables as of the last day of the most recently ended Collection Period and (II) the denominator of which is the aggregate outstanding principal balance of all automobile receivables acquired or originated by the Originator in such Vintage Pool as of the last day of the most recently ended Collection Period; provided that, if any sale or securitization of receivables by the Originator occurs on a servicing released basis, the Originator and the Administrative Agent shall negotiate in good faith to amend the calculation of Delinquency Rate if such sale or securitization results in an inability to calculate the Delinquency Rate because of lack of information upon which to make such calculation.

"**Delinquent Receivable**" means, with respect to any date of determination, a Receivable (or any other automobile receivable acquired or originated by the Originator) with respect to which the related Obligor is more than thirty (30) days but less than one hundred and twenty (120) days past due with respect to 10% or more of a Scheduled Receivable Payment.

"Depository Institution" means, collectively, any "depository institution" or any "subsidiary" of a depository institution, as such terms are defined in the Federal Deposit Insurance Act of 1950, as amended to date.

"Dollars" and the sign "\$" mean the lawful money of the United States of America.

"Effective Advance Rate" means a fraction (i) the numerator of which is equal to the Total Utilization of Revolving Commitments and (ii) the denominator of which is the aggregate principal balance of all Eligible Receivables.

"Eligible Dealer" means a Dealer that satisfies the following criteria: (a) the related Dealer Agreement provides for full recourse to the Dealer in the event of any fraud or misrepresentation on the part of the Dealer, (b) to the extent applicable, the Dealer has obtained all applicable Governmental Authorizations, and (c) the Dealer otherwise qualifies as an "Eligible Dealer" in accordance with the Underwriting Policies.

"Eligible Hedge Counterparty" means (i) the Lender or any of its Affiliates or (ii) any party that enters into an Interest Rate Agreement with the Borrower; provided, that such hedge

counterparty must maintain long term senior unsecured debt ratings of "A" and "A2" by S&P and Moody's, respectively, and be otherwise acceptable to the Administrative Agent.

"Eligible Obligor" means an Obligor that (a) is not currently in bankruptcy, (b) is not a party to more than one (1) installment loan contract with the Originator or any of its Affiliates, (c) is not an employee, or affiliated with any employee of, the Originator or any of its Affiliates and (d) is domiciled in the United States.

"Eligible Receivable" means a Receivable with respect to which the Eligibility Criteria are satisfied as of any date of determination.

"Eligibility Criteria" means the criteria set forth on Appendix B.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and any successor thereto.

"**Event of Default**" means any of the conditions or events set forth in Section 7.1.

"Excess Concentration Amounts" means, each of the amounts set forth on Appendix C hereto, which amounts shall be excluded from the calculation of the aggregate unpaid principal balance of Eligible Receivables.

"Excess Interest Rate" means, with respect to any Collection Period, a rate equal to (i) the weighted average APR of the Eligible Receivables, minus (ii) the Servicing Fee Rate, minus (iii) the Annualized Third Party Fee Rate, minus (iv) the product of (x) the applicable Interest Rate for such Collection Period (provided, that, in the event the Interest Rate is determined by reference to the Base Rate, the Interest Rate for a Collection Period shall be the weighted average of the Base Rates in effect during such Collection Period) and (y) the Effective Advance Rate.

"Excess Spread Account Amounts" means amounts, if any, in the Spread Account in excess of the Required Spread Account Amount on any date of determination.

"Exchange Act" means the Securities Exchange Act of 1934, as amended to the date hereof and from time to time hereafter, and any successor statute.

"Exclusivity Covenant" as defined in Section 9.22.

"**Facility**" as defined in the preamble hereto.

"Federal Funds Effective Rate" means for any day, the rate per annum (expressed, as a decimal, rounded upwards, if necessary, to the next higher 1/100 of 1%) equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the next succeeding Business Day; provided, (a) if such day is not a Business Day, the Federal Funds Effective Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal

Funds Effective Rate for such day shall be the average rate charged to Fortress on such day on such transactions as determined by the Administrative Agent.

"Fee Letter" means the letter agreement, dated as of the date hereof, by and among the Administrative Agent on behalf of the Lender, the Borrower and CPS.

"Final Bid" as defined in Section 9.22.

"**Financed Vehicle**" means a new or used automobile, van, minivan, sport utility vehicle or light duty truck, together with all accessions thereto, securing an Obligor's indebtedness under a Receivable.

"Fiscal Quarter" means, with respect to a particular Fiscal Year, a fiscal quarter corresponding to such Fiscal Year.

"**Fiscal Year**" means for the Borrower, any consecutive twelve-month period commencing on the date following the last day of the previous Fiscal Year and ending on December 31.

"Fortress" as defined in the Preamble hereto.

"**Fortress Affiliate Lender**" as defined in Section 9.6(j).

"Funding Notice" means a notice substantially in the form of Exhibit A.

"Funding Termination Event" means the occurrence of either of the following: (i) the Originator makes any material change to the Underwriting Guidelines without the prior written consent of the Administrative Agent or (ii) any two (2) Key Employees cease to be involved in the day-to-day operations of the Originator or are unable to work for three (3) consecutive months and are not replaced by successors acceptable to the Administrative Agent within sixty (60) days.

"GAAP" means, subject to the limitations on the application thereof set forth in Section 1.2, United States generally accepted accounting principles in effect as of the date of determination thereof.

"Governmental Authority" means any federal, state, municipal, national or other government, governmental department, commission, board, bureau, court, agency or instrumentality or political subdivision thereof or any entity or officer exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to any government or any court, in each case whether associated with a state of the United States, the United States, or a foreign entity or government.

"Governmental Authorization" means any permit, license, authorization, plan, directive, consent order or consent decree of or from any Governmental Authority.

"Grantor" as defined in the Security Agreement.

"Guarantor" means CPS.

"Highest Lawful Rate" means the maximum lawful interest rate, if any, that at any time or from time to time may be contracted for, charged, or received under the laws applicable to the Lender which are presently in effect or, to the extent allowed by law, under such applicable laws which may hereafter be in effect and which allow a higher maximum nonusurious interest rate than applicable laws now allow.

"Indebtedness," as applied to any Person, means, without duplication, (a) all indebtedness for borrowed money; (b) that portion of obligations with respect to Capital Leases that is properly classified as a liability on a balance sheet in conformity with GAAP; (c) notes payable and drafts accepted representing extensions of credit whether or not representing obligations for borrowed money; (d) any obligation owed for all or any part of the deferred purchase price of property or services (excluding any such obligations incurred under ERISA), which purchase price is (i) due more than six (6) months from the date of incurrence of the obligation in respect thereof or (ii) evidenced by a note or similar written instrument; (e) all indebtedness secured by any Lien on any property or asset owned or held by that Person regardless of whether the indebtedness secured thereby shall have been assumed by that Person or is nonrecourse to the credit of that Person; (f) the face amount of any letter of credit issued for the account of that Person or as to which that Person is otherwise liable for reimbursement of drawings; (g) the direct or indirect guaranty, endorsement (otherwise than for collection or deposit in the ordinary course of business), co-making, discounting with recourse or sale with recourse by such Person of the obligation of another; (h) any obligation of such Person the primary purpose or intent of which is to provide assurance to an obligee that the obligation of the obligor thereof will be paid or discharged, or any agreement relating thereto will be complied with, or the holders thereof will be protected (in whole or in part) against loss in respect thereof; (i) any liability of such Person for an obligation of another through any agreement (contingent or otherwise) (A) to purchase, repurchase or otherwise acquire such obligation or any security therefor, or to provide funds for the payment or discharge of such obligation (whether in the form of loans, advances, stock purchases, capital contributions or otherwise) or (B) to maintain the solvency or any balance sheet item, level of income or financial condition of another if, in the case of any agreement described under subclauses (A) or (B) of this clause (i), the primary purpose or intent thereof is as described in clause (h) above; and (j) all obligations of such Person in respect of any exchange traded or over the counter derivative transaction, including, without limitation, any Interest Rate Agreement, whether entered into for hedging or speculative purposes; provided, in no event shall obligations under any Interest Rate Agreement be deemed "Indebtedness" for any purpose under Section 6.1.

"Indemnified Liabilities" means, collectively, any and all liabilities, obligations, losses, damages, penalties, claims, costs, expenses and disbursements of any kind or nature whatsoever (including the reasonable, documented, out-of-pocket fees and disbursements of counsel for Indemnitees in connection with any investigative, administrative or judicial proceeding commenced by any Person, whether or not any such Indemnitee shall be designated as a party or a potential party thereto, and any reasonable, documented, out-of-pocket fees or expenses incurred by Indemnitees in enforcing the indemnification provisions of Section 9.3), whether direct, indirect or consequential and whether based on any federal, state or foreign laws, statutes, rules or regulations (including securities and commercial laws, statutes, rules or regulations, on

common law or equitable cause or on contract or otherwise) that may be imposed on, incurred by, or asserted against any such Indemnitee, in any manner relating to or arising out of this Agreement or the other Credit Documents or the transactions contemplated hereby or thereby (including the Lender's agreement to make Credit Extensions or the use or intended use of the proceeds thereof, or any enforcement of any of the Credit Documents (including any sale of, collection from, or other realization upon any of the Collateral or the enforcement of the Limited Guaranty or the Borrowing Base Guaranty)); provided, however, that "Indemnified Liabilities" shall not include any liabilities, obligations, losses, damages, penalties, claims, costs, expenses and disbursements resulting from credit losses on or diminution in value of Receivables or other Collateral unless such credit loss or diminution in value was a result of the action or inaction of the Borrower or the Servicer in contravention of the Credit Documents.

"Indemnitee" as defined in Section 9.3(a).

"**Independent Accountants**" means (a) Crowe Horwath LLP or (b) a firm of independent certified public accountants registered with the Public Company Accounting Oversight Board and otherwise acceptable to the Administrative Agent.

"**Independent Manager**" means an employee of Lord Securities Corporation, or another natural person meeting the qualifications set forth in Section 6.15 and otherwise acceptable to the Administrative Agent in its sole discretion.

"Indicative Bid" as defined in Section 9.22.

"**Interest Period**" means, (a) with regard to the initial Settlement Date, the period commencing on (and including) the Closing Date and ending on the last day of the immediately preceding calendar month; and (b) with regard to any other Settlement Date, the immediately preceding calendar month; provided, no Interest Period with respect to any portion of the Revolving Loans shall extend beyond the Revolving Commitment Termination Date.

"Interest Rate" means, with respect to (i) any Revolving Loan that is a LIBOR Rate Loan and any Interest Period, the Adjusted LIBOR Rate plus the Applicable Margin for such Interest Period and (ii) any Revolving Loan that is a Base Rate Loan and any Interest Period, the Base Rate plus the Applicable Margin for such Interest Period.

"Interest Rate Agreement" means any interest rate swap agreement, interest rate cap agreement, interest rate collar agreement, interest rate hedging agreement or other similar agreement or arrangement that (a) has been entered into by the Borrower for the purpose of hedging the interest rate exposure associated with the Borrower's operations and not for speculative purposes, (b) is entered into with an Eligible Hedge Counterparty, (c) is approved by the Administrative Agent in its reasonable discretion and (d) has been collaterally assigned to the Collateral Agent pursuant to an instrument satisfactory to the Administrative Agent.

"Interest Rate Reset Date" means, with respect to any Interest Period, the date that is two (2) Business Days prior to the first day of such Interest Period.

"Internal Revenue Code" means the Internal Revenue Code of 1986, as amended to the date hereof and from time to time hereafter, and any successor statute.

"Investment" means (a) any direct or indirect purchase or other acquisition by the Borrower of, or of a beneficial interest in, any of the Securities of any other Person; (b) any direct or indirect redemption, retirement, purchase or other acquisition for value, from any Person, of any Capital Stock of such Person; and (c) any direct or indirect loan, advance or capital contributions by the Borrower to any other Person, including all indebtedness and accounts receivable from that other Person that are not current assets or did not arise from sales to that other Person in the ordinary course of business. The amount of any Investment shall be the original cost of such Investment plus the cost of all additions thereto, without any adjustments for increases or decreases in value, or write-ups, write-downs or write-offs with respect to such Investment.

"Key Employee" means each of Charles Bradley, Jr., Robert Riedl and Chris Terry or any successor thereto approved by the Administrative Agent in its sole discretion.

"Lender" means Fortress and its successors and assigns.

"LIBOR Rate Loan" means a Revolving Loan bearing interest at a rate determined by reference to the Adjusted LIBOR Rate and maturing on the last day of each Interest Period (unless otherwise continued pursuant to Section 2.5).

"LIBOR Unavailability" as defined in Section 2.14(a).

"Lien" means (a) any lien, mortgage, pledge, assignment, security interest, charge or encumbrance of any kind (including any agreement to give any of the foregoing, any conditional sale or other title retention agreement, and any lease in the nature thereof) and any option, trust or other preferential arrangement having the practical effect of any of the foregoing and (b) in the case of Securities, any purchase option, call or similar right of a third party with respect to such Securities.

"**Limited Guaranty**" means that certain Limited Guaranty dated as of September 25, 2009 by CPS in favor of the Administrative Agent, on behalf of the Lender, as may be further amended, modified or supplemented from time to time in accordance with the terms thereof.

"Liquidation Proceeds" as defined in the Servicing Agreement.

"Lockbox Account" as defined in the Security Agreement.

"Lockbox Control Agreement" as defined in the Security Agreement.

"Lockbox System" as defined in Section 5.9(a)(i).

"LTV" means, with respect to any Receivable, the ratio, at the time of origination, of (i) the unpaid principal balance of such Receivable to (ii) the wholesale book value of the related Financed Vehicle as set forth in the Kelly Blue Book® or the NADA Official Used Car Guide®.

"Margin Stock" as defined in Regulation U of the Board of Governors of the Federal Reserve System as in effect from time to time.

"Material Adverse Effect" means, a material adverse effect on (a) the business operations, assets, condition (financial or otherwise), liabilities or prospects of a Credit Party, (b) the ability of a Credit Party to fully and timely perform its obligations under the Credit Documents (including, without limitation, the Obligations of the Borrower); (c) the legality, validity, binding effect, or enforceability against a Credit Party of any Credit Document to which it is a party; or (d) the rights, remedies and benefits available to, or conferred upon, any Agent, the Lender or any Secured Party under any Credit Document.

"Material Contract" means any contract or other arrangement to which a Credit Party is a party (other than the Credit Documents) for which breach, nonperformance, cancellation or failure to renew could reasonably be expected to have a Material Adverse Effect.

"Maturity Date" means the second (2nd) anniversary of the Closing Date.

"**Maximum Advance**" means, as of any date of determination, the lesser of (A) \$50,000,000 and (B) the product of (x) the Average Maximum Advance Rate and (y)(i) the aggregate unpaid principal balance of the Eligible Receivables minus (ii) any Excess Concentration Amounts.

"Maximum Advance Rate" means,

- (i) with respect to each Receivable that is an Eligible Receivable and is not a Delinquent Receivable or a Defaulted Receivable, the lesser of (A) 75% and (B) the product of (x) one (1) minus the weighted average Purchase Discount and (y) 88%;
 - (ii) with respect to each Eligible Receivable that is a Delinquent Receivable, 35%; and
 - (iii) with respect to each Defaulted Receivable, 0%.

"Monthly Servicing Report" means that Monthly Servicing Report in the form attached as Exhibit A to the Servicing Agreement.

"Moody's" means Moody's Investor Services, Inc., and any successor thereto.

"**Net Insurance Proceeds**" means an amount equal to: (a) any Cash payments or proceeds received by the Borrower under any casualty, business interruption or "**key man**" insurance policies in respect of any covered loss thereunder, <u>minus</u> (b) any actual and reasonable costs incurred by the Borrower in connection with the adjustment or settlement of any claims of the Borrower in respect thereof.

"Net Liquidation Proceeds" as defined in the Servicing Agreement.

"**Obligations**" means all obligations of every nature of the Borrower from time to time owed to the Agents (including former Agents), the Lender (including in its capacity as a counterparty to an Interest Rate Agreement) or any of them, under any Credit Document (including, without limitation, with respect to an Interest Rate Agreement, obligations owed thereunder to any person who was a Lender or an Affiliate of a Lender at the time such Interest

Rate Agreement was entered into), whether for principal, interest (including interest which, but for the filing of a petition in bankruptcy with respect to the Borrower, would have accrued on any Obligation, whether or not a claim is allowed against the Borrower for such interest in the related bankruptcy proceeding), payments for early termination of Interest Rate Agreements, fees, expenses, indemnification or otherwise.

"**Obligor**" means, with respect to a Receivable, the purchaser or co-purchasers of the related Financed Vehicle or any other Person who owes or may be liable for payments under such Receivable.

"Offering Materials" as defined in Section 9.22.

"Organizational Documents" means (a) with respect to any corporation, its certificate or articles of incorporation or organization, as amended, and its by-laws, as amended, (b) with respect to any limited partnership, its certificate of limited partnership, as amended, and its partnership agreement, as amended, (c) with respect to any general partnership, its partnership agreement, as amended, and (d) with respect to any limited liability company, its articles of organization, as amended, and its operating agreement, as amended. In the event any term or condition of this Agreement or any other Credit Document requires any Organizational Document to be certified by a secretary of state or similar governmental official, the reference to any such "Organizational Document" shall only be to a document of a type customarily certified by such governmental official.

"Originator" means CPS.

"**Permitted Expenses**" means the reasonable costs and expenses incurred by the Administrative Agent (and its agents or professional advisors) in connection with the preparation, administration, amendment and due diligence of this Agreement and the other Credit Documents and, which costs and expenses the Borrower shall reimburse to the Administrative Agent or shall pay or cause to be paid. "Permitted Expenses" shall include, without limitation, the expenses set forth in Sections 5.4, 5.5, 5.14 and 9.2 hereof.

"Permitted Investments" means the following, subject to qualifications hereinafter set forth:

- (i) obligations of, or obligations guaranteed as to principal and interest by, the U.S. government or any agency or instrumentality thereof, when such obligations are backed by the full faith and credit of the United States of America;
- (ii) federal funds, unsecured certificates of deposit, time deposits, banker's acceptances, and repurchase agreements having maturities of not more than 365 days of any bank, the short-term debt obligations of which are rated A-1+ (or the equivalent) by each of the Rating Agencies and, if it has a term in excess of three (3) months, the long-term debt obligations of which are rated AAA (or the equivalent) by each of the Moody's and S&P;
 - (iii) deposits that are fully insured by the Federal Deposit Insurance Corp. (FDIC);

- (iv) investments in money market funds (including those owned or managed by the Controlled Account Bank) rated in the highest investment category by each of the Moody's and S&P; and
 - (v) such other investments as to which the Administrative Agent consents.

Notwithstanding the foregoing, "Permitted Investments" (i) shall exclude any security with the S&P's "r" symbol (or any other Rating Agency's corresponding symbol) attached to the rating (indicating high volatility or dramatic fluctuations in their expected returns because of market risk), as well as any mortgage-backed securities and any security of the type commonly known as "strips"; (ii) shall not have maturities in excess of one (1) year; (iii) shall be limited to those instruments that have a predetermined fixed dollar of principal due at maturity that cannot vary or change; and (iv) shall exclude any investment where the right to receive principal and interest derived from the underlying investment provides a yield to maturity in excess of 120% of the yield to maturity at par of such underlying investment. Interest may either be fixed or variable, and any variable interest must be tied to a single interest rate index plus a single fixed spread (if any), and move proportionately with that index. No investment shall be made which requires a payment above par for an obligation if the obligation may be prepaid at the option of the issuer thereof prior to its maturity. All investments shall mature or be redeemable upon the option of the holder thereof on or prior to the earlier of (x) three (3) months from the date of their purchase or (y) the Business Day preceding the day before the date such amounts are required to be applied hereunder.

"Permitted Liens" means:

- (i) Liens imposed by law for taxes, assessments or other governmental charges payable by the Borrower that are not yet due or are being contested in compliance with Section 5.3;
- (ii) Liens arising in favor of the applicable financial institution under the Lockbox Account Control Agreement or the Controlled Account Control Agreement;
- (iii) Liens arising in favor of the applicable counterparty under any Interest Rate Agreement permitted under this Agreement; and
- (iv) Liens on Financed Vehicles that are junior in right to the Lien of the Borrower, or that are possessory liens (such as for storage or repair), tax liens (such as property taxes or registration fees), or statutory enforcement liens (such as for parking tickets).

"Person" means and includes natural persons, corporations, limited partnerships, general partnerships, limited liability companies, limited liability partnerships, joint stock companies, joint ventures, associations, companies, trusts, banks, trust companies, land trusts, business trusts or other organizations, whether or not legal entities, and Governmental Authorities.

"**Prime Rate**" means the rate of interest quoted in The Wall Street Journal, Money Rates Section as the Prime Rate (currently defined as the base rate on corporate loans posted by at least 75% of the nation's 30 largest banks), as in effect from time to time.

"**Principal Office**" means, for the Administrative Agent, 1345 Avenue of the Americas, 46th Floor, New York, New York 10105 (or such other location in the United States of America as the Administrative Agent may from time to time designate in writing to the Borrower and the Lender).

"Proposed Facility" as defined in Section 9.22.

"Protective Advances" as defined in Section 2.1(c).

"**Purchase Agreement**" means that certain Receivables Purchase Agreement dated as of September 25, 2009, by and among the Originator and the Borrower, as may be further amended, modified or supplemented from time to time in accordance with the terms thereof.

"Purchase Discount" means, with respect to any Eligible Receivable, a fraction, (i) the numerator of which is the difference between (A) the aggregate unpaid principal balance of such Eligible Receivable at the time of acquisition or origination by the Originator and (B) the purchase price paid with respect to such Eligible Receivable by the Originator to the related Eligible Dealer, and (ii) the denominator of which is the aggregate unpaid principal balance of such Eligible Receivable at the time of acquisition or origination by the Originator.

"Rating Agencies" means each of Moody's and S&P.

"**Receivable**" means each non-cancelable, unconditional, fixed-rate, automobile installment sale contract or promissory note secured by a first priority, perfected security interest in a Financed Vehicle that was originated or acquired by the Originator, sold by the Originator to the Borrower.

"Receivable File" as defined in the Custodial Agreement.

"**Recoveries**" means, with respect to a Defaulted Receivable, the monies collected from whatever source during any Collection Period following the Collection Period in which such Receivable became a Defaulted Receivable, net of any amounts required by law to be remitted to the Obligor.

"Register" as defined in Section 2.3(a).

"Regulation D" means Regulation D of the Board of Governors of the Federal Reserve System, as in effect from time to time.

"Related Agreements" means, collectively, the Purchase Agreement, the Servicing Agreement, the Custodial Agreement, the Backup Servicing Agreement and any Interest Rate Agreements.

"**Reporting Date**" means the tenth (10th) calendar day of each month (or if such day is not a Business Day, the immediately succeeding Business Day).

"Repossession Rate" means, with respect to any date of determination, a rate, expressed as a percentage, equal to a fraction (I) the numerator of which is the aggregate number of Financed Vehicles that have been repossessed and not resold as of the last day of the most recently ended Collection Period (excluding, however, any Financed Vehicles that had been repossessed and not resold as of the date of any sale or securitization of all or substantially all of the Receivables) and (II) the denominator of which is the aggregate number of Financed Vehicles as of the last day of the most recently ended Collection Period.

"**Repurchase Event**" means, at any time during which a Borrowing Base Deficiency exists, with respect to any Receivable, the failure of such Receivable to satisfy the Eligibility Criteria at the time of its pledge under the Security Agreement.

"Repurchase Price" means, with respect to any Receivable and any date of determination, the principal balance of such Receivable, as of the close of business on the last day of the immediately preceding Collection Period plus all accrued and unpaid interest thereon through the date on which such Receivable is repurchased.

"Required Spread Account Amount" means, as of any date of determination, and subject to Section 5.10 hereof, (i) if no Spread Account Increase Event has occurred and is continuing, the Required Spread Account Floor and (ii) if a Spread Account Increase Event has occurred and is continuing, an amount equal to the product of (x) 10% multiplied by (y) the Total Utilization of Revolving Commitments as of such date (after giving effect to any borrowings hereunder on such date).

"Required Spread Account Deposit" as defined in Section 5.10(b).

"Required Spread Account Floor" means, as of any date of determination, an amount equal to the lesser of (x) the product of (i) 0.50% <u>multiplied</u> <u>by</u> (ii) the Revolving Commitments in effect as of such date, and (y) the then outstanding principal balance of the Revolving Loan (after giving effect to any borrowings hereunder on such date).

"Revolving Availability" means, as of any date of determination, the difference of (i) the lesser of (a) the Revolving Commitments and (b) the Borrowing Base, minus (ii) the Total Utilization of Revolving Commitments.

"Revolving Commitment" means the commitment of the Lender to make or otherwise fund any Revolving Loan. The aggregate amount of the Revolving Commitments is \$50,000,000.

"Revolving Commitment Period" means the period from the Closing Date to but excluding the Revolving Commitment Termination Date.

"**Revolving Commitment Termination Date**" means the earliest to occur of (a) the Maturity Date, (b) the date on which a Funding Termination Event occurs, and (c) the date of the termination of the Revolving Commitments pursuant to Section 7.1.

"Revolving Loan" as defined in Section 2.1.

"**Revolving Loan Note**" means a promissory note substantially in the form of <u>Exhibit B</u>, as it may be amended, supplemented or otherwise modified from time to time in accordance with the terms thereof.

"Right of First Refusal" as defined in Section 9.22.

"S&P" means Standard & Poor's Ratings Group, a division of The McGraw Hill Corporation, and any successor thereto.

"Scheduled Receivable Payment" means, for any Collection Period and for any Receivable, the amount indicated in such Receivable as required to be paid by the Obligor in such Collection Period. If after the Closing Date the Obligor's obligation under such Receivable with respect to a Collection Period has been modified so as to differ from the amount specified in such Receivable as a result of (i) the order of a court in an insolvency proceeding involving the Obligor, (ii) pursuant to the Servicemembers Civil Relief Act or (iii) modifications or extensions of the Receivable permitted by the Credit Documents and the Servicing Agreement, the Scheduled Receivable Payment with respect to such Collection Period shall refer to the Obligor's payment obligation with respect to such Collection Period as so modified.

"Secured Party" as defined in the Security Agreement.

"Securities" means any stock, shares, partnership interests, limited liability company interests, voting trust certificates, certificates of interest or participation in any profit-sharing agreement or arrangement, options, warrants, bonds, debentures, notes, or other evidences of indebtedness, secured or unsecured, convertible, subordinated or otherwise, or in general any instruments commonly known as "securities" or any certificates of interest, shares or participations in temporary or interim certificates for the purchase or acquisition of, or any right to subscribe to, purchase or acquire, any of the foregoing.

"Securities Act" means the Securities Act of 1933, as amended to the date hereof and from time to time hereafter, and any successor statute.

"Security Agreement" means the Security Agreement, dated as of the date hereof, by and between the Borrower and the Collateral Agent on behalf of the Secured Parties, as it may be amended, supplemented or otherwise modified from time to time in accordance with the terms thereof.

"Servicer" means, initially CPS, subject to removal pursuant to the terms of the Servicing Agreement, and thereafter shall mean the Backup Servicer, or any successor servicer appointed pursuant to the Servicing Agreement.

"Servicer Default" as defined in the Servicing Agreement.

"Servicing Agreement" means that certain Servicing Agreement dated as of September 25, 2009, by and among the Borrower, the Servicer and the Administrative Agent, as may be

further amended, modified or supplemented from time to time in accordance with the terms thereof.

"Servicing Fee" as defined in the Servicing Agreement.

"Servicing Fee Rate" as defined in the Servicing Agreement.

"**Settlement Date**" means (a) the fifteenth (15th) calendar day of each month (or if such day is not a Business Day, the immediately succeeding Business Day) beginning in the month of October 2009 and (b) the Maturity Date.

"Shortfall Percentage" means, (i) with respect to any Settlement Date beginning on the sixth (6th) Settlement Date following the Closing Date and prior to the twelfth (12th) Settlement Date following the Closing Date, an amount equal to the excess, if any, of (x) 50% minus (y) the percentage equal to (A) the average Total Utilization of Revolving Commitments during the immediately preceding Collection Period; and (ii) with respect to any Settlement Date beginning on the twelfth (12th) Settlement Date following the Closing Date, an amount equal to the excess, if any, of (x) 75% minus (y) the percentage equal to (A) the average Total Utilization of Revolving Commitments during the immediately preceding Collection Period divided by (B) the Revolving Commitments during the immediately preceding Collection Period.

"Solvency Certificate" means a Solvency Certificate of the chief financial officer of CPS or the Borrower, as the case may be, substantially in the form of Exhibit E-2.

"Solvent" means, with respect the Borrower or CPS, that as of the date of determination, both (a) (i) the sum of such entity's debt (including contingent liabilities) does not exceed the present fair saleable value of such entity's present assets; (ii) such entity's capital is not unreasonably small in relation to its business as contemplated on the Closing Date; and (iii) such entity has not incurred and does not intend to incur, or believe (nor should it reasonably believe) that it will incur, debts beyond its ability to pay such debts as they become due (whether at maturity or otherwise); and (b) such entity is "solvent" within the meaning given that term and similar terms under applicable laws relating to fraudulent transfers and conveyances. For purposes of this definition, the amount of any contingent liability at any time shall be computed as the amount that, in light of all of the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability (irrespective of whether such contingent liabilities meet the criteria for accrual under Statement of Financial Accounting Standard No. 5).

"**Spread Account**" as defined in the Security Agreement.

"**Spread Account Increase Event**" means an uncured breach of a Tier 1 Performance Trigger; <u>provided</u>, that once a Tier 1 Performance Trigger has been breached, such Tier 1 Performance Trigger shall be deemed to be "uncured" until such time as such Tier 1 Performance Trigger has been cured for three (3) consecutive months.

"Subsidiary" means, with respect to any Person, any corporation, partnership, limited liability company, association, joint venture or other business entity of which more than 50% of

the total voting power of shares of stock or other ownership interests entitled (without regard to the occurrence of any contingency) to vote in the election of the Person or Persons (whether directors, managers, trustees or other Persons performing similar functions) having the power to direct or cause the direction of the management and policies thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof; <u>provided</u>, in determining the percentage of ownership interests of any Person controlled by another Person, no ownership interest in the nature of a "qualifying share" of the former Person shall be deemed to be outstanding.

"Tax" means any present or future tax, levy, impost, duty, assessment, charge, fee or deduction or withholding in respect thereof of any similar nature and whatever called, imposed by a Governmental Authority, on whomsoever and wherever imposed, levied, collected, withheld or assessed; provided, "Tax on the overall net income" of a Person shall be construed as a reference to a Tax imposed by the jurisdiction in which that Person is organized or in which that Person's applicable principal office (and/or, in the case of the Lender, its lending office) is located or in which that Person (and/or, in the case of the Lender, its lending office) is deemed to be doing business on all or part of the net income, profits or gains (whether worldwide, or only insofar as such income, profits or gains are considered to arise in or to relate to a particular jurisdiction, or otherwise) of that Person (and/or, in the case of the Lender, its applicable lending office) and shall include any backup or other withholding tax that shall be eligible to be credited against any such Tax.

"Tier 1 Performance Trigger" means the breach of any of the collateral performance tests set forth on Appendix D hereto.

"Tier 2 Performance Trigger" means the breach of any of the collateral performance tests set forth on Appendix E hereto.

"**Total Utilization of Revolving Commitments**" means, as of any date of determination, an amount equal to the aggregate principal amount of all outstanding Revolving Loans.

"**Transaction Costs**" means the fees, costs and expenses payable by CPS or the Borrower on or before the Closing Date in connection with the transactions contemplated by the Credit Documents and the Related Agreements, to the extent approved in writing by the Administrative Agent.

"UCC" means the Uniform Commercial Code (or any similar or equivalent legislation) as in effect in any applicable jurisdiction.

"**Underwriting Policies**" means the credit policies and practices and underwriting guidelines of the Originator in effect as of the date hereof and attached hereto as <u>Exhibit F</u>, as such guidelines may be amended from time to time, (x) in the case of any material change, with the prior written consent of the Administrative Agent (such consent to not be unreasonably withheld or delayed) and (y) in the case of any other change, with written notice to the Administrative Agent within ten (10) Business Days of the implementation of any such change.

"Vintage Pool" means, as of any date of determination occurring after the last day of the last calendar quarter of 2009, the pool of all automobile receivables originated, or acquired from

Eligible Dealers by the Originator during any completed calendar quarter. The first such calendar quarter to be measured will be the quarter ending December 31, 2009.

"Warrants" means (i) that certain Warrant, dated as of September 25, 2009, issued by CPS to Drawbridge Special Opportunities Fund LP and (ii) any additional Warrants issued by CPS in connection with an assignment in whole or in part of a warrant by Fortress or a designated Affiliate of Fortress, in each case as the same may be further amended, modified or supplemented from time to time in accordance with the terms thereof.

1.2. Accounting Terms.

- (a) <u>Generally</u>. All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing CPS' audited financial statements, <u>except</u> as otherwise specifically prescribed herein.
- (b) <u>Changes in GAAP</u>. If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Credit Document, and either the Borrower, CPS or the Lender shall so request, the Administrative Agent, the Lender, CPS and the Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP; <u>provided that</u>, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) CPS and the Borrower shall provide to the Administrative Agent and the Lender financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.
- 1.3. Interpretation, etc. Any of the terms defined herein may, unless the context otherwise requires, be used in the singular or the plural, depending on the reference. References herein to any Section, Appendix, Schedule or Exhibit shall be to a Section, an Appendix, a Schedule or an Exhibit, as the case may be, hereof unless otherwise specifically provided. The use herein of the word "include" or "including," when following any general statement, term or matter, shall not be construed to limit such statement, term or matter to the specific items or matters set forth immediately following such word or to similar items or matters, whether or not no limiting language (such as "without limitation" or "but not limited to" or words of similar import) is used with reference thereto, but rather shall be deemed to refer to all other items or matters that fall within the broadest possible scope of such general statement, term or matter. The words "hereof", "herein", "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement. Unless the context requires otherwise or otherwise specified in any applicable Credit Document, (a) reference to any Person include that Person's successors and assignees, (b) any definition of or reference to any Credit Document, agreement, instrument or other document herein shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified (subject to any restrictions on such

amendments, supplements, or modifications set forth herein or therein), and (c) any reference to any law or regulation herein shall refer to such law or regulation as amended, modified or supplemented from time to time.

SECTION REVOLVING LOANS

2.1. Revolving Loans.

(a) Revolving Commitments. During the Revolving Commitment Period, subject to the terms and conditions hereof, the Lender agrees from time to time to make loans to the Borrower (each a "Revolving Loan" and collectively, the "Revolving Loans") in an aggregate amount up to but not exceeding the Revolving Commitment; provided, that after giving effect to the making of any Revolving Loans in no event shall the Total Utilization of Revolving Commitments exceed the lesser of (i) the Revolving Commitments then in effect and (ii) the Borrowing Base. Amounts borrowed pursuant to this Section 2.1(a) may be repaid and reborrowed during the Revolving Commitment Period. The Revolving Commitment shall expire on the Revolving Commitment Termination Date and all Revolving Loans and all other amounts owed hereunder with respect to the Revolving Loans and the Revolving Commitments shall be paid in full no later than such date.

(b) Borrowing Mechanics for Revolving Loans.

- (i) Revolving Loans shall be in an aggregate minimum amount of \$250,000.
- (ii) Whenever the Borrower desires that the Lender make Revolving Loans, the Borrower shall deliver to the Administrative Agent a fully executed and delivered Funding Notice together with a Borrowing Base Certificate no later than 1:00 p.m. (New York City time) at least two (2) Business Days in advance of the proposed Credit Date. Each such Funding Notice shall be delivered reflecting sufficient Revolving Availability for the requested Revolving Loans.
- (iii) Notice of receipt of each Funding Notice and Borrowing Base Certificate in respect of Revolving Loans, together with the applicable Interest Rate, shall be provided by the Administrative Agent to the Lender by telefacsimile with reasonable promptness, but not later than 2:00 p.m. (New York City time) on the same day as the Administrative Agent's receipt of such Funding Notice from the Borrower (provided the Administrative Agent shall have received such notice from the Borrower by 1:00 p.m. (New York City time)).
- (iv) The Lender shall make the amount of its Revolving Loan available to the Administrative Agent not later than 12:00 p.m. (New York City time) on the applicable Credit Date by wire transfer of same day funds in Dollars, at the Administrative Agent's Principal Office. Except as provided herein, upon satisfaction or waiver by the Administrative Agent of the conditions precedent specified herein, the Administrative Agent shall make the proceeds of such Revolving Loans available to the Borrower on the applicable Credit Date by causing an amount of same day funds in Dollars equal to the proceeds of all such Revolving Loans received by the Administrative

Agent from the Lender to be credited to the account of the Borrower at the Administrative Agent's Principal Office or such other account as may be designated in writing to the Administrative Agent by the Borrower.

- **(v)** Unless otherwise permitted by the Administrative Agent in its sole and absolute discretion and subject to an administration fee of \$2,500 per Revolving Loan, no more than two (2) Revolving Loans shall be made per calendar week.
- (c) Protective Advances. Subject to the limitations set forth below and in the proviso to the first sentence of Section 2.1(a), and whether or not an Event of Default or a Default shall have occurred and be continuing, the Administrative Agent is authorized by the Borrower and the Lender, from time to time in the Administrative Agent, in its sole good faith discretion, to make Revolving Loans to the Borrower on behalf of the Lender, which the Administrative Agent, in its sole discretion, deems necessary (i) to preserve or protect the Collateral, or any portion thereof, (ii) to enhance the likelihood of, or maximize the amount of, repayment of the Revolving Loans and other Obligations, or (iii) to pay any other amount chargeable to or required to be paid by the Borrower pursuant to the terms of this Agreement and the other Credit Documents, including, without limitation, payments of principal, interest, fees and reimbursable expenses, in the event that any amounts are still owing after application of the amounts in the Spread Account pursuant to Sections 2.10 or 2.12 (any of such Revolving Loans are herein referred to as "Protective Advances"). Notwithstanding anything to the contrary set forth herein, in no event shall the aggregate amount of Protective Advances made by the Administrative Agent pursuant to this Section 2.1(c) exceed \$5,000,000. Protective Advances may be made even if the conditions precedent set forth in Section 3 have not been satisfied. The Protective Advances shall be secured by the Collateral and shall constitute Obligations. The Borrower shall pay the unpaid principal amount and all unpaid and accrued interest of each Protective Advance on the earlier of the Revolving Commitment Termination Date and within two (2) Business Days following demand for payment by the Administrative Agent. All Protective Advances shall be Base Rate Loans.
- **2.2.** <u>Use of Proceeds</u>. The proceeds of the Revolving Loans, if any, made on the Closing Date shall be applied by the Borrower to finance the acquisition of Eligible Receivables from the Originator and to pay Transaction Costs. The proceeds of the Revolving Loans made after the Closing Date shall be applied by the Borrower to finance the acquisition of Receivables from the Originator pursuant to the Purchase Agreement, to fund any Required Spread Account Amount in connection with the funding of a Revolving Loan, to pay distributions on its Capital Stock to CPS and to pay ongoing operating expenses of the Borrower. No portion of the proceeds of any Credit Extension shall be used in any manner that causes such Credit Extension or the application of such proceeds to violate Regulation T, Regulation U or Regulation X of the Board of Governors of the Federal Reserve System or any other regulation thereof or to violate the Exchange Act.

2.3. Register; Notes.

(a) <u>Register</u>. The Administrative Agent shall maintain at its Principal Office a register for the recordation of the names and addresses of the Lender and the Revolving Commitments and Revolving Loans from time to time (the "**Register**"). The Register shall be

available for inspection by the Borrower or the Lender at any reasonable time and from time to time upon reasonable prior notice to the Administrative Agent. The Administrative Agent shall record in the Register the Revolving Commitments and the Revolving Loans, and each repayment or prepayment in respect of the principal amount of the Revolving Loans, and any such recordation shall be conclusive and binding on the Borrower and the Lender, absent manifest error; provided, failure to make any such recordation, or any error in such recordation, shall not affect the Revolving Commitments or the Borrower's Obligations in respect of any Revolving Loan. The Borrower hereby designates the entity serving as Administrative Agent to serve as the Borrower's agent solely for purposes of maintaining the Register as provided in this Section 2.3, and the Borrower hereby agrees that, to the extent such entity serves in such capacity, the entity serving as Administrative Agent and its officers, directors, employees, agents and affiliates shall constitute "Indemnitees."

(b) Revolving Loan Notes. If so requested by the Lender prior to the Closing Date, or upon two (2) Business Days prior written notice at any time after the Closing Date, the Borrower shall execute and deliver to the Lender (and/or, if applicable and if so specified in such notice, to any Person who is an assignee of the Lender pursuant to Section 9.6) on the Closing Date (or, if such notice is delivered after the Closing Date, promptly after the Borrower's receipt of such notice) a Revolving Loan Note or Revolving Loan Notes, as so requested, to evidence the Revolving Loan.

2.4. Interest on Revolving Loans.

- (a) Except as otherwise set forth herein, each Revolving Loan shall bear interest on the unpaid principal amount thereof from the date made through repayment (whether by acceleration or otherwise) as follows: (i) if a Base Rate Loan, at the Base Rate plus the Applicable Margin; or (ii) if a LIBOR Rate Loan, at the Adjusted LIBOR Rate plus the Applicable Margin.
- (b) Interest payable pursuant to Section 2.4(a) shall be computed on the basis of a 360-day year, in each case for the actual number of days elapsed in the period during which it accrues. In computing interest on any Revolving Loan, the date of the making of such Revolving Loan or the first day of an Interest Period applicable to such Revolving Loan shall be included, and the date of payment of such Revolving Loan or the expiration date of an Interest Period applicable to such Revolving Loan shall be excluded; <u>provided</u>, if a Revolving Loan is repaid on the same day on which it is made, one (1) day's interest shall be paid on that Revolving Loan.
- (c) Except as otherwise set forth herein, interest on each Revolving Loan shall be payable in arrears on (i) each Settlement Date applicable to that Revolving Loan; (ii) with respect to any prepayment in whole or in part of such Revolving Loan, whether voluntary or mandatory, the Settlement Date immediately following such prepayment in an amount equal to the interest accrued and unpaid on the amount so prepaid to the date of prepayment; and (iii) at maturity.
- **2.5.** Continuation. Subject to Section 2.14 and so long as no Default or Event of Default shall have occurred and be continuing, upon the expiration of any Interest Period

applicable to any LIBOR Rate Loan, such LIBOR Rate Loan shall automatically continue for an additional Interest Period at the Adjusted LIBOR Rate calculated as of the most recent Interest Rate Reset Date.

- **2.6. Default Interest**. Upon the occurrence and during the continuance of an Event of Default, the principal amount of all Revolving Loans outstanding and, to the extent permitted by applicable law, any interest payments on the Revolving Loans or any fees or other amounts owed hereunder, shall thereafter bear interest (including post-petition interest in any proceeding under the Bankruptcy Code or other applicable Debtor Relief Laws) payable in accordance with the provisions of Section 2.10 or 2.12, as the case may be, at a rate that is 3% per annum in excess of the Interest Rate otherwise payable hereunder with respect to the applicable Revolving Loans until no Event of Default is then continuing. Payment or acceptance of the increased rates of interest provided for in this Section 2.6 is not a permitted alternative to timely payment and shall not constitute a waiver of any Event of Default or otherwise prejudice or limit any rights or remedies of the Administrative Agent or the Lender.
 - **2.7. Fees.** The Borrower and CPS agree, jointly and severally, to pay each of the fees referred to in the Fee Letter.

2.8. Call Protection.

- (a) The Facility shall not be terminated prior to the Maturity Date.
- (b) Notwithstanding the foregoing, in the event the Facility is terminated by the Borrower or CPS prior to the Maturity Date, the Borrower shall pay to the Lender, on the date of such termination, a fee equal to the product of (x) the applicable Interest Rate at the time of such termination, (y) the Revolving Commitments and (z) the remaining term of the Facility (expressed as an annualized number).

2.9. Mandatory Prepayments.

- (a) <u>Borrowing Base Deficiency</u>. The Borrower shall prepay the Revolving Loans within one (1) Business Day of the earlier of (i) an Authorized Officer of the Borrower becoming aware that a Borrowing Base Deficiency exists and (ii) receipt by the Borrower of notice from the Administrative Agent that a Borrowing Base Deficiency exists, in each case in an amount equal to such Borrowing Base Deficiency, which shall be applied to repay the Revolving Loans.
- (b) <u>Repurchase Events</u>. Upon the occurrence of a Repurchase Event: (i) the Borrower shall cause the Originator to repurchase each affected Receivable at a price equal to the Repurchase Price and (ii) the Borrower shall prepay the Revolving Loans in an amount equal to the Repurchase Price with respect to the related Receivable(s).
- (c) <u>Prepayment Certificate</u>. Concurrently with any prepayment of the Revolving Loans pursuant to Sections 2.9(a) and 2.9(b), the Borrower shall deliver, or cause to be delivered, to the Administrative Agent a certificate of an Authorized Officer demonstrating the calculation of the amount of the applicable net proceeds.

2.10. Payments during Event of Default or Following a Funding Termination Event. Upon the occurrence and during the continuance of an	
Event of Default or a Funding Termination Event on each Settlement Date, (a) all payments made hereunder and under the other Credit Documents (including	ng
in respect of proceeds from any Collateral) and (b) all amounts on deposit in the Controlled Accounts and any income and gains from investment of funds in	1
any Controlled Account that accrued during the immediately preceding Collection Period shall be applied by the Controlled Account Bank at the written	
direction of the Collateral Agent as follows:	

(i) First, to the payment of, and in the same priority as, items (i) through (vi) and (viii) through (ix) in Section 2.12

below;

- (ii) Second, to the Lender, to reduce the outstanding principal balance on the Revolving Loans to zero; and
- (iii) Third, to the Borrower, any remaining amounts.

2.11. Controlled Accounts and Amounts.

- (a) On or prior to the date hereof, the Borrower shall cause to be established and maintained, (i) securities accounts at the Controlled Account Bank in the name of the Collateral Agent designated as the Collection Account and the Spread Account as to which the Collateral Agent for the benefit of the Lender has control over such accounts within the meaning of Section 9-106 of the UCC pursuant to the Controlled Account Control Agreement (each, a "Controlled Account" and collectively, the "Controlled Accounts"), and (ii) a deposit account at the Lockbox Account Bank in the name of the Borrower designated as the Lockbox Account as to which the Collateral Agent has control over such account for the benefit of the Lender within the meaning of Section 9-104(a)(2) of the UCC pursuant to the Lockbox Account Control Agreement.
- (b) So long as no Event of Default has occurred and shall be continuing, the Borrower or the Servicer shall be permitted to direct the investment of the funds from time to time held in the Controlled Accounts in Permitted Investments and to sell or liquidate such Permitted Investments and reinvest proceeds from such sale or liquidation in other Permitted Investments (but none of the Collateral Agent, the Administrative Agent, or the Lender shall have liability whatsoever in respect of any failure by the Controlled Account Bank to do so), with all such proceeds and reinvestments to be held in the applicable Controlled Account; provided, however, that the maturity of the Permitted Investments on deposit in the Controlled Accounts shall be no later than the Business Day immediately preceding the date on which such funds are required to be withdrawn therefrom pursuant to this Agreement; and, provided further, that the Borrower shall remit into the applicable Controlled Account an amount equal to any losses realized on Permitted Investments contained therein. No Permitted Investment shall be liquidated at a loss at the direction of the Borrower except to the extent necessary to make a required payment as described herein. All income and gains from the investment of funds in the Controlled Accounts shall be retained in the respective Controlled Account from which they were derived, until the next Settlement Date, at which time such income and gains shall be applied in accordance with Section 2.10 or Section 2.12, as the case may be. As between the

Borrower and the Collateral Agent, the Borrower shall treat all income, gains and losses from the investment of amounts in the Controlled Accounts as its income or loss for federal, state and local income tax purposes.

(c) The Control Agreements will provide that all funds in the Lockbox will be swept daily into the Collection Account.

2.12. Application of Collections.

To the extent no Event of Default or Funding Termination Event has occurred and is continuing, the Collateral Agent will instruct the Controlled Account Bank (or will instruct the Servicer to instruct the Controlled Account Bank pursuant to the Servicing Agreement), on each Settlement Date, to transfer collected funds held by the Controlled Account Bank in the Collection Account, in the following amounts and priority in accordance with the Monthly Servicing Report:

- (i) to any Eligible Hedge Counterparty, amounts due under any outstanding Interest Rate Agreement, except amounts due as fees, expenses or as a result of the occurrence of an event of default or termination event under such Interest Rate Agreement or otherwise due upon termination of such Interest Rate Agreement, which amounts shall be paid as provided in (vi) below;
 - (ii) to the Servicer, any unpaid Servicing Fees;
- (iii) on a *pari passu* basis, (1) to the Custodian, the Custodian Fees and Expenses accrued and unpaid as of the last day of the preceding month, (2) to the Backup Servicer, the Backup Servicing Fees and reimbursable expenses (including, without limitation, any transition costs) of the Backup Servicer accrued and unpaid as of the last day of the preceding month, and (3) to the Controlled Account Bank, the Controlled Account Bank Fees accrued and unpaid as of the last day of the preceding month;
 - (iv) to the Administrative Agent, to pay any costs or fees due to the Administrative Agent and the Collateral Agent,
- (v) to the Lender, any accrued but unpaid interest, fees and expenses in connection with this Agreement and any other Credit Document;
- (vi) to any Eligible Hedge Counterparty, any unpaid fees, expenses or amounts owed as a consequence of an event of default or termination event under an outstanding Interest Rate Agreement or otherwise due upon termination of such Interest Rate Agreement;
 - (vii) to the Lender, any amounts necessary to reduce the Borrowing Base Deficiency, if any, to zero;
- (viii) to the Administrative Agent, to be deposited in the Spread Account to fund the excess of the Required Spread Account Amount over the amounts then on deposit in the Spread Account;

(ix) on a *pari passu* basis, (1) to the Custodian, any other amounts payable to the Custodian in its capacity as Custodian pursuant to this Agreement or the Custodial Agreement to the extent unpaid by the Borrower and not covered under item (iii) above, (2) to the Backup Servicer, any other amounts payable to the Backup Servicer pursuant to this Agreement or the Backup Servicing Agreement to the extent unpaid by the Borrower and not covered under item (iii) above, and (3) to the Controlled Account Bank, any other amounts payable to the Controlled Account Bank pursuant to this Agreement or the Control Agreements to the extent unpaid by the Borrower hereunder and not covered under item (iii) above; and

(x) prior to the Maturity Date, and <u>provided</u> that no Borrowing Base Deficiency would occur after giving effect to such distribution, to the Borrower for its own account.

2.13. General Provisions Regarding Payments.

- (a) All payments by the Borrower of principal, interest, fees and other Obligations shall be made in Dollars in immediately available funds, without defense, recoupment, setoff or counterclaim, free of any restriction or condition, and delivered to the Administrative Agent, for the account of the Lender, not later than 3:00 p.m. (New York City time) on the date due at 1345 Avenue of the Americas, 46th Floor, New York, New York 10105 or via wire transfer of immediately available funds to account number 2047628893919 maintained by the Administrative Agent with Bank of America (ABA No. 026 009 593) in New York City (or at such other location or bank account within the City and State of New York as may be designated by the Administrative Agent from time to time); funds received by the Administrative Agent after that time on such due date shall be deemed to have been paid by the Borrower on the next Business Day (except to the extent such delay in payment results solely from the Controlled Account Bank's failure to distribute funds on deposit in the Collection Account and available for distribution as of 3:00 p.m. on such Business Day in accordance with Section 2.10 or 2.12).
- (b) All payments in respect of the principal amount of any Revolving Loan (other than voluntary or mandatory prepayments of any Revolving Loan as provided in Section 2.4(c)) shall be accompanied by payment of accrued interest on the principal amount being repaid or prepaid.
- (c) The Administrative Agent shall promptly distribute to the Lender at such address or via wire transfer as the Lender shall indicate in writing, all payments and prepayments of principal and interest due hereunder, together with all other amounts due with respect thereto, including, without limitation, all fees payable with respect thereto, to the extent received by the Administrative Agent.
- (d) Notwithstanding the foregoing provisions hereof, if the Lender makes Base Rate Loans in lieu of LIBOR Rate Loans, the Administrative Agent shall give effect thereto in apportioning payments received thereafter.

- (e) Subject to the proviso set forth in the definition of "**Interest Period**," whenever any payment to be made hereunder shall be stated to be due on a day that is not a Business Day, such payment shall be made on the next succeeding Business Day and such extension of time shall be included in the computation of the payment of interest hereunder.
- (f) The Borrower hereby authorizes the Administrative Agent to charge the Borrower's accounts with the Administrative Agent or any of its Affiliates in order to cause timely payment to be made to the Administrative Agent of all principal, interest, fees and expenses due hereunder (subject to sufficient funds being available in its accounts for that purpose).
- (g) The Administrative Agent shall give prompt telephonic notice to the Borrower and the Lender (confirmed in writing) if any payment is not made in conformity with this Section 2.13. Interest shall continue to accrue on any principal as to which a non-conforming payment is made until such funds become available funds (but in no event less than the period from the date of such payment to the next succeeding applicable Business Day) at the Default Rate determined pursuant to Section 2.6 (if applicable) from the date such amount was due and payable until the date such amount is paid in full.

2.14. Making or Maintaining LIBOR Rate Loans.

(a) Inability to Determine Applicable Interest Rate. In the event that the Administrative Agent shall have reasonably determined in good faith (which determination shall be final and conclusive and binding upon all parties hereto), on any Interest Rate Reset Date with respect to any LIBOR Rate Loans, that by reason of circumstances affecting the London interbank market adequate and fair means do not exist for ascertaining the interest rate applicable to such LIBOR Rate Loans on the basis provided for in the definition of Adjusted LIBOR Rate ("LIBOR Unavailability"), the Administrative Agent shall on such date give notice (by telefacsimile or by telephone confirmed in writing) to the Borrower and the Lender of such determination, whereupon (i) no Revolving Loans may be made as LIBOR Rate Loans until such time as the Administrative Agent notifies the Borrower and the Lender that the circumstances giving rise to such notice no longer exist, (ii) the Borrower shall have the right to rescind any Funding Notice previously given by the Borrower with respect to the Revolving Loans in respect of which such determination was made by giving notice (by telefacsimile or by telephone confirmed in writing) to the Administrative Agent of such rescission on the date on which the Administrative Agent gives notice of its determination as described above (which notice of rescission the Administrative Agent shall promptly transmit to the Lender), (iii) all then-existing Revolving Loans shall convert automatically to Base Rate Loans at the end of the then-applicable Interest Period if such circumstances still exist at such time and (iv) any subsequent borrowings shall be made as Base Rate Loans until such circumstances no longer exist. At such time as the Administrative Agent shall notify the Borrower and the Lender that any period of LIBOR Unavailability has ended, on the first day of the Interest Period next following such determination, all Base Rate Loans carried by the Lender as a consequence of this Section 2.14(a) shall autom

- Illegality or Impracticability of LIBOR Rate Loans. In the event that on any date the Lender shall have reasonably determined in good faith (which determination shall be final and conclusive and binding upon all parties hereto but shall be made only after consultation with the Borrower and the Administrative Agent) that the making or maintaining of its LIBOR Rate Loans (i) has become unlawful after the date hereof as a result of compliance by such Lender in good faith with any law, treaty, governmental rule, regulation, guideline or order (or would conflict with any such treaty, governmental rule, regulation, guideline or order not having the force of law even though the failure to comply therewith would not be unlawful), or (ii) has become impracticable, as a result of contingencies occurring after the date hereof which materially and adversely affect the London interbank market or the position of the Lender in that market, then, and in any such event, the Lender shall on that day give notice (by telefacsimile or by telephone confirmed in writing) to the Borrower and the Administrative Agent of such determination. Thereafter (1) the obligation of the Lender to make Revolving Loans as LIBOR Rate Loans shall be suspended until such notice shall be withdrawn by the Lender at such time as the circumstances giving rise to such notice no longer exist, (2) to the extent such determination by the Lender relates to a LIBOR Rate Loan then being requested by the Borrower pursuant to a Funding Notice, the Lender shall make such Revolving Loan (or continue such Revolving Loan) as a Base Rate Loan, (3) the Lender's obligation to maintain its outstanding LIBOR Rate Loans (the "Affected Loans") shall be terminated at the earlier to occur of the expiration of the Interest Period then in effect with respect to the Affected Loans or when required by law, and (4) the Affected Loans shall automatically convert into Base Rate Loans on the date of such termination. Notwithstanding the foregoing, to the extent a determination by the Lender as described above relates to a LIBOR Rate Loan then being requested by the Borrower pursuant to a Funding Notice, the Borrower shall have the option, notwithstanding anything to the contrary in Section 2.1(b)(ii), to rescind such Funding Notice by giving notice (by telefacsimile or by telephone confirmed in writing) to the Administrative Agent of such rescission on the date on which the Lender gives notice of its determination as described above (which notice of rescission the Administrative Agent shall promptly transmit to the Lender).
- (c) <u>Compensation for Breakage or Non-Commencement of Interest Periods</u>. The Borrower shall compensate the Lender, upon written request by the Lender (which request shall set forth the basis for requesting such amounts), for all reasonable losses, expenses and liabilities (including any interest paid or calculated to be due and payable by the Lender to lenders of funds borrowed by it to make or carry its LIBOR Rate Loans and any loss, expense or liability sustained by the Lender in connection with the liquidation or re-employment of such funds but excluding loss of anticipated profits) which the Lender would sustain: (i) if for any reason (other than a default by the Lender) a borrowing of any LIBOR Rate Loan does not occur on a date specified therefor in a Funding Notice; (ii) if any prepayment or other principal payment of any of its LIBOR Rate Loans occurs on any day other than the last day of an Interest Period applicable to that Revolving Loan (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise); or (iii) if any prepayment of any of its LIBOR Rate Loans is not made on any date specified in a notice of prepayment given by the Borrower.
- (d) <u>Booking of LIBOR Rate Loans</u>. The Lender may make, carry or transfer LIBOR Rate Loans at, to, or for the account of any of its branch offices or the office of an Affiliate of such Lender.

2.15. <u>Increased Costs; Capital Adequacy</u>.

(a) <u>Compensation For Increased Costs and Taxes</u>. Subject to the provisions of Section 2.16 (which shall be controlling with respect to the matters covered thereby), in the event that the Lender shall determine (which determination shall, absent manifest error, be final and conclusive and binding upon all parties hereto) that any law, treaty or governmental rule, regulation or order, or any change therein or in the interpretation, administration or application thereof (including the introduction of any new law, treaty or governmental rule, regulation or order), or any determination of a court or governmental authority, in each case that becomes effective after the date hereof, or compliance by the Lender with any guideline, request or directive issued or made after the date hereof by any central bank or other governmental or quasi-governmental authority (whether or not having the force of law): (i) subjects the Lender (or its applicable lending office) to any additional Tax (other than any Tax on the overall net income of the Lender) with respect to this Agreement or any of the other Credit Documents or any of its obligations hereunder or thereunder or any payments to the Lender (or its applicable lending office) of principal, interest, fees or any other amount payable hereunder; (ii) imposes, modifies or holds applicable any reserve (including any marginal, emergency, supplemental, special or other reserve), special deposit, compulsory loan, FDIC insurance or similar requirement against assets held by, or deposits or other liabilities in or for the account of, or advances or loans by, or other credit extended by, or any other acquisition of funds by, any office of the Lender (other than any such reserve or other requirements); or (iii) imposes any other condition (other than with respect to a Tax matter) on or affecting the Lender (or its applicable lending office) or its obligations hereunder or the London interbank market; and the result of any of the foregoing is to increase the cost to the Lender of agreeing to make, making or maintaining Revolving Loans hereunder or to reduce any amount received or receivable by the Lender (or its applicable lending office) with respect thereto; then, in any such case, the Borrower shall pay to the Lender within ten (10) Business Days of receipt of the statement referred to in the next sentence, such additional amount or amounts (in the form of an increased rate of, or a different method of calculating, interest or otherwise as the Lender in its sole discretion shall determine) as may be necessary to compensate the Lender for any such increased cost or reduction in amounts received or receivable hereunder. The Lender shall deliver to the Borrower (with a copy to the Administrative Agent) a written statement, setting forth in reasonable detail the basis for calculating the additional amounts owed to such Lender under this Section 2.15(a), which statement shall be conclusive and binding upon all parties hereto absent manifest error.

(b) Capital Adequacy Adjustment.

(i) In the event that any Lender shall have determined that the adoption, effectiveness, phase-in or applicability after the Closing Date of any law, rule or regulation (or any provision thereof) regarding capital adequacy, or any change therein or in the interpretation or administration thereof by any Governmental Authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by any Lender (or its applicable lending office) with any guideline, request or directive regarding capital adequacy (whether or not having the force of law) of any such Governmental Authority, central bank or comparable agency, has or would have the effect of reducing the rate of return on the capital of such Lender or any corporation controlling such Lender as a consequence of, or with reference to, such Lender's

Revolving Loans or Revolving Commitments or participations therein or other obligations hereunder with respect to the Revolving Loans to a level below that which such Lender or such controlling corporation could have achieved but for such adoption, effectiveness, phase-in, applicability, change or compliance (taking into consideration the policies of such Lender or such controlling corporation with regard to capital adequacy), then from time to time, within ten (10) Business Days after receipt by the Borrower from such Lender of the statement referred to in the next sentence, the Borrower shall pay to such Lender such additional amount or amounts as will compensate such Lender or such controlling corporation on an after-tax basis for such reduction. Such Lender shall deliver to the Borrower (with a copy to the Administrative Agent) a written statement, setting forth in reasonable detail the basis for calculating the additional amounts owed to the Lender under this Section 2.15(b), which statement shall be conclusive and binding upon all parties hereto absent manifest error.

- (ii) If any of the events requiring payments of additional amounts by Borrower under Section 2.15(a) or (b) occurs, Lender shall take such steps as may be reasonable to avoid Borrower being required to pay such additional amounts and shall consult with Borrower in good faith with a view to agreeing to alternative arrangements whereby any such requirement can be avoided or mitigated, including without limitation, fulfilling any of Lender's obligations through another branch or affiliate in the United States, *provided* that no such alternative arrangements will be required that would be disadvantageous to Lender and would result in significantly increased cost to Lender.
- (iii) If Lender demands payment with respect to amounts owed under Section 2.15(a) or (b), Borrower shall have the right, if no Default or Event of Default has occurred and is then continuing, within ninety (90) days after receipt of such demand, to remove Lender (the "Affected Person") and to designate another lender (the "Replacement Person") reasonably acceptable to the Administrative Agent to purchase the Affected Person's outstanding Revolving Loans and to assume the Affected Person's obligations under this Agreement; provided that increased costs incurred by such Lender prior to the date of its replacement shall have been paid as provided herein. The Affected Person agrees to sell to the Replacement Person its outstanding Loans (at par, with accrued interest through the date of purchase, in immediately available funds) and to delegate to the Replacement Person its obligations to Borrower under this Agreement. Upon such sale and delegation by the Affected Person and the purchase and assumption by the Replacement Person, and compliance with the provisions of Section 9.6 hereof, the Affected Person shall cease to be a Lender hereunder and the Replacement Person shall become a Lender under this Agreement. Each Affected Person shall continue to be entitled to receive from Borrower its share of interest, fees, costs and other sums which have not been assigned by the Affected Person to the Replacement Person.

2.16. Taxes; Withholding, etc.

(a) <u>Payments to Be Free and Clear</u>. All sums payable by each Credit Party hereunder and under the other Credit Documents shall (except to the extent required by law) be paid free and clear of, and without any deduction or withholding on account of, any Tax (other than a Tax on the overall net income of the Lender) imposed, levied, collected, withheld or

assessed by or within the United States of America or any political subdivision in or of the United States of America or any other jurisdiction from or to which a payment is made by or on behalf of a Credit Party or by any federation or organization of which the United States of America or any such jurisdiction is a member at the time of payment.

- (b) Withholding of Taxes. If a Credit Party or any other Person is required by law to make any deduction or withholding on account of any such Tax from any sum paid or payable by any Credit Party in respect of interest paid or payable by the Borrower to the Administrative Agent or the Lender under any of the Credit Documents: (i) the Borrower shall notify the Administrative Agent as soon as practicable of any such requirement or any change in any such requirement as soon as the Borrower becomes aware of it; (ii) the Borrower shall pay any such Tax before the date on which penalties attach thereto, such payment to be made (if the liability to pay is imposed on any Credit Party) for its own account or (if that liability is imposed on the Administrative Agent or the Lender, as the case may be) on behalf of and in the name of the Administrative Agent or the Lender; (iii) the sum payable by any Credit Party in respect of which the relevant deduction, withholding or payment is required shall be increased to the extent necessary to ensure that, after the making of that deduction, withholding or payment, the Administrative Agent or the Lender, as the case may be, receives on the due date a net sum equal to what it would have received had no such deduction, withholding or payment been required or made; and (iv) within thirty (30) days (or as soon as practicable thereafter) after paying any sum from which it is required by law to make any deduction or withholding, and within thirty (30) days (or as soon as practicable thereafter) after the due date of payment of any Tax which it is required by clause (ii) above to pay, the Borrower shall use commercially reasonable efforts to deliver to the Administrative Agent evidence satisfactory to the other affected parties of such deduction, withholding or payment and of the remittance thereof to the relevant taxing or other authority; provided, no such additional amount shall be required to be paid to any Lender under clause (iii) above (and in the case of any payment required under clause (ii) above, such payment shall be treated as a payment under the Credit Documents to the Borrower, the Administrative Agent or the Lender as the case may be) except to the extent that any change after the date hereof in any such requirement for a deduction, withholding or payment as is mentioned therein shall result in an increase in the rate of such deduction, withholding or payment from that in effect at the date hereof, in respect of payments to the Lender.
- (c) The Borrower shall not be required to pay any additional amounts to the Lender in respect of United States federal withholding taxes pursuant to Section 2.16(b) above if the obligation to pay such additional amounts would not have arisen but for a failure by Lender to comply with the provisions of Section 2.16(d) hereof. In the event the Borrower has actual knowledge that it is required to, or there arises, in the Borrower's reasonable opinion, a substantial likelihood that the Borrower will be required to, pay an increased amount or otherwise indemnify the Lender for or on account of any Taxes pursuant to Section 2.16(b), the Borrower shall promptly notify the Lender of the nature of such Taxes and shall furnish such information to Lender as it may reasonably request. In the event the Borrower provides the notice described in the previous sentence, the Lender shall consult with the Borrower in good faith to determine what action may be required to avoid or reduce such Taxes and shall use reasonable efforts to avoid or reduce such Taxes, *provided* that no action shall be required to be taken that would be disadvantageous to the Lender and would result in significantly increased cost to such Lender.

(d) If the Lender is not incorporated under the laws of the United States, any state thereof or the District of Columbia, prior to the first payment to the Lender under this Agreement, the Lender shall deliver to the Borrower (i) two duly completed copies of United States Internal Revenue Service Form W-8BEN or Form W-8ECI (or applicable successor form and any related forms as may from time to time be adopted to document a claim to which such form relates), and (ii) a duly completed copy of United States Internal Revenue Service Form W-8 or W-9 (or applicable successor form and any related forms as may from time to time be adopted to document a claim to which such form relates). The Lender shall certify that (x) in the case of any form provided pursuant to clause (i) of the preceding sentence it is entitled to receive payments hereunder without deduction or withholding of any United States federal income taxes and (y) in the case of any form provided pursuant to clause (ii) of the preceding sentence it is entitled to receive payments hereunder without deduction or withholding of any United States federal backup withholding taxes. The Lender also agrees to deliver further copies of said Form W-8BEN, W-8ECI or W-9 (or applicable successor form and any related forms as may from time to time be adopted to document a claim to which such form relates), and any related certification as described in the preceding sentence, as the case may be, (i) on or before the date that any such form previously provided expires or becomes obsolete, (ii) after the occurrence of any event requiring a change in the most recent form previously provided unless a change in a treaty, law or regulation has occurred prior to the date on which delivery would otherwise be required that renders all such forms inapplicable or that would prevent the Lender from duly completing and delivering any such form with respect to it and the Lender so advises the Borrower and (iii) upon reasonable request of the Borrower.

2.17. Obligation to Mitigate.

(a) If the Administrative Agent or any Lender determines, in its sole and reasonable discretion, that it has received a refund of any Taxes as to which it has been indemnified by the Borrower or with respect to which the Borrower has paid additional amounts pursuant to Section 2.16, it shall pay to the Borrower an amount equal to such refund, as determined in good faith by the Administrative Agent or the Lender in its sole and reasonable discretion (but only to the extent of indemnity payments made, or additional amounts paid, by the Borrower under Section 2.16 with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses incurred by the Administrative Agent or the Lender, as the case may be, and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund); provided that the Borrower, upon the request of the Administrative Agent or the Lender, agrees to repay the amount paid over to the Borrower (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) to the Administrative Agent or the Lender in the event the Administrative Agent or the Lender is required to repay such refund to such governmental authority. This subsection shall not be construed to require the Administrative Agent or the Lender to make available its tax returns (or any other information relating to its taxes that it deems confidential) to the Borrower or any other Person.

2.18. <u>Determination of Borrowing Base; Eligible Receivables.</u>

(a) The Borrowing Base at any time shall be determined by reference to the most recent Borrowing Base Certificate delivered to the Administrative Agent as may be modified by the Administrative Agent in good faith (subject to the following clause (ii)) to

assure that the Borrowing Base is accurately calculated in accordance with the terms of this Agreement.

(b) Subject to the relevant terms and provisions set forth in this Agreement, including, without limitation, Section 9.5, the Administrative Agent shall be entitled at all times to make reasonable changes to the Eligibility Criteria, upon thirty (30) days prior written notice to the Borrower setting forth such changes and consultation with the Borrower and CPS; provided, that such changes shall (1) only apply to Receivables originated or purchased after the date of such change and (2) be effected only to the extent necessary to reflect circumstances that adversely impact the Lender's valuation of the Receivables.

SECTION CONDITIONS PRECEDENT

- **3.1.** Closing Date. The obligation of the Lender to make a Credit Extension on the Closing Date is subject to the satisfaction, or waiver in accordance with Section 9.5, of the following conditions on or before the Closing Date:
- (a) <u>Credit Documents</u>. The Administrative Agent shall have received copies of each Credit Document originally executed and delivered by each applicable Credit Party.
- (b) <u>Organizational Documents; Incumbency.</u> The Administrative Agent shall have received copies of (i) each Organizational Document executed and delivered by each Credit Party, as applicable, and, to the extent applicable, certified as of a recent date by the appropriate governmental official, each dated the Closing Date or a recent date prior thereto; (ii) signature and incumbency certificates of the officers of such Person executing the Credit Documents to which it is a party; (iii) resolutions of the board of directors, board of managers or similar governing body of each Credit Party approving and authorizing the execution, delivery and performance of this Agreement and the other Credit Documents and the Related Agreements to which it is a party or by which it or its assets may be bound as of the Closing Date, certified as of the Closing Date by its secretary or an assistant secretary as being in full force and effect without modification or amendment; (iv) a good standing certificate from the applicable Governmental Authority of each Credit Party's jurisdiction of incorporation, organization or formation, each dated a recent date prior to the Closing Date; and (v) such other documents as the Administrative Agent may reasonably request.
- (c) <u>Consummation of Transactions Contemplated by Related Agreements.</u> The Administrative Agent shall have received a fully executed or conformed copy of each Related Agreement and any documents executed in connection therewith. Each Related Agreement shall be in full force and effect, shall include terms and provisions reasonably satisfactory to the Administrative Agent and no provision thereof shall have been modified or waived in any respect determined by the Administrative Agent to be material, in each case without the consent of the Administrative Agent.
- (d) <u>Transaction Costs</u>. On or prior to the Closing Date, the Borrower shall have delivered to the Administrative Agent the Borrower's reasonable best estimate of the Transaction Costs (other than fees payable to any Agent).

(e) Governmental Authorizations and Consents. Each Credit Party shall have obtained all Governmental Authorizations and all consents
of other Persons, in each case that are necessary or advisable in connection with the transactions contemplated by the Credit Documents and the Related
Agreements and each of the foregoing shall be in full force and effect and in form and substance reasonably satisfactory to the Administrative Agent. All
applicable waiting periods shall have expired without any action being taken or threatened by any competent authority which would restrain, prevent or
otherwise impose adverse conditions on the transactions contemplated by the Credit Documents or the Related Agreements or the financing thereof and no
action, request for stay, petition for review or rehearing, reconsideration, or appeal with respect to any of the foregoing shall be pending, and the time for any
applicable agency to take action to set aside its consent on its own motion shall have expired.

- (f) <u>Collateral</u>. In order to create in favor of the Collateral Agent, for the benefit of Secured Parties, a valid, perfected first priority Lien in the Collateral and in the Capital Stock of the Borrower, the Collateral Agent shall have received:
 - (i) evidence satisfactory to the Administrative Agent of the compliance by each of the Originator and the Borrower of their obligations under the Security Agreement and the other Collateral Documents (including, without limitation, their obligations to authorize or execute, as the case may be, and deliver UCC financing statements, originals of securities, instruments and chattel paper and any agreements governing deposit accounts as provided therein);
 - (ii) the results of a recent search of all effective UCC financing statements (or equivalent filings) made with respect to any personal or mixed property of the Originator in California and the Borrower in Delaware together with copies of all such filings disclosed by such search, which shall be provided by CPS;
 - (iii) UCC termination statements (or similar documents) duly approved by all applicable Persons for filing in all applicable jurisdictions as may be necessary to terminate any effective UCC financing statements (or equivalent filings) disclosed in such search with respect to the Collateral;
 - (iv) evidence that the Originator and the Borrower shall have taken or caused to be taken any other action, executed and delivered or caused to be executed and delivered any other agreement, document and instrument and made or caused to be made any other filing and recording (other than as set forth herein) reasonably required by the Collateral Agent or the Administrative Agent;
 - (v) opinions of counsel (which counsel shall be reasonably satisfactory to the Collateral Agent) with respect to the creation and perfection of the security interests in favor of the Collateral Agent in such Collateral and such other matters governed by the laws of each jurisdiction in which the Borrower or the Collateral is located as the Administrative Agent may reasonably request, in each case in form and substance reasonably satisfactory to the Administrative Agent; and

- (vi) evidence that any Indebtedness (other than the Obligations) secured by the Collateral has been paid in full.
- (g) <u>Financial Statements; Forecasts</u>. The Administrative Agent shall have received from CPS (i) any historical financial information regarding the Credit Parties requested by the Administrative Agent, (ii) any financial projections with respect to the Credit Parties requested by the Administrative Agent and (iii) any other financial information regarding the Credit Parties as the Administrative Agent, in its sole discretion, may request.
- (h) Evidence of Insurance. The Collateral Agent shall have received a certificate from the Servicer's insurance broker, or other evidence satisfactory to it that all insurance required to be maintained hereunder and under the Servicing Agreement is in full force and effect, and the Administrative Agent shall have completed its review of the insurance coverage for CPS and the Borrower and the results of such review shall be satisfactory to the Administrative Agent.
- (i) <u>Opinions of Counsel to Credit Parties</u>. The Administrative Agent shall have received originally executed copies of the favorable written opinions of Andrews Kurth LLP, counsel for CPS and the Borrower, and internal counsel for CPS and the Borrower, and as to such matters as the Administrative Agent may reasonably request, dated as of the Closing Date and otherwise in form and substance reasonably satisfactory to the Administrative Agent and its counsel (and CPS and the Borrower hereby instruct counsel for CPS and the Borrower to deliver such opinions to the Agents and the Lender).
 - (j) <u>Upfront Fee</u>. The Borrower and/or CPS shall have paid to the Lender the fee payable on the Closing Date referred to in the Fee Letter.
- (k) <u>Solvency Certificate</u>. On the Closing Date, the Administrative Agent shall have received Solvency Certificates from the Borrower and CPS dated as of the Closing Date and addressed to the Administrative Agent and Lender, substantially in the form of <u>Exhibit E-2</u>, attesting that before and after giving effect to the consummation of the transactions contemplated by the Credit Documents and the Related Agreements, each of the Borrower and CPS, as the case may be, is Solvent.
- (l) <u>Closing Date Certificate</u>. Each of CPS and the Borrower shall have delivered to the Administrative Agent an originally executed Closing Date Certificate, together with all attachments thereto.
- (m) No <u>Litigation</u>. There shall not exist any action, suit, investigation, litigation or proceeding or other legal or regulatory developments, pending or threatened in any court or before any arbitrator or Governmental Authority that, in the reasonable opinion of the Administrative Agent, singly or in the aggregate, materially impairs the transactions contemplated by the Related Agreements or any of the other transactions contemplated by the Credit Documents, or that could reasonably be expected to have a Material Adverse Effect, except has been previously disclosed in writing to the Administrative Agent or in CPS' public reports filed under the Exchange Act, provided that CPS has notified the Administrative Agent of such filing.

No New Information. The Administrative Agent shall not have become aware, since August 31, 2009, of any new information or

No Closing Date Material Adverse Change. A Closing Date Material Adverse Change shall not have occurred.

- other matters not previously disclosed to the Administrative Agent relating to the Borrower, CPS or its Subsidiaries or the transactions contemplated herein which the Administrative Agent, in its reasonable judgment, deems inconsistent in a material and adverse manner with the information or other matters previously disclosed to the Administrative Agent relating to the Borrower, CPS or its Subsidiaries.
- Delivery of Receivable Files to Custodian; Collateral Receipt. In accordance with the terms of the Custodial Agreement, the Borrower has delivered, or caused to be delivered, to the Custodian, the related Receivable File and the Collateral Agent has received a Collateral Receipt and Exception Report from the Custodian, which Collateral Receipt and Exception Report is acceptable to the Collateral Agent in its sole discretion.
- Service of Process. On the Closing Date, the Administrative Agent shall have received evidence that each of CPS and the Borrower has appointed an agent in New York City for the purpose of service of process in New York City and such agent shall agree in writing to give the Administrative Agent notice of any resignation of such service agent or other termination of the agency relationship.
- Borrowing Base Confirmation. The Administrative Agent shall have received a Borrowing Base Certificate evidencing that there is sufficient Revolving Availability with respect to the initial Credit Extension requested by the Borrower.
- Servicing Report. The Administrative Agent shall have received a pro forma Monthly Servicing Report as of August 31, 2009, acceptable to the Administrative Agent in its sole discretion.

(t) **Diligence**

- The Administrative Agent shall have completed its diligence, including but not limited to, legal diligence and due diligence on the Credit Parties and the results of such diligence are satisfactory to the Administrative Agent in its sole discretion;
- The Administrative Agent shall be satisfied, in its sole discretion, with the results of background investigations, if any, performed on principals and employees of any Credit Party, including, without limitation, Charles Bradley, Jr.; and
- The Administrative Agent shall be satisfied, in its sole discretion, with the Credit Parties' cash management systems and the Servicer's procedures and systems.
- Backup Servicing. The Backup Servicer shall have completed all required data mapping as set forth in the Backup Servicing Agreement.

3.2. Conditions to Each Credit Extension.

- (a) <u>Conditions Precedent</u>. The obligation of the Lender to make any Revolving Loan, on any Credit Date, including the Closing Date, are subject to the satisfaction, or waiver in accordance with Section 9.5, of the following conditions precedent:
 - (i) the Administrative Agent shall have received a fully executed and delivered Funding Notice together with a Borrowing Base Certificate two (2) Business Days prior to such Credit Date, evidencing sufficient Revolving Availability with respect to the requested Revolving Loan;
 - (ii) after making the Credit Extensions requested on such Credit Date, the Total Utilization of Revolving Commitments as of such Credit Date shall not exceed the lesser of (A) the Revolving Commitments then in effect and (B) the Borrowing Base;
 - (iii) as of such Credit Date, the representations and warranties made by each of the Credit Parties contained herein and in the other Credit Documents shall be true and correct in all material respects on and as of that Credit Date to the same extent as though made on and as of that date, except to the extent such representations and warranties specifically relate to an earlier date, in which case such representations and warranties shall have been true and correct in all material respects on and as of such earlier date;
 - (iv) as of such Credit Date, after giving effect to such Revolving Loan, no event shall have occurred and be continuing or would result from the consummation of the applicable Credit Extension that would constitute an Event of Default or a Default;
 - (v) as of such Credit Date, after giving effect to such Revolving Loan, the Spread Account is fully funded at the Required Spread Account Amount;
 - (vi) as of such Credit Date, the Collateral Agent shall have received satisfactory evidence of the valid transfer of the Eligible Receivables comprising the Borrowing Base to the Borrower; and
 - (vii) in accordance with the terms of the Custodial Agreement, the Borrower has delivered, or caused to be delivered, to the Custodian, the related Receivable File and the Collateral Agent has received a Collateral Receipt and Exception Report from the Custodian, which Collateral Receipt and Exception Report is acceptable to the Collateral Agent in its sole discretion.

Any Agent shall be entitled, but not obligated to, request and receive, prior to the making of any Credit Extension, additional information reasonably satisfactory to the requesting party confirming the satisfaction of any of the foregoing if, in the good faith judgment of such Agent such request is warranted under the circumstances.

(b) <u>Funding Notices</u>. Any Funding Notice shall be executed by an Authorized Officer of the Borrower delivered to the Administrative Agent. In lieu of delivering a Funding Notice, the Borrower may give the Administrative Agent telephonic notice by the required time

of any proposed borrowing; <u>provided</u> each such notice shall be promptly confirmed in writing by delivery of the applicable Funding Notice and Borrowing Base Certificate to the Administrative Agent on or before the applicable Credit Date. Neither the Administrative Agent nor the Lender shall incur any liability to the Borrower in acting upon any telephonic notice referred to above that the Administrative Agent believes in good faith to have been given by a duly Authorized Officer of the Borrower.

SECTION REPRESENTATIONS AND WARRANTIES

In order to induce Agents and the Lender to enter into this Agreement and to make each Credit Extension to be made thereby, each of the Borrower and CPS (with respect to itself) represents and warrants, to each Agent and the Lender, on the Closing Date and on each Credit Date, that the following statements are true and correct (it being understood and agreed that the representations and warranties made on the Closing Date are deemed to be made concurrently with the consummation of the transactions contemplated by Related Agreements):

- **4.1.** Organization; Requisite Power and Authority; Qualification; Other Names. Each of the Borrower and CPS (a) is duly organized or formed, validly existing and in good standing under the laws of the State of its organization, (b) has all requisite power and authority to own and operate its properties, to carry on its business as now conducted and as proposed to be conducted, to enter into the Credit Documents and the Related Agreements to which it is a party, and to carry out the transactions contemplated thereby and fulfill its Obligations thereunder, (c) is qualified to do business and in good standing in every jurisdiction where its assets are located and wherever necessary to carry out its business and operations, except in jurisdictions where the failure to be so qualified or in good standing has not had, and could not be reasonably expected to have, a Material Adverse Effect. The Borrower does not operate or do business under any assumed, trade or fictitious name. The Borrower has no Subsidiaries.
- **4.2.** <u>Capital Stock and Ownership</u>. The Capital Stock of the Borrower has been duly authorized and validly issued and is fully paid and non-assessable. There is no existing option, warrant, call, right, commitment or other agreement to which the Borrower is a party requiring, and there is no Capital Stock of the Borrower outstanding which upon conversion or exchange would require, the issuance by the Borrower of Capital Stock of the Borrower or other Securities convertible into, exchangeable for or evidencing the right to subscribe for or purchase, Capital Stock of the Borrower.
- **4.3. Due Authorization**. The execution, delivery and performance of the Credit Documents and the Related Agreements to which each Credit Party is a party have been duly authorized by all necessary action on the part of such Credit Party.
- **4.4.** No Conflict. The execution, delivery and performance by each Credit Party of the Credit Documents and the Related Agreements to which it is a party and the consummation of the transactions contemplated by the Credit Documents and the Related Agreements do not and will not (a) (i) violate any provision of any law or any governmental rule or regulation applicable to such Credit Party, (ii) violate any of the Organizational Documents of such Credit Party, or (iii) violate any order, judgment or decree of any court or other agency of government binding on such Credit Party; (b) conflict with, result in a breach of or constitute (with due notice

or lapse of time or both) a default under any Contractual Obligation of such Credit Party, except as could not reasonably be expected to result in a Material Adverse Effect; (c) result in or require the creation or imposition of any Lien upon any of the properties or assets of such Credit Party (other than any Liens created under any of the Credit Documents in favor of the Collateral Agent, on behalf of the Secured Parties); or (d) require any approval of stockholders, members or partners or any approval or consent of any Person under any Contractual Obligation of such Credit Party, except for such approvals or consents which will be obtained on or before the Closing Date and disclosed in writing to the Administrative Agent.

- **4.5.** Governmental Consents. The execution, delivery and performance by each Credit Party of the Credit Documents and the Related Agreements to which it is a party and the consummation of the transactions contemplated by the Credit Documents and the Related Agreements do not and will not require any registration with, consent or approval of, or notice to, or other action to, with or by, any Governmental Authority, except for filings and recordings with respect to the Collateral to be made, or otherwise delivered to the Collateral Agent for filing and/or recordation, as of the Closing Date.
- **4.6. Binding Obligation**. Each Credit Document and each Related Agreement to which each Credit Party is a party has been duly executed and delivered by such Credit Party and is the legally valid and binding obligation of such Credit Party and is in full force and effect, enforceable against such Credit Party in accordance with its respective terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws relating to or limiting creditors' rights generally or by equitable principles relating to enforceability.
- **4.7. Receivables.** Each Receivable is a bona fide existing payment obligation of a Obligor, owed to the Borrower without defenses, disputes, offsets, counterclaims, or rights of return or cancellation. Each Receivable that is identified by the Borrower as an Eligible Receivable on a Borrowing Base Certificate or Funding Notice, or by the Servicer on a Monthly Servicing Report, satisfies the Eligibility Criteria. No Depository Institution assisted in the origination of any Receivable and at no time has any Receivable been owned, purchased, or serviced by a Depository Institution.
- **4.8. No Material Adverse Effect**. Since August 31, 2009, no event, circumstance or change has occurred that has caused or evidences, either individually or in the aggregate, a Material Adverse Effect.
- **4.9.** Adverse Proceedings, etc. There are no Adverse Proceedings pending, individually or in the aggregate, that could reasonably be expected to have a Material Adverse Effect. No Credit Party is (a) in violation of any applicable laws that, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect, or (b) subject to or in default with respect to any final judgments, writs, injunctions, decrees, rules or regulations of any court or any federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, that, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.
- **4.10.** Payment of Taxes. Except as otherwise permitted under Section 5.3, all tax returns and reports of the Borrower required to be filed have been timely filed, and all taxes

shown on such tax returns to be due and payable and all assessments, fees and other governmental charges upon the Borrower and upon its properties, assets, income, businesses and franchises which are due and payable have been paid when due and payable. The Borrower knows of no proposed tax assessment against it which is not being actively contested by the Borrower in good faith and by appropriate proceedings; <u>provided</u>, such reserves or other appropriate provisions, if any, as shall be required in conformity with GAAP shall have been made or provided therefor.

- **4.11. Title to Assets**. The Borrower has good and valid title to all of its assets reflected in the most recent financial statements delivered pursuant to Section 5.13. Except as permitted by this Agreement, all such properties and assets are free and clear of Liens, other than Permitted Liens.
- **4.12. No Indebtedness.** The Borrower does not have any Indebtedness, other than Indebtedness incurred under (or contemplated by) the terms of this Agreement, the other Credit Documents or otherwise permitted hereunder.
- **4.13.** No Defaults. Other than the defaults set forth on Schedule 1 hereto, no Credit Party is in default in the performance, observance or fulfillment of any of the obligations, covenants or conditions contained in any of its Contractual Obligations, and to each Credit Party's knowledge no condition exists which, with the giving of notice or the lapse of time or both, could constitute such a default, except where, (a) such defaults have been waived, or (b) individually or in the aggregate, the consequences, direct or indirect, of such default or defaults, if any, could not reasonably be expected to have a Material Adverse Effect.
- **4.14. Governmental Regulation**. The Borrower is not subject to regulation under the Investment Company Act of 1940 or under any other federal or state statute or regulation which may limit its ability to incur Indebtedness or which may otherwise render all or any portion of the Obligations unenforceable. The Borrower is not a **"registered investment company"** or a company **"controlled"** by a **"registered investment company"** or a **"principal underwriter"** of a **"registered investment company"** as such terms are defined in the Investment Company Act of 1940.
- **4.15.** Margin Stock. The Borrower is not engaged in the business of extending credit for the purpose of purchasing or carrying any Margin Stock. No part of the proceeds of the Revolving Loans made to the Borrower will be used directly or indirectly to purchase or carry any such Margin Stock, for the purpose of reducing or retiring any Indebtedness which was originally incurred to purchase or carry Margin Stock, to extend credit to others for the purpose of purchasing or carrying any such Margin Stock or for any purpose that violates, or is inconsistent with, the provisions of Regulation T, U or X of the Board of Governors of the Federal Reserve System.

4.16. [Reserved].

4.17. <u>Certain Fees</u>. No broker's or finder's fee or commission will be payable by the Borrower with respect hereto or any of the transactions contemplated hereby.

4.18. Solvency and Fraudulent Conveyance. The Borrower is and, upon the incurrence of any Credit Extension by the Borrower on any date on which this representation and warranty is made, will be, Solvent. Neither the Borrower nor CPS is transferring any Collateral with any intent to hinder, delay or defraud any of its creditors. Neither the Borrower nor CPS shall use the proceeds from the transactions contemplated by this Agreement to give preference to any class of creditors. The Borrower has given fair consideration and reasonably equivalent value in exchange for the sale of the Receivables under the Purchase Agreement.

4.19. Related Agreements.

- (a) <u>Delivery</u>. The Borrower has delivered, or caused to be delivered, to the Administrative Agent complete and correct copies of (i) each Related Agreement and of all exhibits and schedules thereto as of the date hereof, and (ii) copies of any material amendment, restatement, supplement or other modification to or waiver of each Related Agreement entered into after the date hereof.
- (b) Representations and Warranties. Except to the extent otherwise expressly set forth herein or in the schedules hereto, and subject to the qualifications set forth therein, each of the representations and warranties given by each of the Borrower and CPS in any Related Agreement is true and correct in all material respects as of the Closing Date (or as of any earlier date to which such representation and warranty specifically relates). Notwithstanding anything in the Related Agreement to the contrary, the representations and warranties of each of the Borrower and CPS set forth in any Related Agreement shall, solely for purposes hereof, survive the Closing Date for the benefit of the Lender.
- (c) <u>Governmental Approvals</u>. All Governmental Authorizations and all other authorizations, approvals and consents of any other Person required by the Related Agreements or to consummate the transactions contemplated therein have been obtained and are in full force and effect.
- (d) <u>Conditions Precedent</u>. On the Closing Date, all of the conditions to effecting or consummating the transactions described herein and set forth in the Credit Documents and in the Related Agreements have been duly satisfied or, with the consent of the Administrative Agent, waived, and the transactions described set forth in the Related Agreements, have been consummated in accordance with the Related Documents
- **4.20.** <u>Compliance with Statutes, etc.</u> Each Credit Party is in compliance with all applicable statutes, regulations and orders of, and all applicable restrictions imposed by, all Governmental Authorities, in respect of the conduct of its business and the ownership of its property, except such non-compliance that, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect.
- **4.21.** <u>Disclosure</u>. The representations and warranties of each of the Borrower and CPS contained in any Credit Document or in any other documents, certificates or written statements furnished to the Lender by or on behalf of CPS or any of its Subsidiaries for use in connection with the transactions contemplated hereby, taken as a whole, do not contain any untrue statement of a material fact or omit to state a material fact (known to CPS or the Borrower, in the case of

any document not furnished by either of them) necessary in order to make the statements contained herein or therein not misleading in light of the circumstances in which the same were made. Any projections and pro forma financial information contained in such materials are based upon good faith estimates and assumptions by the preparer thereof believed to be reasonable at the time made, it being recognized by the Lender that such projections as to future events are not to be viewed as facts and that actual results during the period or periods covered by any such projections may differ from the projected results. There are no facts known (or which should upon the reasonable exercise of diligence be known) to the Borrower (other than matters of a general economic nature) that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect and that have not been disclosed herein or in such other documents, certificates and statements furnished to the Lender for use in connection with the transactions contemplated hereby.

4.22. Money Control Acts/FCPA. To the extent applicable, each of the Borrower and CPS is in compliance, in all material respects, with the (a) Trading with the Enemy Act, as amended, and each of the foreign assets control regulations of the United States Treasury Department (31 C.F.R., Subtitle B, Chapter V, as amended) and any other enabling legislation or executive order relating thereto, and (b) Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism (USA Patriot Act of 2001) (the "Act"). No part of the proceeds of the Revolving Loans will be used, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.

SECTION AFFIRMATIVE COVENANTS

5.

Each of CPS and the Borrower covenants and agrees that so long as any Revolving Commitment is in effect and until payment in full of all of the Obligations (other than contingent indemnification obligations), CPS and/or the Borrower, as applicable, shall perform all covenants in this Section 5.

- **5.1.** Reports. Unless otherwise provided below, the Borrower shall deliver, or cause to be delivered, to the Administrative Agent:
- (a) <u>Collateral Reporting</u>. On each Credit Date and at such other times as the Administrative Agent shall request, the Borrower shall deliver a Borrowing Base Certificate to the Administrative Agent, in form and substance satisfactory to the Administrative Agent. Each Borrowing Base Certificate delivered to the Administrative Agent shall bear a signed statement by an Authorized Officer certifying the accuracy and completeness of all information included therein. The execution and delivery of a Borrowing Base Certificate shall in each instance constitute a representation and warranty by the Borrower to the Lender that each Eligible Receivable included therein satisfies the Eligibility Criteria. In the event any Funding Notice with respect to a Revolving Loan or other information required by this Section 5.1(a) is delivered to the Administrative Agent by the Borrower electronically or otherwise without signature, such Funding Notice or other information shall, upon such delivery, be deemed to be signed and certified on behalf of the Borrower by an Authorized Officer and constitute a representation to

the Administrative Agent as to the authenticity thereof. The Administrative Agent shall have the right to review and adjust any such calculation of the Borrowing Base in accordance with Section 2.18 hereof;

- (b) Notice of Default. Promptly upon any Authorized Officer of CPS or the Borrower obtaining knowledge (i) of any condition or event that constitutes a Default or an Event of Default; (ii) that any Person has given any notice to CPS or the Borrower or taken any other action with respect to any event or condition set forth in Section 7.1(b) (captioned "Cross-Defaults"); or (iii) of the occurrence of any event or change that has caused or evidences, either individually or in the aggregate, a Material Adverse Effect, a certificate of its Authorized Officers specifying the nature and period of existence of such condition, event or change, or specifying the notice given and action taken by any such Person and the nature of such claimed Event of Default, Default, default, event or condition, and what action CPS or the Borrower, as applicable, has taken, is taking and proposes to take with respect thereto;
- (c) Notice of Litigation. Promptly upon any Authorized Officer of CPS or the Borrower obtaining knowledge of (i) the institution of, or non-frivolous threat of, any Adverse Proceeding not previously disclosed in writing by CPS or the Borrower to the Lender, or (ii) any material development in any Adverse Proceeding that, in the case of either clause (i) or (ii) if adversely determined, is reasonably likely to result in a judgment in an amount in excess of \$250,000, or seeks to enjoin or otherwise prevent the consummation of, or to recover any damages or obtain relief as a result of, the transactions contemplated hereby, written notice thereof together with such other information as may be reasonably available to CPS or the Borrower, as applicable, to enable the Lender and its counsel to evaluate such matters;

(d) [Reserved].

- (e) <u>Breach of Representations and Warranties</u>. Promptly upon CPS or the Borrower becoming aware of a material breach with respect to any representation or warranty made or deemed made by CPS or the Borrower in any Credit Document or in any certificate at any time given by CPS or the Borrower in writing pursuant hereto or thereto or in connection herewith or therewith, a certificate of its Authorized Officers specifying the nature and period of existence of such breach and what action CPS or the Borrower, as applicable, has taken, is taking and proposes to take with respect thereto;
- (f) <u>Information Regarding Collateral</u>. The Borrower will furnish to the Collateral Agent prior written notice of any change (i) in its corporate name, (ii) in its identity, corporate structure or jurisdiction of organization, or (iii) in its Federal Taxpayer Identification Number. The Borrower agrees not to effect or permit any change referred to in the preceding sentence unless all filings have been made under the UCC or otherwise that are required in order for the Collateral Agent to continue at all times following such change to have a valid, legal and perfected security interest in all the Collateral. The Borrower also agrees promptly to notify the Collateral Agent if any material portion of the Collateral is damaged or destroyed; and
- (g) <u>Tax Returns</u>. As soon as practicable and in any event within fifteen (15) days following the filing thereof, the Borrower shall provide to the Administrative Agent copies of each federal income tax return filed by or on behalf of the Borrower.

- **5.2.** Existence. Each of CPS and the Borrower shall at all times preserve and keep in full force and effect its existence and all rights and franchises, licenses and permits material to its business.
- **5.3.** Payment of Taxes and Claims. The Borrower shall pay all Taxes imposed upon it or any of its properties or assets or in respect of any of its income, businesses or franchises before any penalty or fine accrues thereon, and all claims (including claims for labor, services, materials and supplies) for sums that have become due and payable and that by law have or may become a Lien upon any of its properties or assets, prior to the time when any penalty or fine shall be incurred with respect thereto; provided, no such Tax or claim need be paid if it is being contested in good faith by appropriate proceedings promptly instituted and diligently conducted, so long as (a) adequate reserve or other appropriate provision, as shall be required in conformity with GAAP shall have been made therefor, and (b) in the case of a Tax or claim which has or may become a Lien against any of the Collateral, such contested proceedings conclusively operate to stay the sale of any portion of the Collateral to satisfy such Tax or claim. The Borrower shall not file or consent to the filing of any consolidated income tax return with any Person (other than CPS or any of its Subsidiaries).
- **5.4.** Audits of Receivables. Each of CPS and the Borrower shall, at any time and as frequently as may be reasonably requested by the Administrative Agent, at the Borrower's expense, permit the Administrative Agent, or its agents or professional advisors, at reasonable times during business hours and upon reasonable notice to CPS or the Borrower, as applicable, to audit the Receivables and to inspect, audit, copy and take extracts from its and their financial and accounting records, and to discuss its and their affairs, financings and accounts with any Person, including, without limitation, employees of the Borrower or CPS and independent public accountants. The Borrower agrees to pay the reasonable out-of-pocket expenses incurred by the Administrative Agent in connection with its field examinations and audits and the preparation of reports thereof performed or prepared. Subject to Section 2.18(b), the Administrative Agent may redetermine the value of Eligible Receivables based on any such audits, or updates of audits, and, as a result, redetermine the Borrowing Base.
- **5.5.** <u>Annual Meetings.</u> Each of CPS and the Borrower will, upon the request of the Administrative Agent or the Lender, participate in a meeting of the Administrative Agent and the Lender once during each Fiscal Year to be held at the Borrower's corporate offices (or at such other location as may be agreed to by CPS, the Borrower, the Administrative Agent and the Lender) at such time as may be agreed to by CPS, the Borrower, the Administrative Agent and the Lender.
- **5.6.** <u>Compliance with Laws</u>. Each of CPS and the Borrower shall comply with the requirements of all applicable laws, rules, regulations and orders of any Governmental Authority noncompliance with which could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- **5.7. Further Assurances**. At any time or from time to time upon the request of the Administrative Agent, each of CPS and the Borrower will, at its expense, promptly execute, acknowledge and deliver such further documents and do such other acts and things as the Administrative Agent or the Collateral Agent may reasonably request in order to effect fully the

			C . 1' I	7	1 1 1		41	т	1.1.		` - · · · - · · · · · ·				C 1 1	Λ 1Λ
п	nur	JOSES OT THE	C real L	Jocuments	incilialno	nrowining	THE	i ender	with a	nw int	ormation.	reasonanis	7 reallestea	pursuant to	Section	чтч
	Jui	JOSES OF LIFE	Cicuiti	Jocuminemo,	meruanis	providing	uic	LCHUCI	williu	11 9 1111	OIIIIuuioii	i cusoniubi	requesteu	puisuuii to	occuon.	0.10

- **5.8.** Separateness. The Borrower acknowledges that the Lender is entering into this Agreement in reliance upon the Borrower's identity as a legal entity that is separate from any other Person. Therefore, from and after the date of this Agreement, the Borrower shall take all reasonable steps, including without limitation, all steps that the Lender may from time to time reasonably request, to maintain the Borrower's identity as a separate legal entity and to make it manifest to third parties that the Borrower is a separate legal entity. Without limiting the generality of the foregoing, the Borrower agrees that it has not and shall not:
- (a) engage, either directly or indirectly, in any business or activity other than the acquisition, ownership, financing and disposition of the Receivables in accordance with the Credit Documents and the Related Agreements and activities incidental thereto;
 - (b) acquire or own any material asset other than the Collateral and proceeds thereof;
- (c) merge into or consolidate with any Person or entity or dissolve, terminate or liquidate in whole or in part, transfer or otherwise dispose of all or substantially all of its assets or change its legal structure, without in each case, to the extent permitted by law, the Administrative Agent's consent;
- (d) fail to preserve its existence as an entity duly organized, validly existing and in good standing (if applicable) under the laws of the jurisdiction of its formation, or without the prior written consent of the Administrative Agent, amend, modify, change, repeal, terminate or fail to comply with the provisions of the Borrower's certificate of formation, or its limited liability company agreement, as the case may be; <u>provided</u>, <u>however</u>, the Borrower may amend its operating agreement without the Administrative Agent's consent (i) to cure any ambiguity, (ii) with respect to administrative matters, or (iii) to convert or supplement any provision in a manner consistent with the intent of this Agreement or the other Credit Documents or the Related Agreements;
 - (e) own any Subsidiary or make any investment in, any Person or entity without the consent of the Administrative Agent;
 - (f) commingle its assets with the assets of any of its general partners, members, Affiliates, principals or any other Person or entity;
 - (g) incur any Indebtedness except the Obligations and Indebtedness with respect to any Interest Rate Agreement required hereby;
 - (h) fail to remain Solvent;
- (i) fail to maintain its records, books of account and bank accounts, separate and apart from those of the general partners, members, principals and Affiliates of the Borrower or the Affiliates of a general partner or member of the Borrower or any other Person;

(j) except for the Credit Documents and the Related Agreements, and as otherwise expressly permitted by the Credit Documents, enter
into any contract or agreement with any general partner, member, principal or Affiliate of the Borrower, CPS, or any general partner, member, principal or
Affiliate thereof, except with the Administrative Agent's consent and upon terms and conditions that are intrinsically fair and substantially similar to those
that would be available on an arms-length basis with third parties other than any general partner, member, principal or affiliate of the Borrower, CPS, or any
general partner, member, principal or Affiliate thereof or fail to maintain separate financial statements from those of its general partners, members, principles
and Affiliates; provided, however, the Borrower's financial position, assets, liabilities, net worth and operating results may be included in the consolidated
financial statements of CPS and its Affiliates; provided that such consolidated financial statements disclose that the Borrower is a separate legal entity and
that its assets are not generally available to satisfy the claims of creditors of CPS;

- (k) seek the dissolution or winding up, in whole or in part, of the Borrower or take any action that would cause the Borrower to become insolvent;
 - (1) fail to take reasonable efforts to correct any misunderstanding known to the Borrower regarding the separate identity of the Borrower;
- (m) maintain its assets in such a manner that it will be costly or difficult to segregate, ascertain or identify its individual assets from those of any other Person;
- (n) except as provided in the Credit Documents, assume or guaranty the debts of any other Person, hold itself out to be responsible for the debts of any other Person, or otherwise pledge its assets for the benefit of any other Person or hold out its credit as being available to satisfy the obligations of any other Person;
- (o) except as provided in the Credit Documents, make any loans or advances to any third party, including any general partner, member, principal or affiliate of the Borrower, or any general partner, member, principal or Affiliate thereof;
- (p) fail either to hold itself out to the public as a legal entity separate and distinct from any other entity or Person or to conduct its business solely in its own name in order not (i) to mislead others as to the identity with which such other party is transacting business, or (ii) to suggest that the Borrower is responsible for the debts of any third party (including any general partner, member, principal or Affiliate of the Borrower, or any general partner, member, principal or Affiliate thereof);
- (q) fail to maintain adequate capital for the normal obligations reasonably foreseeable in a business of its size and character and in light of its contemplated business operations;
- (r) file or consent to the filing of any petition, either voluntary or involuntary, to take advantage of any applicable insolvency, bankruptcy, liquidation or reorganization statute, or make an assignment for the benefit of creditors;

- (s) hold itself out as or be considered as a department or division (other than for tax purposes) of any general partner, principal, member or Affiliate of the Borrower or any other Person or entity;
- (t) fail to allocate fairly and reasonably shared expenses (including, without limitation, shared office space and services performed by an employee of an Affiliate) among the Persons sharing such expenses and to use separate stationery, invoices and checks;
 - (u) acquire obligations or securities of its partners, members, shareholders of other Affiliates, as applicable;
- (v) violate or cause to be violated the assumptions made with respect to the Borrower in any opinion letter pertaining to substantive consolidation delivered to the Lender in connection with the Credit Documents;
 - (w) [Reserved];
- (x) fail to have Organizational Documents that provide that, so long as the Obligations of the Borrower shall be outstanding, the Borrower shall not (i) seek the dissolution or winding up in whole, or in part, of the Borrower or (ii) file or consent to the filing of any petition, either voluntary or involuntary, or commence a case under any applicable insolvency, bankruptcy, liquidation or reorganization statute, or make an assignment for the benefit of creditors without the consent of the Independent Manager;
 - (y) fail to observe all requisite organizational formalities under Delaware law;
- (z) account for or treat (whether in financial statements or otherwise) the transactions contemplated by the Purchase Agreement in any manner other than the sale of the Receivables to the Borrower or in any other respect account for or treat the transactions contemplated therein in any manner other than as a sale of Receivables to the Borrower; <u>provided</u>, that the Receivables may be reflected on the consolidated balance sheets of CPS in accordance with GAAP.
 - (aa) make any revision or amendment to the Purchase Agreement without the consent of the Administrative Agent; and
- (bb) fail to cause its members, managers, directors, officers, agents and other representatives to act at all times with respect to the Borrower consistently and in furtherance of the foregoing and in the best interests of the Borrower.

In the event of any inconsistency between the covenants set forth in this Section 5.8 or the other covenants set forth in this Agreement, or in the event that any covenant set forth in this Section 5.8 poses a greater restriction or obligation than is set forth elsewhere in this Agreement, the covenants set forth in this Section 5.8 shall control.

- **5.9.** <u>Cash Management Systems</u>. The Borrower shall establish and maintain cash management systems as set forth below.
 - (a) Lockbox System.
- (i) The Borrower has established, or has caused the Servicer to establish, pursuant to the Control Agreements for the benefit of the Collateral Agent, on behalf of the Secured Parties, a lockbox and Lockbox Account as described in Section 2.11 (the "Lockbox System") into which all Collections shall be deposited.
- (ii) The Borrower shall not establish any new lockbox or lockbox arrangement without consent of the Administrative Agent in its sole discretion, and prior to establishing any such new lockbox or lockbox arrangement, the Borrower shall cause each bank or financial institution with which it seeks to establish such a lockbox or lockbox arrangement to enter into a control agreement with respect thereto.
- (iii) Without the prior written consent of the Administrative Agent, the Borrower shall not, in a manner adverse to the Lender, (A) change the general instructions given to the Servicer in respect of payments on account of Receivables to be deposited in the Lockbox System or (B) change any instructions given to any bank or financial institution which in any manner redirects the proceeds of any collections in the Lockbox System to any account which is not a Controlled Account.
- (iv) The Borrower acknowledges and agrees that (A) the funds on deposit in the Lockbox System shall continue to be collateral security for the Obligations secured thereby, (B) upon the occurrence and during the continuance of an Event of Default, the funds on deposit in the Lockbox System shall be applied as provided in Section 2.10 and (C) the Control Agreements will provide that all funds in the Lockbox will be swept daily into the Collection Account.
- (b) Payment Collection. The Borrower has directed, and will at all times hereafter direct, the Servicer to direct each of the Obligors to forward all payments on account of Receivables directly to the Lockbox System in accordance with Section 2.11. The Borrower agrees (i) to instruct the Servicer to instruct each Obligor to make all payments with respect to Receivables directly to the Lockbox System and (ii) promptly (and, except as set forth in the proviso to this Section 5.9(b), in no event later than two (2) Business Days following receipt) to deposit all payments received by it on account of Receivables, whether in the form of cash, checks, notes, drafts, bills of exchange, money orders or otherwise, in the Lockbox System in precisely the form in which they are received (but with any endorsements of the Borrower necessary for deposit or collection), and until they are so deposited to hold such payments in trust for and as the property of the Collateral Agent; provided, however, that with respect to any payment received that does not contain sufficient identification of the account number to which such payment relates or cannot be processed due to an act beyond the control of the Servicer, such deposit shall be made no later than the second Business Day following the date on which such account number is identified or such payment can be processed, as applicable.

5.10. Spread Account.

- (a) The Borrower shall maintain at all times a minimum balance in the Spread Account equal to the Required Spread Account Amount. On the initial Credit Date, the Required Spread Account Amount shall be \$250,000.
- (b) On each Credit Date and on each Settlement Date, and on any other Business Day, in accordance with Section 2.12, the Borrower shall deposit in the Spread Account an amount equal to the excess of (i) the Required Spread Account Amount over (ii) the balance of the Spread Account on such day (such excess, the "Required Spread Account Deposit").
- (c) After the occurrence of a Spread Account Increase Event, if no Tier 1 Performance Trigger is breached for three (3) consecutive months, such Spread Account Increase Event shall no longer be continuing, the Required Spread Account Percentage shall be reduced by 25% each calendar month until the amount on deposit in the Spread Account equals the Required Spread Account Floor and any Excess Spread Account Amounts (after giving effect to such adjustments to the Required Spread Account Percentage) shall promptly be released to the Borrower; <u>provided</u>, that, no Default or Event of Default has occurred and is continuing, and no Borrowing Base Deficiency exists.
- (d) After the occurrence of a Tier 2 Performance Trigger, all amounts on deposit in the Spread Account shall be applied in accordance with Section 2.10 hereof.
- (e) Any transfers from the Spread Account or reductions in the Required Spread Account Amount resulting from application of this Section 5.10 shall be effected by the Controlled Account Bank at the written direction of the Administrative Agent.
- **5.11.** <u>Interest Rate Agreements</u>. In the event that the Excess Interest Rate is less than or equal to 1.50% for any Collection Period, the Borrower shall promptly enter into and maintain one or more Interest Rate Agreements with one or more Eligible Hedge Counterparties to appropriately hedge the Receivables in a manner satisfactory to the Administrative Agent in form and substance; the Borrower shall observe and perform all covenants and agreements under such Interest Rate Agreements and shall be responsible for any costs incurred with respect to such Interest Rate Agreements.
- **5.12.** Insurance. From the period commencing on the Closing Date until the termination of this Agreement, CPS shall maintain in force an "errors and omissions" insurance policy and an employee fidelity insurance policy, in each case, (i) in an amount not less than \$1,000,000, (ii) in a form reasonably acceptable to the Administrative Agent, (iii) with an insurance company reasonably acceptable to the Administrative Agent and (iv) naming the Administrative Agent, for the benefit of the Lender, as beneficiary and loss payee. Unless otherwise directed by the Administrative Agent, CPS shall prepare and present, on behalf of itself, the Administrative Agent and the Lender, claims under any such policy in a timely fashion in accordance with the terms of such policy, and upon the filing of any claim on any policy described in this Section 5.12, CPS shall promptly notify the Administrative Agent of such claim and deposit, or cause to be deposited, the proceeds of any such claim into the Collection

Account. CPS shall deliver copies of such policies to the Administrative Agent on or prior to the Closing Date together with a certification from the applicable insurance company that such policy is in force on such date. CPS shall deliver proof of maintenance of such policies and payment of premiums no less frequently than annually, in form and substance reasonably acceptable to the Administrative Agent, on the six-month anniversary from the Closing Date and on each succeeding twelve month anniversary thereafter (or if such day is not a Business Day, the next succeeding Business Day).

5.13. <u>Financial Statements</u>.

- (a) As soon as available and no later than ninety (90) days after the end of each fiscal year of the Borrower, the Borrower shall deliver to the Administrative Agent one (1) copy of: (A) the audited balance sheet of the Borrower as of the end of the fiscal year, setting forth in comparative form the figures for the previous fiscal year and accompanied by an opinion of the Independent Accountants stating that such balance sheet presents fairly the financial condition of the companies being reported upon and has been prepared in accordance with GAAP consistently applied (except for changes in application in which such accountants concur); and (B) audited statements of income, stockholders' equity and cash flow of the Borrower for such fiscal year; in each case setting forth in comparative form the figures for the previous fiscal year and accompanied by an opinion of the Independent Accountants stating that such financial statements present fairly the financial condition of the Borrower and have been prepared in accordance with GAAP consistently applied (except for changes in application in which such accountants concur).
- (b) As soon as available and no later than thirty (30) days after the end of each fiscal month in each fiscal year of CPS, the Borrower shall deliver, or cause to be delivered, to the Administrative Agent one (1) copy of: (A) the unaudited consolidated balance sheet of CPS and its consolidated Subsidiaries (including the Borrower) as of the end of such fiscal month, which balance sheet shall be prepared and presented in accordance with, and provide all necessary disclosure required by, GAAP and shall be accompanied by a certificate signed by the financial vice president, treasurer, chief financial officer, chief investment officer or controller of CPS stating that such balance sheet presents fairly the financial condition of CPS and has been prepared in accordance with GAAP consistently applied; and (B) the unaudited consolidated statements of income, stockholders' equity and cash flow of CPS and its consolidated Subsidiaries (including the Borrower) for such fiscal month, which such statements shall be prepared and presented in accordance with, and provide all necessary disclosure required by, GAAP and shall be accompanied by a certificate signed by the financial vice president, treasurer, chief financial officer, chief investment officer or controller of CPS stating that such financial statements present fairly the financial condition and results of operations of CPS and have been prepared in accordance with GAAP consistently applied.
- (c) As soon as available and no later than forty-five (45) days after the end of each fiscal quarter in each fiscal year of CPS, the Borrower shall deliver, or cause to be delivered, to the Administrative Agent one (1) copy of: (A) the unaudited consolidated balance sheet of CPS and its consolidated Subsidiaries (including the Borrower) as of the end of such fiscal quarter, which such balance sheet shall be prepared and presented in accordance with, and provide all necessary disclosure required by GAAP and shall be accompanied by a certificate

signed by the financial vice president, treasurer, chief financial officer, chief investment officer or controller of CPS or another officer of CPS acceptable to the Administrative Agent stating that such balance sheet presents fairly the financial condition of the companies being reported upon and has been prepared in accordance with GAAP consistently applied; and (B) the unaudited consolidated statements of income, stockholders' equity and cash flow of CPS and its consolidated Subsidiaries (including the Borrower) for such fiscal quarter, which such statements shall be prepared and presented in accordance with, and provide all necessary disclosure required by, GAAP and shall be accompanied by a certificate signed by the financial vice president, treasurer, chief financial officer, chief investment officer or controller of CPS or another officer of CPS acceptable to the Administrative Agent stating that such financial statements present fairly the financial condition and results of operations of the companies being reported upon and have been prepared in accordance with GAAP consistently applied;

- (d) As soon as available and no later than ninety (90) days after the end of each fiscal year of CPS, the Borrower shall deliver, or cause to be delivered, to the Administrative Agent one (1) copy of: (A) the audited consolidated balance sheet of CPS and its consolidated Subsidiaries (including the Borrower) as of the end of the fiscal year, setting forth in comparative form the figures for the previous fiscal year and accompanied by an opinion of the Independent Accountants stating that such balance sheet presents fairly the financial condition of the companies being reported upon and has been prepared in accordance with GAAP consistently applied (except for changes in application in which such accountants concur); and (B) the audited consolidated statements of income, stockholders' equity and cash flow of CPS and its consolidated Subsidiaries (including the Borrower) for such fiscal year; in each case setting forth in comparative form the figures for the previous fiscal year and accompanied by an opinion of the Independent Accountants stating that such financial statements present fairly the financial condition of the companies being reported upon and have been prepared in accordance with GAAP consistently applied (except for changes in application in which such accountants concur).
- (e) For so long as CPS is subject to the periodic reporting obligations of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended, Borrower and CPS may comply with the covenants set forth in the preceding paragraphs (c) and (d) by electronic filing of the annual and quarterly reports required by such act; <u>provided</u>, that the Borrower and/or CPS shall notify, or cause to be notified, the Administrative Agent promptly upon any such electronic filing.
- **5.14.** <u>Due Diligence</u>; <u>Access to Certain Documentation</u>. The Administrative Agent, the Lender (and their respective agents or professional advisors) shall have the right under this Agreement, from time to time, at their discretion and upon reasonable prior notice to the relevant party, to examine and audit, during business hours or at such other times as might be reasonable under applicable circumstances, any and all of the books, records, financial statements, credit and collection policies, legal and regulatory compliance, operating and reporting procedures and information systems, their respective directors, officers and employees, or other information and information systems (including without limitation customer service and/or whistleblower hotlines) of the Credit Parties, or held by another for a Credit Party or on its behalf, concerning or otherwise affecting the Receivables or this Agreement. The Administrative Agent (and their respective agents and professional advisors) shall treat as confidential any information obtained

during the aforementioned examinations which is not already publicly known or available; provided, however, that the Administrative Agent (and their respective agents or professional advisors) may disclose such information if required to do so by law or by any regulatory authority.

Upon notice and during regular business hours, each Credit Party agrees to promptly provide the Administrative Agent (and its agents or professional advisors) with access to, copies of and extracts from any and all documents, records, agreements, instruments or information (including, without limitation, any of the foregoing in computer data banks and computer software systems) the Administrative Agent (and its agents or professional advisors) may reasonably require in order to conduct periodic due diligence relating to the Credit Parties in connection with the Receivables and this Agreement.

Each Credit Party will make available to the Administrative Agent (and its agents or professional advisors) knowledgeable financial, accounting, legal and compliance officers for the purpose of answering questions with respect to the Credit Parties and the Receivables and to assist in the Administrative Agent's diligence. In addition, the Borrower shall provide, or shall cause the Servicer to provide, the Administrative Agent with remote access to any electronic Receivable Files and any related documents to the extent CPS or the Borrower provides such access to any Person. Each of CPS and the Borrower agrees that the Administrative Agent will have the right to confirm any information relating to the Receivables directly with the applicable Obligors.

All reasonable costs and expenses incurred by the Administrative Agent (and their respective agents or professional advisors) in connection with the due diligence and other matters outlined in this Section 5.14 shall be Permitted Expenses, which the Borrower shall reimburse to the Administrative Agent, as the case may be, or shall pay or cause to be paid.

Without limiting the generality of the foregoing, the Borrower acknowledges that Lender shall make Revolving Loans to the Borrower based solely upon the information provided by the Credit Parties to the Administrative Agent and the representations, warranties and covenants contained herein, and that the Administrative Agent has the right at any time and from time to time to conduct a partial or complete due diligence review, at their option and, to the extent that the expense of such review exceeds the Permitted Expenses discussed above, at their expense, on some or all of the Receivables, including, without limitation, re-generating the information used to originate such Receivables.

5.15. Financial Covenants. CPS shall maintain (i) as of the end of any fiscal quarter, an Adjusted Tangible Net Worth of at least \$60,000,000 plus 50% of positive net income for each fiscal year after December 31, 2008, (ii) at each month end, unrestricted Cash of at least \$8,500,000, (iii) as of the end of any fiscal quarter, a ratio of Indebtedness (exclusive of nonrecourse debt) to Adjusted Tangible Net Worth of no more than 2.5:1 and (iv) beginning with the fiscal year ended 2010 and each fiscal year thereafter, a positive net income.

SECTION NEGATIVE COVENANTS

Each of CPS and the Borrower covenants and agrees that, so long as any Revolving Commitment is in effect and until payment in full of all Obligations (other than contingent indemnification obligations), CPS and/or the Borrower, as applicable, shall perform all covenants in this Section 6.

- **6.1.** <u>Indebtedness</u>. The Borrower shall not directly or indirectly, create, incur, assume or guaranty, or otherwise become or remain directly or indirectly liable with respect to any Indebtedness, except the Obligations.
- **6.2.** Liens. The Borrower shall not, directly or indirectly, create, incur, assume or permit to exist any Lien on or with respect to any property or asset of any kind (including any document or instrument in respect of goods or accounts receivable) of the Borrower, whether now owned or hereafter acquired, or any income or profits therefrom, or file or permit the filing of, or permit to remain in effect, any financing statement or other similar notice of any Lien with respect to any such property, asset, income or profits under the UCC of any State or under any similar recording or notice statute, except (i) Liens in favor of the Collateral Agent for the benefit of Secured Parties granted pursuant to any Credit Document; (ii) Permitted Liens (provided, that the existence of a non-Permitted Lien on a Financed Vehicle shall not be a breach of this Section 6.2 to the extent that the resulting removal of the related Receivable from the Borrowing Base does not then result in an uncured Borrowing Base Deficiency); and (iii) financing statements (w) naming Consumer Portfolio Services, Inc. as debtor and Page Four Funding LLC as secured party in accordance with the Purchase Agreement, (x) naming Page Four Funding LLC as debtor and Fortress Credit Corp., as Collateral Agent, as the secured party in accordance with the Security Agreement or (z) filed in connection with other Permitted Liens.

6.3. [Reserved]

- **6.4.** <u>Investments</u>. The Borrower shall not make or own any Investment, except Investments in Cash, Cash Equivalents and Receivables, and Permitted Investments in the Controlled Accounts.
- **6.5.** Fundamental Changes; Disposition of Assets; Acquisitions. The Borrower shall not (i) enter into any transaction of merger or consolidation, or liquidate, wind-up or dissolve itself (or suffer any liquidation or dissolution), or (ii) convey, sell, lease or sub-lease (as lessor or sublessor), exchange, transfer or otherwise dispose of, in one transaction or a series of transactions, all or any part of its business, assets (including, but not limited to, the Receivables) or property of any kind whatsoever, whether real, personal or mixed and whether tangible or intangible, whether now owned or hereafter acquired, except the sale, in the ordinary course of business, for Cash of Charge-Off Receivables to a Person that is not an Affiliate of the Borrower, or (iii) acquire by purchase or otherwise the business, property or fixed assets of, or stock or other evidence of beneficial ownership of, any Person or any division or line of business or other business unit of any Person, except Investments made in accordance with Section 6.4. CPS shall not enter into any transaction of merger or consolidation in which CPS is not the surviving

entity, liquidate, wind-up or dissolve itself (or suffer any liquidation or dissolution) without the prior written consent of the Administrative Agent (such consent not to be unreasonably withheld or delayed).

- **6.6.** Material Contracts and Organizational Documents. The Borrower shall not (a) enter into any Material Contract with any Person; (b) agree to any material amendment, restatement, supplement or other modification to, or waiver of, any of its material rights under any Related Agreement after the Closing Date or (c) amend or permit any amendments to its Organizational Documents (other than as permitted by Section 5.8(d)), without in each case obtaining the prior written consent of the Administrative Agent to such amendment, restatement, supplement, modification or waiver, as the case may be.
- **6.7.** Sales and Lease-Backs. The Borrower shall not directly or indirectly become or remain liable as lessee or as a guarantor or other surety with respect to any lease of any property (whether real, personal or mixed), whether now owned or hereafter acquired, which the Borrower (a) has sold or transferred or is to sell or to transfer to any other Person, or (b) intends to use for substantially the same purpose as any other property which has been or is to be sold or transferred by the Borrower to any Person in connection with such lease.
- **6.8.** Transactions with Shareholders and Affiliates. The Borrower shall not directly or indirectly, enter into or permit to exist any transaction (including the purchase, sale, lease or exchange of any property or the rendering of any service) with CPS, any holder of 5% or more of any class of Capital Stock of CPS or any of its Subsidiaries or with any Affiliate of CPS or of any such holder other than the transactions contemplated by the Credit Documents and the Related Agreements.
- **6.9.** Conduct of Business. From and after the Closing Date, the Borrower shall not engage in any business other than the businesses engaged in by the Borrower on the Closing Date.
 - **6.10. Fiscal Year**. The Borrower shall not change its Fiscal Year-end.
- **6.11.** Accounts. The Borrower shall not establish or maintain a deposit account or a securities account that is not the Lockbox Account or a Controlled Account and the Borrower shall not, nor direct any Person to, deposit Collections in a deposit account or a securities account that is not the Lockbox Account or a Controlled Account.
- **6.12.** <u>Underwriting Policies</u>. CPS shall not change or modify the Underwriting Policies in any material respect without the prior written consent of the Administrative Agent (such consent to not be unreasonably withheld or delayed). CPS shall provide written notice to the Administrative Agent of any other change or modification to the Underwriting Policies within ten (10) Business Days of the implementation of any such change or modification.
- **6.13.** <u>Prepayments of Certain Indebtedness</u>. The Borrower shall not, directly or indirectly, voluntarily purchase, redeem, defease or prepay any principal of, premium, if any, interest or other amount payable in respect of any Indebtedness prior to its scheduled maturity, other than the Obligations.

- **6.14. Servicing Agreement and Backup Servicing Agreement**. The Borrower shall not (a) terminate the Servicing Agreement or Backup Servicing Agreement or (b) select a replacement servicer, in each case without the consent of the Administrative Agent.
- **6.15. Independent Manager.** The Borrower shall not fail at any time to have at least one (1) Independent Manager that is not and has not been for at least five (5) years, either (i) a shareholder (or other equity owner) of, or an officer, director, partner, manager, member (other than as a special member in the case of single member Delaware limited liability companies), employee, attorney or counsel of, the Borrower or any of its Affiliates; (ii) a customer or creditor of, or supplier to, the Borrower or any of its Affiliates who derives any of its purchases or revenue from its activities with the Borrower or any Affiliate thereof (other than a *de minimis* amount); (iii) a person who controls or is under common control with any such officer, director, partner, manager, member, employee, supplier, creditor or customer; or (iv) a member of the immediate family of any such officer, director, partner, manager, member, employee, supplier, creditor or customer; provided that the foregoing subclause (i) shall not apply to any Person who serves as an independent director or an independent manager for any Affiliate of the Borrower; provided, further, that upon the death or incapacity of such Independent Manager, the Borrower will have a period of ten (10) Business Days following such event to appoint a replacement Independent Manager; provided, further, that the Borrower shall cause the Independent Manager not to resign until a replacement independent manager has been appointed.
- **6.16.** <u>Asset Sales</u>. The Borrower shall not sell Receivables to any third party without the prior written consent of the Administrative Agent, subject to Section 9.22.

SECTION EVENTS OF DEFAULT 7.

- **7.1.** Events of Default. If any one or more of the following conditions or events shall occur:
- (a) <u>Failure to Make Payments When Due</u>. Other than with respect to a Borrowing Base Deficiency the failure by the Borrower or the Servicer to make any payment or deposit required under any Credit Document within two (2) Business Days of the date such payment or deposit is due; or
- (b) Cross Defaults. (i) Failure of a Credit Party to pay when due any principal of or interest on or any other amount payable in respect of one or more items of Indebtedness in excess of \$250,000 (other than Indebtedness in respect of the Obligations) beyond the grace period, if any, provided therefor; or (ii) breach or default by a Credit Party with respect to any other material term of (1) one or more items of the Indebtedness referred to in clause (i) above, or (2) any loan agreement, mortgage, indenture or other agreement relating to such item(s) of Indebtedness, in each case beyond the grace period, if any, provided therefor, if the effect of such breach or default is to cause that Indebtedness to be declared (in writing, to the extent required by the related loan agreement, mortgage, indenture or other agreement) due and payable (or subject to a compulsory repurchase or redeemable) prior to its stated maturity or the stated maturity of any underlying obligation, as the case may be; or

- (c) <u>Breach of Certain Covenants</u>. Failure of a Credit Party to perform or comply with any term or condition contained in Section 5.2, Section 5.6, Section 5.8, Section 5.9, Section 5.10, Section 6 or Section 9.22(e); or
- (d) <u>Breach of Representations, etc.</u> Any representation, warranty, certification or other statement made or deemed made by any Credit Party in any Credit Document or in any statement or certificate at any time given by any Credit Party or any of its Subsidiaries in writing pursuant hereto or thereto or in connection herewith or therewith shall be false in any material respect as of the date made or deemed made and which adversely affects the Lender's or any Agent's interests and shall not have been remedied or waived within ten (10) Business Days after the earlier of (i) an Authorized Officer of the applicable Credit Party becoming aware of such breach, or (ii) receipt by the Borrower of written notice from the Administrative Agent or the Lender of such breach.
- (e) Other Defaults Under Credit Documents. Any Credit Party shall default in the performance of or compliance with any covenant or other term contained herein or any of the other Credit Documents, other than any such term referred to in any other provision of this Section 7.1, and such default adversely affects the Lender's or any Agent's interests and shall not have been remedied or waived within ten (10) Business Days after the earlier of (i) an Authorized Officer of such Credit Party becoming aware of such default, or (ii) receipt by the Borrower of written notice from the Administrative Agent or the Lender of such default; or
- (f) Involuntary Bankruptcy; Appointment of Receiver, etc. (i) A court of competent jurisdiction shall enter a decree or order for relief (other than a decree or order described in clause (ii)) in respect of any Credit Party in an involuntary case under the Bankruptcy Code or under any other applicable bankruptcy, insolvency or similar law now or hereafter in effect, which decree or order is not stayed; or any other similar relief shall be granted under any applicable federal or state law; or (ii) an involuntary case shall be commenced against any Credit Party under the Bankruptcy Code or under any other applicable bankruptcy, insolvency or similar law now or hereafter in effect; or a decree or order of a court having jurisdiction in the premises for the appointment of a receiver, liquidator, sequestrator, trustee, custodian or other officer having similar powers over such Credit Party, or over all or a substantial part of its property, shall have been entered; or there shall have occurred the involuntary appointment of an interim receiver, trustee or other custodian of such Credit Party for all or a substantial part of its property, and any such event described in this clause (ii) shall continue for sixty (60) days without having been dismissed, bonded or discharged; or
- (g) Voluntary Bankruptcy; Appointment of Receiver, etc. (i) Any Credit Party shall commence a voluntary case under the Bankruptcy Code or under any other applicable bankruptcy, insolvency or similar law now or hereafter in effect, or shall consent to the entry of an order for relief in an involuntary case, or to the conversion of an involuntary case to a voluntary case, under any such law, or shall consent to the appointment of or taking possession by a receiver, trustee or other custodian for all or a substantial part of its property; or the Borrower shall make any assignment for the benefit of creditors; or (ii) any Credit Party shall be unable, or shall fail generally, or shall admit in writing its inability, to pay its debts as such debts become due; or the board of directors (or similar governing body) of such Credit Party (or any

	1 11 1 .	- 4	1	(1	•	41			- C (1-			4 - 1		•		7 1/6	A
committee thereof	า รทลม ลกก	$\mathbf{n}\mathbf{r}$ an \mathbf{w}	resolution	ar otner	พารค สมม	เทกทารค ล	iny action to	i annrowe a	inv of the	e actions	reterren	to ne	rein or	ın 🔨	PCTION	/ 111	ar or
Committee mercor	, siidii ddo	or uniy	i Coolution (ouici	WISC du	1101120	my action to	, αρρίονε α	my or mi	c actions	rerered	to me	CIII OI	111 0	ccuon	/ • - (-	.,, 01

- (h) <u>Judgments and Attachments</u>. Any money judgment, writ or warrant of attachment or similar process involving (i) in any individual case an amount in excess of \$150,000 or (ii) in the aggregate at any time an amount in excess of \$500,000 (in either case to the extent not adequately covered by insurance as to which a solvent and unaffiliated insurance company has not denied coverage) shall be entered or filed against a Credit Party or any of its assets and shall remain undischarged, unvacated, unbonded or unstayed for a period of sixty (60) days (or in any event later than five (5) days prior to the date of any proposed sale thereunder in connection with any enforcement proceedings commenced by a creditor upon such judgment, writ, warrant of attachment or similar process); or
- (i) <u>Dissolution</u>. Any order, judgment or decree shall be entered against any Credit Party decreeing the dissolution or split up of the Borrower and such order shall remain undischarged or unstayed for a period in excess of sixty (60) days; or

(j) [Reserved]

- (k) <u>Change of Control</u>. A Change of Control shall occur with respect to any Credit Party, without the prior written consent of the Administrative Agent; or
- (l) Collateral Documents and other Credit Documents. At any time after the execution and delivery thereof, (i) this Agreement or any Collateral Document ceases to be in full force and effect (other than by reason of a release of Collateral in accordance with the terms hereof or the satisfaction in full of the Obligations in accordance with the terms hereof) or shall be declared null and void, or the Collateral Agent shall not have or shall cease to have a valid and perfected Lien in any Collateral purported to be covered by the Collateral Documents with the priority required by the relevant Collateral Document, in each case for any reason other than the failure of the Collateral Agent or any Secured Party to take any action within its control; or (ii) any of the Credit Documents identified in clause (a) of the definition thereof for any reason, other than the satisfaction in full of all Obligations (other than contingent indemnification obligations), shall cease to be in full force and effect (other than in accordance with its terms) or shall be declared to be null and void or a party thereto, as the case may be, shall repudiate its obligations thereunder or shall contest the validity or enforceability of any Credit Document or Related Agreement in writing; or
 - (m) Servicing Agreement. A Servicer Default shall have occurred and be continuing; or
- (n) <u>Defaults Under Guaranties.</u> A default in CPS' obligations under the Limited Guaranty or the Borrowing Base Guaranty shall have occurred; or
- (o) Borrowing Base Deficiency. Failure by the Borrower to pay any Borrowing Base Deficiency within two (2) Business Days after the due date thereof: or

- (p) <u>Maturity Date.</u> Failure of the Borrower to pay the unpaid principal amount of and accrued interest on the Revolving Loans and all other Obligations on the Maturity Date; or
- (q) <u>Financial Statements</u>. The auditor's opinion accompanying the audited annual financial statements of any Credit Party delivered hereunder is qualified in any manner other than a qualification that relates solely to the internal controls or accounting processes and which is, in any event, not classified as a material weakness; or
 - (r) <u>Performance Triggers</u>. The breach of a Tier 2 Performance Trigger; or
- (s) <u>Material Adverse Change</u>. A material adverse change in the financial condition of the Borrower (provided, that for the avoidance of doubt, any change to the market value of the Receivables or other Collateral shall not be deemed a material adverse change in the financial condition of the Borrower).

THEN, (A) upon the occurrence of any Event of Default described in Section 7.1(f), 7.1(g) or 7.1(i), automatically, and (B) upon the occurrence of any other Event of Default, at the request of (or with the consent of) the Administrative Agent and the Lender, upon written notice to the Borrower and the Backup Servicer by the Administrative Agent, (w) the Revolving Commitments, if any, shall immediately terminate; (x) each of the following shall immediately become due and payable, in each case without presentment, demand, protest or other requirements of any kind, all of which are hereby expressly waived by the Borrower: (1) the unpaid principal amount of and accrued interest on the Revolving Loans and (2) all other Obligations; (y) all amounts on deposit in the Spread Account, if any, shall be distributed pursuant to Section 2.10 and (z) the Administrative Agent shall cause the Collateral Agent to enforce any and all Liens and security interests created pursuant to the Collateral Documents.

SECTION AGENTS

8.

- **8.1.** Appointment of Agents. Fortress is hereby appointed Administrative Agent and Collateral Agent hereunder and under the other Credit Documents and the Lender hereby authorizes Fortress, in such capacity, to act as its agent in accordance with the terms hereof and the other Credit Documents. Each Agent hereby agrees to act upon the express conditions contained herein and the other Credit Documents, as applicable. The provisions of this Section 8 are solely for the benefit of Agents and the Lender and the Borrower shall not have any rights as a beneficiary of any of the provisions thereof. In performing its functions and duties hereunder, each Agent shall act solely as an agent of the Lender and does not assume and shall not be deemed to have assumed any obligation towards or relationship of agency or trust with or for the Borrower.
- **8.2. Powers and Duties**. The Lender irrevocably authorizes each Agent to take such action on the Lender's behalf and to exercise such powers, rights and remedies hereunder and under the other Credit Documents as are specifically delegated or granted to such Agent by the terms hereof and thereof, together with such powers, rights and remedies as are reasonably incidental thereto. Each Agent shall have only those duties and responsibilities that are expressly specified herein and the other Credit Documents. Each Agent may exercise such powers, rights

and remedies and perform such duties by or through its agents or employees. No Agent shall have, by reason hereof or any of the other Credit Documents, a fiduciary relationship in respect of the Lender; and nothing herein or any of the other Credit Documents, expressed or implied, is intended to or shall be so construed as to impose upon any Agent any obligations in respect hereof or any of the other Credit Documents except as expressly set forth herein or therein.

8.3. <u>Collateral Documents and Guaranties</u>. The Lender hereby further authorizes the Administrative Agent or the Collateral Agent, as applicable, on behalf of and for the benefit the Lender, to be the agent for and representative of the Lender with respect to the Limited Guaranty, the Borrowing Base Guaranty, the Collateral and the Collateral Documents. Subject to Section 9.5, without further written consent or authorization from the Lender, the Administrative Agent or the Collateral Agent, as applicable may execute any documents or instruments necessary to release any Lien encumbering any item of Collateral that is the subject of a sale or other disposition of assets permitted hereby or to which the Lender has otherwise consented, or with respect to which the Lender has otherwise consented.

SECTION MISCELLANEOUS

- **9.1.** Notices. Unless otherwise specifically provided herein, any notice or other communication herein required or permitted to be given to CPS, the Borrower, any other Credit Party, the Collateral Agent or the Administrative Agent shall be sent to such Person's address as set forth on Appendix A or in the other relevant Credit Document, and in the case of the Lender, the address as indicated on Appendix A or otherwise indicated to the Administrative Agent in writing. Each notice hereunder shall be in writing and may be personally served, sent by telefacsimile (with telephonic confirmation of receipt) or courier service and shall be deemed to have been given when delivered in person or by courier service and signed for against receipt thereof, upon receipt of telefacsimile; provided, no notice to any Agent shall be effective until received by such Agent.
- **9.2.** Expenses. Whether or not the transactions contemplated hereby shall be consummated, the Borrower agrees to pay promptly (a) all the Agents' actual and reasonable, documented, out-of-pocket costs and expenses of preparation of the Credit Documents and any consents, amendments, waivers or other modifications thereto; (b) all the reasonable, documented fees, expenses and disbursements of counsel to the Agents in connection with the negotiation, preparation, execution and administration of the Credit Documents and any consents, amendments, waivers or other modifications thereto and any other documents or matters requested by the Borrower; (c) all the actual costs and reasonable, documented, out-of-pocket expenses of creating and perfecting Liens in favor of the Collateral Agent, for the benefit of the Secured Parties, including filing and recording fees, expenses and taxes, stamp or documentary taxes, search fees, title insurance premiums and reasonable fees, expenses and disbursements of counsel to each Agent; (d) each Agent's actual costs and reasonable documented, out-of-pocket fees, expenses for, and disbursements of any of such Agent's, auditors, accountants, consultants or appraisers whether internal or external, and all reasonable, documented attorneys' fees (including expenses and disbursements of outside counsel) incurred by such Agent; (e) all the actual costs and reasonable, documented, out-of-pocket expenses (including the reasonable fees, expenses and disbursements of any appraisers, consultants, advisors and agents employed or retained by the Collateral Agent and its counsel) in connection

with the custody or preservation of any of the Collateral; (f) all other actual and reasonable, documented out-of-pocket costs and expenses incurred by each Agent in connection with the syndication of the Revolving Loans and Commitments and the negotiation, preparation and execution of the Credit Documents and any consents, amendments, waivers or other modifications thereto and the transactions contemplated thereby; provided, that such costs and expenses shall not exceed \$10,000 unless such syndication is in connection with an increase in the Revolving Commitment; and (g) after the occurrence of a Default or an Event of Default, all documented costs and expenses, including reasonable attorneys' fees and costs of settlement, incurred by any Agent and the Lender in enforcing any Obligations of or in collecting any payments due from any Credit Party hereunder or under the other Credit Documents by reason of such Default or Event of Default (including in connection with the sale of, collection from, or other realization upon any of the Collateral or the enforcement of the Limited Guaranty or the Borrowing Base Guaranty) or in connection with any refinancing or restructuring of the credit arrangements provided hereunder in the nature of a "work-out" or pursuant to any insolvency or bankruptcy cases or proceedings.

9.3. Indemnity.

- (a) IN ADDITION TO THE PAYMENT OF EXPENSES PURSUANT TO SECTION 9.2, WHETHER OR NOT THE TRANSACTIONS CONTEMPLATED HEREBY SHALL BE CONSUMMATED, EACH OF CPS AND THE BORROWER AGREES, SEVERALLY BUT NOT JOINTLY, TO DEFEND (SUBJECT TO INDEMNITEES' SELECTION OF COUNSEL), INDEMNIFY, PAY AND HOLD HARMLESS, EACH AGENT AND THE LENDER, THEIR AFFILIATES AND THEIR RESPECTIVE OFFICERS, PARTNERS, DIRECTORS, TRUSTEES, EMPLOYEES AND AGENTS OF EACH AGENT AND THE LENDER (EACH, AN "INDEMNITEE"), FROM AND AGAINST ANY AND ALL INDEMNIFIED LIABILITIES, IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE, CONTRIBUTORY, OR SOLE NEGLIGENCE OF SUCH INDEMNITEE; PROVIDED, NEITHER CPS NOR THE BORROWER SHALL HAVE ANY OBLIGATION TO ANY INDEMNITEE HEREUNDER WITH RESPECT TO ANY INDEMNIFIED LIABILITIES TO THE EXTENT SUCH INDEMNIFIED LIABILITIES ARISE FROM THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF SUCH INDEMNITEE, AS DETERMINED BY A COURT OF COMPETENT JURISDICTION IN A FINAL NON-APPEALABLE ORDER OR JUDGMENT. TO THE EXTENT THAT THE UNDERTAKINGS TO DEFEND, INDEMNIFY, PAY AND HOLD HARMLESS SET FORTH IN THIS SECTION 9.3 MAY BE UNENFORCEABLE IN WHOLE OR IN PART BECAUSE THEY ARE VIOLATIVE OF ANY LAW OR PUBLIC POLICY, CPS OR THE BORROWER, AS APPLICABLE, SHALL CONTRIBUTE THE MAXIMUM PORTION THAT IT IS PERMITTED TO PAY AND SATISFY UNDER APPLICABLE LAW TO THE PAYMENT AND SATISFACTION OF ALL INDEMNIFIED LIABILITIES INCURRED BY INDEMNITEES OR ANY OF THEM.
- (b) To the extent permitted by applicable law, neither CPS nor the Borrower shall assert, and each of CPS and the Borrower hereby waives, any claim against the Lender, the Agents and their respective Affiliates, directors, employees, attorneys or agents, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) (whether or not the claim therefor is based on contract, tort or duty imposed by any

applicable legal requirement) arising out of, in connection with, as a result of, or in any way related to, this Agreement or any Credit Document or any agreement or instrument contemplated hereby or thereby or referred to herein or therein, the transactions contemplated hereby or thereby, any Revolving Loan or the use of the proceeds thereof or any act or omission or event occurring in connection therewith, and each of CPS and the Borrower hereby waives, releases and agrees not to sue upon any such claim or any such damages, whether or not accrued and whether or not known or suspected to exist in its favor.

9.4. Set-Off. In addition to any rights now or hereafter granted under applicable law and not by way of limitation of any such rights, upon the occurrence of any Event of Default the Lender and its Affiliates each is hereby authorized by the Borrower at any time or from time to time subject to the consent of the Administrative Agent, without notice to the Borrower or to any other Person (other than the Administrative Agent) except to the extent required by applicable law, any such notice being hereby expressly waived to the maximum extent under applicable law, and subject to any requirements or limitations imposed by applicable law, to set off and to appropriate and to apply any and all deposits (general or special, including Indebtedness evidenced by certificates of deposit, whether matured or unmatured, but not including trust accounts (in whatever currency)) and any other Indebtedness at any time held or owing by the Lender to or for the credit or the account of the Borrower (in whatever currency) against and on account of the obligations and liabilities of the Borrower to the Lender arising hereunder or under the other Credit Documents, including all claims of any nature or description arising out of or connected hereto or with any other Credit Document, irrespective of whether or not (a) the Lender shall have made any demand hereunder, (b) the principal of or the interest on the Revolving Loans or any other amounts due hereunder shall have become due and payable pursuant to Section 2 and although such obligations and liabilities, or any of them, may be contingent or unmatured or (c) such obligation or liability is owed to a branch or office of the Lender different from the branch or office holding such deposit or obligation or such Indebtedness.

9.5. Amendments and Waivers.

- (a) <u>Agent Consent</u>. No amendment, modification, termination or waiver of any provision of the Credit Documents, or consent to any departure by any Credit Party therefrom, shall in any event be effective without the written concurrence of each Credit Party that is party thereto and the Administrative Agent.
- (b) <u>Lender Consent</u>. Without the written consent of the Lender, no amendment, modification, termination, or consent shall be effective if the effect thereof would:
 - (i) extend the scheduled final maturity of any Revolving Loan or Revolving Loan Note;
 - (ii) waive, reduce or postpone any scheduled repayment;
 - (iii) reduce the rate of interest on any Revolving Loan (other than any waiver of any increase in the interest rate applicable to any Revolving Loan pursuant to Section 2.6) or any fee payable hereunder;

- (iv) extend the time for payment of any such interest or fees;
- (v) reduce the principal amount of any Revolving Loan;
- (vi) (A) amend the definition of "**Borrowing Base**" in a manner that increases the Revolving Availability to the Borrower or (B) amend, modify, terminate or waive any provision of this Section 9.5(b) or Section 9.5(c);
- (vii) release all or substantially all of the Collateral, the Guarantor from the Borrowing Base Guaranty or the Limited Guaranty, except as expressly provided in the Credit Documents; or
- (viii) consent to the assignment or transfer by any Credit Party of any of its rights and obligations under any Credit Document.
- (c) Other Consents. No amendment, modification, termination or waiver of any provision of the Credit Documents, or consent to any departure by any Credit Party therefrom, shall:
 - (i) increase the Revolving Commitment without the consent of the Lender;
 - (ii) amend, modify, terminate or waive any provision of Section 3.2(a) with regard to any Credit Extension without the consent of the Lender;
 - (iii) amend, modify, terminate or waive any provision of Section 8 as the same applies to any Agent, or any other provision hereof as the same applies to the rights or obligations of any Agent, in each case without the consent of such Agent; or
 - (iv) adversely affect the Controlled Account Bank, the Lockbox Account Bank, the Backup Servicer or the Custodian without the consent of such affected party.
- (d) Execution of Amendments, etc. The Administrative Agent may, but shall have no obligation to, with the concurrence of the Lender, execute amendments, modifications, waivers or consents on behalf of the Lender. Any waiver or consent shall be effective only in the specific instance and for the specific purpose for which it was given. No notice to or demand on a Credit Party in any case shall entitle such Credit Party to any other or further notice or demand in similar or other circumstances. Any amendment, modification, termination, waiver or consent effected in accordance with this Section 9.5 shall be binding upon the Lender at the time outstanding, each future Lender and, if signed by a Credit Party, upon such Credit Party. Notwithstanding anything to the contrary contained in this Section 9.5, if the Administrative Agent and the Credit Parties shall have jointly identified an obvious error or any error or omission of a technical nature, in each case that is immaterial (as determined by the Administrative Agent in its sole discretion), in any provision of the Credit Documents, then the Administrative Agent (in their respective applicable capacities as the Administrative Agent or Collateral Agent) and the Credit Parties shall be permitted to amend such provision and such amendment shall become effective without any further action or consent by the Lender if the

same is not objected to in writing by the Lender within five (5) Business Days following receipt of notice thereof.

9.6. Successors and Assigns; Participations.

- (a) <u>Generally</u>. This Agreement shall be binding upon the parties hereto and their respective successors and assigns and shall inure to the benefit of the parties hereto and the successors and assigns of the Lender. None of CPS' nor the Borrower's rights or obligations hereunder nor any interest herein may be assigned or delegated without the prior written consent of the Lender. Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, Indemnitees under Section 9.3, their respective successors and assigns permitted hereby and, to the extent expressly contemplated hereby, Affiliates of each of the Agents and the Lender) any legal or equitable right, remedy or claim under or by reason of this Agreement.
- (b) Register. The Borrower, CPS, the Administrative Agent and the Lender shall deem and treat the Persons listed as "Lenders" in the Register as the holders and owners of the corresponding Revolving Commitments and Revolving Loans listed therein for all purposes hereof, and no assignment or transfer of any such Revolving Commitment or Revolving Loan shall be effective, in each case, unless and until an Assignment Agreement effecting the assignment or transfer thereof shall have been delivered to and accepted by the Administrative Agent and recorded in the Register as provided in Section 9.6(e). Prior to such recordation, all amounts owed with respect to the applicable Revolving Commitment or Revolving Loan shall be owed to the Lender listed in the Register as the owner thereof, and any request, authority or consent of any Person who, at the time of making such request or giving such authority or consent, is listed in the Register as a Lender shall be conclusive and binding on any subsequent holder, assignee or transferee of the corresponding Revolving Commitments or Revolving Loans.
- (c) <u>Right to Assign</u>. Each Agent and the Lender shall have the right at any time to sell, assign or transfer all or a portion of its rights and obligations under this Agreement, including, without limitation, all or a portion of its Revolving Commitment or Revolving Loans owing to it or other Obligations to any Person that, at the time of such assignment, is able to perform the obligations of the Lender hereunder, or to any Fortress Affiliate Lender without the consent of the Borrower or CPS.
- (d) <u>Rating Agencies</u>. Each of the Borrower and CPS agrees that the Lender and the Administrative Agent shall have the right to disclose the terms of this Agreement and the transactions contemplated hereby to any Rating Agency. In addition, each of the Borrower and CPS agrees to provide, or cause to be provided, to the Rating Agencies any information, books, records, financial statements or other documents as reasonably requested by the Rating Agencies.
- (e) <u>Notice of Assignment</u>. Upon its receipt and acceptance of a duly executed and completed Assignment Agreement, any forms, certificates or other evidence required by this Agreement in connection therewith, the Administrative Agent shall record the information contained in such Assignment Agreement in the Register, shall give prompt notice thereof to CPS and the Borrower and shall maintain a copy of such Assignment Agreement.

- (f) Representations and Warranties of Assignee. Each assignee of the Lender, upon executing and delivering an Assignment Agreement, represents and warrants as of the applicable Effective Date (as defined in the applicable Assignment Agreement) that (i) it has experience and expertise in the making of or investing in commitments or loans such as the applicable Revolving Commitments or Revolving Loans, as the case may be; (ii) it will make or invest in, as the case may be, its Revolving Commitments or Revolving Loans for its own account in the ordinary course of its business and without a view to distribution of such Revolving Commitments or Revolving Loans within the meaning of the Securities Act or the Exchange Act or other federal securities laws (it being understood that, subject to the provisions of this Section 9.6, the disposition of such Revolving Commitments or Revolving Loans or any interests therein shall at all times remain within its exclusive control); and (iii) such assignee does not own or control, or own or control any Person owning or controlling, any trade debt or Indebtedness of CPS or the Borrower other than the Obligations or any Capital Stock of CPS or the Borrower.
- Assignment Agreement: (i) the assignee thereunder shall have the rights and obligations of a "Lender" hereunder to the extent such rights and obligations hereunder have been assigned to it pursuant to such Assignment Agreement and shall thereafter be a party hereto and a "Lender" for all purposes hereof; (ii) the assigning Lender thereunder shall, to the extent that rights and obligations hereunder have been assigned thereby pursuant to such Assignment Agreement, relinquish its rights (other than any rights which survive the termination hereof under Section 9.8) and be released from its obligations hereunder (and, in the case of an Assignment Agreement covering all or the remaining portion of an assigning Lender's rights and obligations hereunder, such Lender shall cease to be a party hereto; provided, anything contained in any of the Credit Documents to the contrary notwithstanding, such assigning Lender shall continue to be entitled to the benefit of all indemnities hereunder as specified herein with respect to matters arising out of the prior involvement of such assigning Lender as a Lender hereunder); (iii) the Revolving Commitments shall be modified to reflect the Revolving Commitment of such assignee and any Revolving Commitment of such assigning Lender, if any; and (iv) if any such assignment occurs after the issuance of any Revolving Loan Note hereunder, the assigning Lender shall, upon the effectiveness of such assignment or as promptly thereafter as practicable, surrender its applicable Revolving Loan Notes to the Administrative Agent for cancellation, and thereupon the Borrower shall issue and deliver new Revolving Loan Notes, if so requested by the assignee and/or outstanding Revolving Loans of the assignee and/or the assigning Lender.
- (h) <u>Participations</u>. The Lender shall have the right at any time to sell one or more participations to any Person (other than CPS, any of its Subsidiaries or any of its Affiliates) in all or any part of its Revolving Commitments, Revolving Loans or in any other Obligation; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, no participations may be sold to any Person acquiring such participation with the assets of, or for the benefit of, any employee benefit plan subject to Title I of ERISA, any "plan" subject to Section 4975 of the Internal Revenue Code, or any entity whose underlying assets include plan assets by reason of a plan's investment in such entity. The holder of any such participation, other than an Affiliate of the Lender

granting such participation, shall not be entitled to require such Lender to take or omit to take any action hereunder except with respect to any amendment, modification or waiver that would (i) extend the final scheduled maturity of any Revolving Loan or Revolving Loan Note in which such participant is participating, or reduce the rate or extend the time of payment of interest or fees thereon (except in connection with a waiver of applicability of any postdefault increase in interest rates) or reduce the principal amount thereof, or increase the amount of the participant's participation over the amount thereof then in effect (it being understood that a waiver of any Default or Event of Default or of a mandatory reduction in the Revolving Commitment shall not constitute a change in the terms of such participation, and that an increase in any Revolving Commitment or Revolving Loan shall be permitted without the consent of any participant if the participant's participation is not increased as a result thereof), (ii) consent to the assignment or transfer by CPS or the Borrower of any of its rights and obligations under this Agreement, or (iii) release all or substantially all of the Collateral under the Collateral Documents, the Guarantor from the Borrowing Base Guaranty or the Limited Guaranty (in each case, except as expressly provided in the Credit Documents) supporting the Revolving Loans hereunder in which such participant is participating. Each of CPS and the Borrower agrees that each participant shall be entitled to the benefits of Sections 2.15 and 2.16 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to clause (c) of this Section 9.6; provided, (i) a participant shall not be entitled to receive any greater payment under Section 2.15 and 2.16 than the applicable Lender would have been entitled to receive with respect to the participation sold to such participant, unless the sale of the participation to such participant is made with the Borrower's prior written consent, (ii) a participant that would be a Non-US Lender if it were a Lender shall not be entitled to the benefits of Section 2.16 unless the Borrower is notified of the participation sold to such participant and such participant complies with Section 2.16 as though it were a Lender and (iii) such participant complies with the written statement and notice requirements therein. To the extent permitted by law, each participant also shall be entitled to the benefits of Section 9.4 as though it were a Lender. Notwithstanding any participation made hereunder (i) such selling Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the Borrower for the performance of its obligations hereunder, and (iii) except as set forth above, the Credit Parties, the Agents and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement, and such Lender shall retain the sole right to enforce the obligations of the Credit Parties relating to the Obligations and to approve, without the consent of or consultation with any participant, any amendment, modification or waiver of any provision of this Agreement. Each Lender that sells a participation shall maintain a register on which it enters the name and address of each participant and the amounts of each participant's participation (the "Participant Register"). The entries in the Participant Register shall be conclusive, absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary.

(i) <u>Certain Other Assignments</u>. In addition to any other assignment permitted pursuant to this Section 9.6, the Lender may assign, pledge and/or grant a security interest in, all or any portion of its Revolving Loans, the other Obligations owed by or to the Lender, and its Revolving Loan Notes, if any, to secure obligations of the Lender including, without limitation, any Federal Reserve Bank as collateral security pursuant to Regulation A of the Board of Governors of the Federal Reserve System and any operating circular issued by such Federal

Reserve Bank; <u>provided</u>, the Lender, as between the Borrower and the Lender, shall not be relieved of any of its obligations hereunder as a result of any such assignment and pledge, and <u>provided further</u>, in no event shall the applicable Federal Reserve Bank, pledgee or trustee be considered to be a "**Lender**" or be entitled to require the assigning Lender to take or omit to take any action hereunder.

- (j) Fortress Affiliate Lenders. Notwithstanding anything to the contrary set forth in this Section 9.6, each party hereto acknowledges and agrees that (a) Fortress and/or its Affiliates may, in its sole discretion, at any time, and from time to time, in connection with any assignment made in accordance with this Section 9.6 to any Affiliates of Fortress (each such Affiliate, a "Fortress Affiliate Lender"), (i) assign to any Fortress Affiliate Lender all or any portion of Fortress' or any Fortress Affiliate Lender's Revolving Loans then existing and/or Fortress' or any such Fortress Affiliate Lender's Revolving Commitments then outstanding in such percentages or fixed dollar amounts as Fortress or any such Fortress Affiliate Lender shall determine in their respective sole discretion, and (ii) in furtherance of the foregoing, request the Borrower to, and the Borrower shall, execute and deliver to Fortress or such Fortress Affiliate Lender, as the case may be, any Revolving Loan Note (or replacement therefor) requested pursuant to Section 2.3(b) to reflect such assignment and (b) Fortress and one or more Fortress Affiliate Lenders may enter into agreements by and among Fortress and such Fortress Affiliate Lenders and (if applicable) the Administrative Agent with respect to such assignments, including, without limitation, the ability to create a sequential pay feature amongst Fortress and such Fortress Affiliate Lenders; provided that the terms of any such agreements do not affect the terms of the Credit Documents (other than the Assignment Agreement to the extent required to reflect such agreements) or the Borrower's rights or obligations under any Credit Document; provided, further, that (1) the percentage or fixed dollar amount of such assignment is set forth in the related Assignment Agreement, (2) the aggregate Revolving Commitment amongst Fortress and such Fortress Affiliate Lenders immediately following any such assignment remains unchanged from the aggregate Revolving Commitments of Fortress and the Fortress Affiliate Lenders immediately prior to such assignment and (3) no such assignment shall increase any of the Borrower's Obligations hereunder; and provided, further, any such Fortress Affiliate Lender subject to any such assignment for any such percentage or fixed dollar amount (whether such assignment is solely of the existing Revolving Loans or the outstanding Revolving Commitments, or both) shall be a "Lender" for all purposes hereunder, under the other Credit Documents and under the Related Agreements subject, solely as among Fortress and the Fortress Affiliate Lenders, to such restrictions set forth amongst Fortress and any such Fortress Affiliate Lenders set forth in the related Assignment Agreement (including the ability for any such assignee to not be assigned any future funding obligations in respect of Revolving Commitments other than to the extent of any repayments received by such assignee on the principal balance of the assigned Revolving Loan (provided that such Assignment Agreement shall also provide that the relevant assignor expressly retains any such future funding obligations beyond principal repayments to such assignee on the principal balance of such assigned Revolving Loan)).
- **9.7.** <u>Independence of Covenants</u>. All covenants hereunder shall be given independent effect so that if a particular action or condition is not permitted by any of such covenants, the fact that it would be permitted by an exception to, or would otherwise be within the limitations of, another covenant shall not avoid the occurrence of a Default or an Event of Default if such action is taken or condition exists.

- **9.8.** <u>Survival of Representations, Warranties and Agreements</u>. All representations, warranties and agreements made herein shall survive the execution and delivery hereof and the making of any Credit Extension. Notwithstanding anything herein or implied by law to the contrary, the agreements of the Borrower set forth in Sections 2.15, 2.16, 9.2, 9.3, 9.4 and 9.10 shall survive the payment of the Revolving Loans and the termination hereof.
- **9.9.** No Waiver; Remedies Cumulative. No failure or delay on the part of any Agent or the Lender in the exercise of any power, right or privilege hereunder or under any other Credit Document shall impair such power, right or privilege or be construed to be a waiver of any default or acquiescence therein, nor shall any single or partial exercise of any such power, right or privilege preclude other or further exercise thereof or of any other power, right or privilege. The rights, powers and remedies given to each Agent and the Lender hereby are cumulative and shall be in addition to and independent of all rights, powers and remedies existing by virtue of any statute or rule of law or in any of the other Credit Documents or any of the Interest Rate Agreements. Any forbearance or failure to exercise, and any delay in exercising, any right, power or remedy hereunder shall not impair any such right, power or remedy or be construed to be a waiver thereof, nor shall it preclude the further exercise of any such right, power or remedy.
- **9.10.** Marshalling; Payments Set Aside. Neither any Agent nor the Lender shall be under any obligation to marshal any assets in favor of the Borrower or any other Person or against or in payment of any or all of the Obligations. To the extent that any Credit Party makes a payment or payments to the Administrative Agent or the Lender (or to the Administrative Agent, on behalf of the Lender), or the Administrative Agent, the Collateral Agent or the Lender enforce any security interests or exercise their rights of setoff, and such payment or payments or the proceeds of such enforcement or setoff or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside and/or required to be repaid to a trustee, receiver or any other party under any bankruptcy law, any other state or federal law, common law or any equitable cause, then, to the extent of such recovery, the obligation or part thereof originally intended to be satisfied, and all Liens, rights and remedies therefor or related thereto, shall be revived and continued in full force and effect as if such payment or payments had not been made or such enforcement or setoff had not occurred.
- **9.11.** Severability. In case any provision or obligation hereunder or any Revolving Loan Note or other Credit Document shall be invalid, illegal or unenforceable in any jurisdiction, the validity, legality and enforceability of the remaining provisions or obligations, or of such provision or obligation in any other jurisdiction, shall not in any way be affected or impaired thereby.
- **9.12. Headings**. Section headings herein are included herein for convenience of reference only and shall not constitute a part hereof for any other purpose or be given any substantive effect.
- 9.13. APPLICABLE LAW. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICT OF LAWS

PRINCIPLES (OTHER THAN SECTIONS 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW) THEREOF.

9.14. CONSENT TO JURISDICTION.

- (a) ALL JUDICIAL PROCEEDINGS BROUGHT AGAINST

 CPS OR THE BORROWER ARISING OUT OF OR RELATING HERETO OR ANY OTHER CREDIT DOCUMENT, OR ANY OF THE OBLIGATIONS, MAY BE BROUGHT IN ANY STATE OR FEDERAL COURT OF COMPETENT JURISDICTION IN THE STATE, COUNTY AND CITY OF NEW YORK. BY EXECUTING AND DELIVERING THIS AGREEMENT, EACH OF CPS AND THE BORROWER, FOR ITSELF AND IN CONNECTION WITH ITS PROPERTIES, IRREVOCABLY (i) ACCEPTS GENERALLY AND UNCONDITIONALLY THE NONEXCLUSIVE JURISDICTION AND VENUE OF SUCH COURTS; (ii) WAIVES ANY DEFENSE OF FORUM NON CONVENIENS; (iii) AGREES THAT SERVICE OF ALL PROCESS IN ANY SUCH PROCEEDING IN ANY SUCH COURT MAY BE MADE BY REGISTERED OR CERTIFIED MAIL, RETURN RECEIPT REQUESTED, TO THE APPLICABLE CREDIT PARTY AT ITS ADDRESS PROVIDED IN ACCORDANCE WITH SECTION 9.1 AND TO ANY PROCESS AGENT SELECTED IN ACCORDANCE WITH SECTION 3.1(q) ABOVE IS SUFFICIENT TO CONFER PERSONAL JURISDICTION OVER CPS OR THE BORROWER, AS APPLICABLE, IN ANY SUCH PROCEEDING IN ANY SUCH COURT, AND OTHERWISE CONSTITUTES EFFECTIVE AND BINDING SERVICE IN EVERY RESPECT; AND (iv) AGREES THAT AGENTS AND THE LENDER RETAIN THE RIGHT TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR TO BRING PROCEEDINGS AGAINST CPS OR THE BORROWER, AS APPLICABLE, IN THE COURTS OF ANY OTHER JURISDICTION.
- (b) EACH OF CPS AND THE BORROWER HEREBY AGREES THAT PROCESS MAY BE SERVED ON IT BY CERTIFIED MAIL, RETURN RECEIPT REQUESTED, TO THE ADDRESSES PERTAINING TO IT AS SPECIFIED IN SECTION 9.1, AND HEREBY APPOINTS CT CORPORATION SYSTEM, 111 8TH AVENUE, NEW YORK, NEW YORK 10011, AS ITS AGENT TO RECEIVE SUCH SERVICE OF PROCESS. ANY AND ALL SERVICE OF PROCESS AND ANY OTHER NOTICE IN ANY SUCH ACTION, SUIT OR PROCEEDING SHALL BE EFFECTIVE AGAINST CPS OR THE BORROWER, AS APPLICABLE, IF GIVEN BY REGISTERED OR CERTIFIED MAIL, RETURN RECEIPT REQUESTED, OR BY ANY OTHER MEANS OR MAIL WHICH REQUIRES A SIGNED RECEIPT, POSTAGE PREPAID, MAILED AS PROVIDED ABOVE. IN THE EVENT CT CORPORATION SYSTEM. SHALL NOT BE ABLE TO ACCEPT SERVICE OF PROCESS AS AFORESAID AND IF CPS OR THE BORROWER, AS APPLICABLE, SHALL NOT MAINTAIN AN OFFICE IN NEW YORK CITY, SUCH CREDIT PARTY SHALL PROMPTLY APPOINT AND MAINTAIN AN AGENT QUALIFIED TO ACT AS AN AGENT FOR SERVICE OF PROCESS WITH RESPECT TO THE COURTS SPECIFIED IN THIS SECTION 9.14 ABOVE, AND ACCEPTABLE TO THE ADMINISTRATIVE AGENT, AS CPS' OR THE BORROWER'S, AS APPLICABLE, AUTHORIZED AGENT TO ACCEPT AND ACKNOWLEDGE ON THE BORROWER'S BEHALF SERVICE OF ANY AND ALL PROCESS WHICH MAY BE SERVED IN ANY SUCH ACTION, SUIT OR PROCEEDING.

9.15. WAIVER OF JURY TRIAL. EACH OF CPS AND THE BORROWER HEREBY AGREES TO WAIVE ITS RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING HEREUNDER OR UNDER ANY OF THE OTHER CREDIT DOCUMENTS OR ANY DEALINGS BETWEEN IT RELATING TO THE SUBJECT MATTER OF THIS LOAN TRANSACTION OR THE LENDER/BORROWER RELATIONSHIP THAT IS BEING ESTABLISHED. THE SCOPE OF THIS WAIVER IS INTENDED TO BE ALL-ENCOMPASSING OF ANY AND ALL DISPUTES THAT MAY BE FILED IN ANY COURT AND THAT RELATE TO THE SUBJECT MATTER OF THIS TRANSACTION, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS AND ALL OTHER COMMON LAW AND STATUTORY CLAIMS. EACH OF CPS AND THE BORROWER ACKNOWLEDGES THAT THIS WAIVER IS A MATERIAL INDUCEMENT TO ENTER INTO A BUSINESS RELATIONSHIP, THAT IT HAS ALREADY RELIED ON THIS WAIVER IN ENTERING INTO THIS AGREEMENT, AND THAT IT WILL CONTINUE TO RELY ON THIS WAIVER IN ITS RELATED FUTURE DEALINGS. EACH OF CPS AND THE BORROWER FURTHER WARRANTS AND REPRESENTS THAT IT HAS REVIEWED THIS WAIVER WITH ITS LEGAL COUNSEL AND THAT IT KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL. THIS WAIVER IS IRREVOCABLE, MEANING THAT IT MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING (OTHER THAN BY A MUTUAL WRITTEN WAIVER SPECIFICALLY REFERRING TO THIS SECTION 9.15 AND EXECUTED BY EACH OF THE PARTIES HERETO), AND THIS WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS HERETO OR ANY OF THE OTHER CREDIT DOCUMENTS OR TO ANY OTHER DOCUMENTS OR AGREEMENTS RELATING TO THE REVOLVING LOANS MADE HEREUNDER. IN THE EVENT OF LITIGATION, THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

9.16. <u>Usury Savings Clause</u>. Notwithstanding any other provision herein, the aggregate interest rate charged or agreed to be paid with respect to any of the Obligations, including all charges or fees in connection therewith deemed in the nature of interest under applicable law shall not exceed the Highest Lawful Rate. If the rate of interest (determined without regard to the preceding sentence) under this Agreement at any time exceeds the Highest Lawful Rate, the outstanding amount of the Revolving Loans made hereunder shall bear interest at the Highest Lawful Rate until the total amount of interest due hereunder equals the amount of interest which would have been due hereunder if the stated rates of interest set forth in this Agreement had at all times been in effect. In addition, if when the Revolving Loans made hereunder are repaid in full the total interest due hereunder (taking into account the increase provided for above) is less than the total amount of interest which would have been due hereunder if the stated rates of interest set forth in this Agreement had at all times been in effect, then to the extent permitted by law, the Borrower shall pay to the Administrative Agent an amount equal to the difference between the amount of interest paid and the amount of interest which would have been paid if the Highest Lawful Rate had at all times been in effect. Notwithstanding the foregoing, it is the intention of the Lender and the Borrower to conform strictly to any applicable usury laws. Accordingly, if the Lender contracts for, charges, or receives any consideration which constitutes interest in excess of the Highest Lawful Rate, then any such excess shall be cancelled automatically and, if previously paid, shall at the Lender's

option be applied to the outstanding amount of the Revolving Loans made hereunder or be refunded to the Borrower. In determining whether the interest contracted for, charged, or received by the Administrative Agent or the Lender exceeds the Highest Lawful Rate, such Person may, to the extent permitted by applicable law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest, throughout the contemplated term of the Obligations hereunder.

- **9.17.** Counterparts. This Agreement may be executed in any number of counterparts, each of which when so executed and delivered shall be deemed an original, but all such counterparts together shall constitute but one and the same instrument. Delivery of an executed signature page to this Agreement by facsimile transmission or other electronic image scan transmission (e.g., "PDF" or "tif" via email) shall be as effective as delivery of a manually signed counterpart of this Agreement.
- **9.18. Effectiveness.** This Agreement shall become effective upon the execution of a counterpart hereof by each of the parties hereto and receipt by the Borrower and the Administrative Agent of written or telephonic notification of such execution and authorization of delivery thereof.
- **9.19.** Money Control Act. The Lender and the Administrative Agent (for itself and not on behalf of the Lender) hereby notifies CPS and the Borrower that pursuant to the requirements of the Act, it is required to obtain, verify and record information that identifies CPS and the Borrower, which information includes the name and address of each of CPS and the Borrower and other information that will allow the Lender or the Administrative Agent, as applicable, to identify CPS and the Borrower in accordance with the Act.
- **9.20. Prior Agreements**. This Agreement and the other Credit Documents contain the entire agreement of the parties hereto and thereto in respect of the transactions contemplated hereby and thereby, and all prior agreements among or between such parties, whether oral or written, are superseded by the terms of this Agreement and the other Credit Documents and unless specifically set forth in a writing contemporaneous herewith the terms, conditions and provisions of any and all such prior agreements do not survive execution of this Agreement.
- **9.21.** Third Party Beneficiaries. The Backup Servicer, the Custodian, the Controlled Account Bank and the Lockbox Account Bank shall be express third party beneficiaries of the provisions of Section 2.10(i), Section 2.12(iii) and Section 2.12(ix).

9.22. Exclusivity; Right of First Refusal.

- (a) Each Credit Party agrees that, subject to the availability of Revolving Commitments, no less than 50% of receivables (by principal balance) that qualify as Eligible Receivables that the Originator or its Affiliates originates or acquires during any Collection Period during the Revolving Commitment Period shall be financed through the Facility (such agreement, the "**Exclusivity Covenant**").
- (b) CPS shall notify Fortress of its intention to enter into as a borrower, either directly or indirectly (acting through a special purpose borrowing entity, or otherwise indirectly)

any financing facility other than the Facility contemplated by this Agreement (any such facility, a "**Proposed Facility**"). Subject to clause (c) below, within ten (10) days of receipt of such notice, Fortress shall have the right to prevent CPS from entering into such Proposed Facility by providing financing on all of the same material terms as the Proposed Facility (such right, the "**Right of First Refusal**"). In no event shall the Borrower enter into any financing facility other than the Facility contemplated by this Agreement.

- (c) Notwithstanding the foregoing, Fortress shall not be entitled to exercise the Right of First Refusal with respect to additional financing obtained by CPS or its Affiliates after the Closing Date (the "Allowed Financing") so long as CPS and its Affiliates remain in compliance with the Exclusivity Covenant. At any time as the amount financed hereunder reaches 85% of the maximum initial Revolving Commitment, the Lender, at its sole discretion, may increase the Revolving Commitment at such time by two (2) \$25,000,000 increments (which increments may be exercised together or separately in Lender's sole discretion), provided, however, that advances pursuant to such increased Revolving Commitment, but not advances pursuant to the initial Revolving Commitment, shall be on substantially the same financial terms as are applicable to the Allowed Financing at the time of the related increase in the Revolving Commitment. In compliance with the foregoing, CPS and its Affiliates shall remain obligated to comply with the Exclusivity Covenant.
- (d) In the event that the Administrative Agent chooses not to provide such financing, the Credit Parties (other than the Borrower) may enter into the Proposed Facility; <u>provided</u>, <u>however</u>, that prior to entering into the Proposed Facility, the Credit Parties shall have taken measures reasonably acceptable to the Administrative Agent to ensure that the selection procedures used to allocate receivables among any Proposed Facility or Allowed Financing, on the one hand, and the Facility contemplated hereby, on the other, will not (x) result in the Receivables pledged hereunder and under the Security Agreement being less desirable or valuable than other comparable Receivables originated or acquired by a Credit Party or (y) materially and adversely affect the interests of the Lender or the Agents.
- (e) In addition, each of CPS and the Borrower agrees that, prior to selling any Receivables to (x) any third party or (y) any Affiliate other than an Affiliate that intends to hold such Receivables until the Revolving Commitment Termination Date or to securitize such Receivables, the Borrower shall offer such Receivable for sale to the Lender; <u>provided</u>, that in no event shall CPS or the Borrower sell any Receivables in a transaction that results in a loss to any Agent or the Lender. The Lender shall have five (5) Business Days following the day on which the Lender receives offering materials (the "Offering Materials") with respect to such Receivables from CPS or the Borrower, as applicable, to provide an indicative bid (the "Indicative Bid"). If CPS or the Borrower, as applicable, accepts a Lender's Indicative Bid, the Lender shall have fifteen (15) Business Days from the time of acceptance of such Indicative Bid to complete its due diligence on the Receivables and either provide its final bid (the "Final Bid") for such Receivables or provide a rejection thereof to CPS or the Borrower, as applicable. After providing a Final Bid, the Lender shall have an additional fifteen (15) Business Days to close the purchase of such Receivables. The failure of the Lender to so respond or purchase within the applicable five (5) or fifteen (15) Business Day periods, shall constitute a waiver of the Lender's right to purchase such Receivables.

(f) Each of CPS and the Borrower agrees that, subject to eligibility criteria used by a Rating Agency in connection with rating a term
securitization of receivables, Receivables selected to be sold, or offered for sale, pursuant to the preceding paragraph shall be selected in a manner designed
not to (x) result in the Receivables pledged hereunder and under the Security Agreement being less desirable or valuable than other comparable receivable
originated or acquired by a Credit Party or (y) materially and adversely affect the interests of the Lender or the Agents.

9.23. Confidentiality. Each Credit Party agrees that the terms included in this Agreement and disclosed in connection with the consummation of the transactions contemplated hereby shall be kept strictly confidential, shall not be reproduced or disclosed (except as required by law, including, without limitation, the filing requirements of the Exchange Act), and shall not be used by either Credit Party other than in connection with the transaction described herein except with the prior written consent of Fortress; provided, however, that Fortress hereby consents to each Credit Party's disclosure of (i) this Agreement to its respective officers, directors, employees, attorneys, accountants, agents and advisors who are directly involved in the implementation of the terms and conditions of this Agreement to the extent such persons agree to hold the same in confidence, (ii) this Agreement as required by applicable law or compulsory legal process (in which case each Credit Party agrees to inform Fortress promptly thereof) and (iii) the terms of this Agreement upon and after its filing by CPS in accordance with the Exchange Act.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered by their respective officers thereunto duly authorized as of the date first written above.

PAGE FOUR FUNDING LLC

By: /s/ MARK CREATURA

Name: Mark Creatura Title: V.P. and Sec'y

CONSUMER PORTFOLIO SERVICES, INC.

By: /s/ ROBERT E. RIEDL

Name: Robert E. Riedl

Title: Sr. VP

FORTRESS CREDIT CORP.,

as Administrative Agent, Collateral Agent and Lender

By: /s/ CONSTANTINE M. DAKOLIAS

Name: Constantine M. Dakolias

Title: President

CERTIFICATION

- I, Charles E. Bradley, Jr., certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Consumer Portfolio Services, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the period presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e)) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f)) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles.
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 16, 2009

/s/ CHARLES E. BRADLEY, JR

Charles E. Bradley, Jr. Chief Executive Officer

CERTIFICATION

- I, Jeffrey P. Fritz, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Consumer Portfolio Services, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the period presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e)) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f)) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles.
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 16, 2009

/s/ JEFFREY P. FRITZ

Jeffrey P. Fritz, Chief Financial Officer

Certification Pursuant To 18 U.S.C. Section 1350, As Adopted Pursuant To Section 906 of The Sarbanes-Oxley Act Of 2002

In connection with the Quarterly Report on Form 10-Q of Consumer Portfolio Services, Inc. (the "Company") for the quarterly period ended September 30, 2009, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Charles E. Bradley, Jr., as Chief Executive Officer of the Company, and Jeffrey P. Fritz, as Chief Financial Officer of the Company, each hereby certifies, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 16, 2009

/s/ CHARLES E. BRADLEY, JR.
Charles E. Bradley, Jr.
Chief Executive Officer

/s/ JEFFREY P. FRITZ
Jeffrey P. Fritz
Chief Financial Officer

This certification accompanies each Report pursuant to § 906 of the Sarbanes-Oxley Act of 2002 and shall not, except to the extent required by the Sarbanes-Oxley Act of 2002, be deemed filed by the Company for purposes of §18 of the Securities Exchange Act of 1934, as amended.

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.